IMPROVING THE IMPLEMENTATION OF TRANSITIONAL CARE INNOVATIONS

AMAL FAKHA

IMPROVING THE IMPLEMENTATION OF TRANSITIONAL CARE INNOVATIONS

Amal Fakha

The research presented in this dissertation was conducted at the Care and Public Health Research Institute (CAPHRI), department of Health Services Research, Maastricht University. CAPHRI is part of The Netherlands School of Public Health and Care Research (CaRe).

This research is part of the TRANS-SENIOR Marie-Curie International Innovative Training network; an EU funded project designed to train healthcare innovators who will shape future care for senior citizens. Funding was provided by the European Union's Horizon 2020 research, an innovation programme under the Marie Sklodowska-Curie grant agreement No 812656, more information is available at <u>https://www.trans-senior.eu</u>.

This dissertation titled 'Improving the Implementation of Transitional Care Innovations' is part of a double degree with Katholieke Universiteit Leuven (KU Leuven).



Cover design: Ilse Modder | www.ilsemodder.nl Layout: Ilse Modder | www.ilsemodder.nl Printing: Gildeprint Enschede | www.gildeprint.nl

ISBN: 978-94-6419-837-9

Copyright © Amal Fakha, Maastricht, 2023

All rights reserved. No part of this thesis may be reproduced or transmitted in any form or by any means, electronic or mechanical, including photocopying, recoding on any information storage or retrieval system without permission from the author, or when appropriate, from the publisher of the publications.

IMPROVING THE IMPLEMENTATION OF TRANSITIONAL CARE INNOVATIONS

DISSERTATION

to obtain the degree of Doctor at Maastricht University, on the authority of the Rector Magnificus, Prof. dr. Pamela Habibović, and the degree of Doctor in biomedical sciences at KU Leuven, on the authority of the Rector, Prof. dr. Luc Sels, in accordance with the decision of the Board of Deans, to be defended in public on Monday July 10, 2023, at 16.00 hours

by

Amal Fakha

Supervisors

Prof. Dr. Hilde Verbeek, Universiteit Maastricht Prof. Dr. Jan P.H. Hamers, Universiteit Maastricht Prof. Dr. Theo van Achterberg, Katholieke Universiteit Leuven, Belgium

Co-supervisor

Dr. Bram de Boer, Universiteit Maastricht

Assessment committee

Prof. Dr. Marjolein de Vugt, Universiteit Maastricht (chair) Prof. Dr. Koen Milisen, Katholieke Universiteit Leuven, Belgium Prof. Dr. Carl Thompson, University of Leeds, United Kingdom Prof. Dr. Trudy van der Weijden, Universiteit Maastricht Prof. Dr. Michel Wensing, Universität Heidelberg, Germany

Dedicated to my Mother

"If you would indeed behold the spirit of death, open your heart wide unto the body of life. For life and death are one, even as the river and the sea are one" -Khalil Gibran, Lebanese poet

CONTENTS

CHAPTER 1. General Introduction	11
CHAPTER 2. A myriad of factors influencing the implementation of transitional care innovations: a scoping review	21
CHAPTER 3. Fostering the implementation of transitional care innovations for older persons: prioritizing the influencing key factors using a modified Delphi technique	65
CHAPTER 4. Implementing four transitional care interventions for older adults: a retrospective collective case study	115
CHAPTER 5 . Stakeholders' experiences and perception on transitional care initiatives within an integrated care project in Belgium: a qualitative interview study	151
CHAPTER 6 . Systematic development of a set of implementation strategies for transitional care innovations in long-term care	179
CHAPTER 7. General Discussion	221
CHAPTER 8. Summary	235
CHAPTER 9. Impact	243
ADDENDA Publications Acknowledgments About the author Living Lab in Ageing and Long-Term Care	249 250 252 255 256



CHAPTER

General Introduction

BACKGROUND

"Long old age" is a present reality for the world population, with an emergent fast rate of global demographic ageing. In an unprecedented time in history, people around the world can now expect to live far beyond 60 years of age (1). In Europe alone, it is projected that by year 2050 the number of people aged 65 or over will increase by 41% (2), and by 2060, this fraction of the population will range between 22-36%, with 12% aged 80 or more (3). However, living longer is at times not accompanied with good health and is often correlated with an increased risk of multimorbidity (4). Therefore, a proportion of the older population lives with chronic diseases and typically receives long-term care from multiple care providers (4, 5). Hence, older persons with chronic diseases move frequently across different healthcare settings and providers in order to address their complex and varied medical needs (6). These movements are highly prevalent, whereby on average an older person with one or more chronic conditions can see eight different physicians over the course of one year (7). Research showed that 23% of hospitalized patients aged 65 and above are usually discharged to another institution (e.g., nursing home, skilled nursing facility), and around 19% of residents in skilled nursing facilities transfer back to a hospital within 30 days (7). Moreover, older persons have at least one movement between care settings towards their end of life, while around 70% of older persons with dementia move from a hospital to a nursing home, and 23% are re-hospitalized annually (8, 9). Consequently, healthcare systems face a great challenge to deliver long-term care services that ensure care continuity for older persons when moving between care providers.

The research studies presented in this dissertation are part of the European TRANS-SENIOR research consortium, which aims to tackle challenges facing long-term care systems in Europe and generate evidence on improving care by avoiding unnecessary transitions and optimizing needed care transitions for older persons.

TRANSITIONAL CARE

Care transitions refer to "the movements that patients make between health care providers and settings as their medical condition and care needs change during the course of a chronic or acute illness" (7). Specifically, among older persons (65 years and above), these transitions are common and frequent due to the prevalence of complex and chronic health conditions and multimorbidity (4, 6, 10). For example, in the course of an acute illness, an older person might receive care in a home care or outpatient setting, then transition to a hospital for inpatient care, then move to a rehabilitation facility to receive the restorative care needed, before finally returning home. Every single move between care providers and settings is defined as a care transition.

GENERAL INTRODUCTION

While these transitions are sometimes predictable, they are recognized as risky and frequently hampered by diverse issues. Notably, breakdowns in patient handovers like disorganized handoff of information between care providers, delayed, missing, or inaccurate discharge summaries, and poor communication among caregivers often occur during transitions and cause errors in treatment plans, ultimately leading to fragmented care and suboptimal care transitions (11, 12). Hence, older persons are particularly vulnerable to poor health outcomes during care transitions, which are usually stressful periods for them and their families and/or informal caregivers (8). Being confused about medications, not obtaining a clear explanation of discharge information, and not receiving care according to needs are among the experiences of an unsafe transition commonly reported by older persons (13). Therefore poor care transitions can have a large impact on older persons and may lead to various adverse medical repercussions such as medication errors, hospital readmissions, and mortality (11, 14).

Transitional care is defined as "a broad range of time-limited services designed to ensure health care continuity, avoid preventable poor outcomes among at-risk populations, and promote the safe and timely transfer of patients from one level of care to another or from one type of setting to another", and hence aims to improve the overall patient experience during transitions by closing gaps in care (7, 15). Whereas the delivery of transitional care seems straightforward, it is rather filled with challenges. Innovative solutions are needed to ensure proper transitional care and promote better and safer care transitions.

INNOVATIONS IN TRANSITIONAL CARE

Innovation can be defined as "the intentional introduction and application within a role, group, or organization, of ideas, processes, products or procedures, new to the relevant unit of adoption, designed to significantly benefit the individual, the group, or wider society" (16). Hence, innovation aims to solve a problem or create value and has become a popular concept adopted in various healthcare settings. Likewise, multiple new practices and models of care are continuously developed and considered as key to the future of long-term care. These different innovations may provide promising solutions to reduce costs and demand for care, ensure the quality of care, improve outcomes for care recipients, and increase productivity (17, 18). Consequently, the current negative impact of poor care transitions on older persons drove the development of Transitional Care Innovation(s) (TCIs) designed to enhance care continuity and coordination for older persons when transferring between multiple care settings and providers.

In general, each TCI is characterized by a bundle of care services, focuses on specific care transition pathways (e.g., home or hospital to nursing home, hospital to home, or hospital to

CHAPTER 1

rehabilitation center then to home), and intends to either improve a necessary care transition or to prevent an avoidable one. Examples of TCIs' core components include patient/ caregiver engagement and education, complexity management, and care coordination (19). The presence of a healthcare professional assigned with a transition role (e.g., care transition nurse, transitional care manager, or care coordinator) is another key feature of some TCIs (20). For instance, a care transition nurse can help older persons during their transition between hospital and home by providing follow-up visits, developing individualized care plans, and coordinating care (21, 22). Other TCIs focus on transfer units within a residential care facility or a community setting. These units host older persons discharged from the hospital for a short period and provide them with restorative/rehabilitation therapies so they can regain their functional capacity and independence before transferring back home or to a nursing facility (23, 24). Multiple studies showed promising evidence on the effectiveness of TCIs to enhance transitional care for older persons (25). Reduced hospital readmission rates, decreased emergency room visits, and healthcare cost savings, as well as enhancements in the older person's satisfaction level and functional capacity, are examples of suggested positive effects of TCIs (26-30).

However, the recent rapid pace of developing new TCIs and testing for their effectiveness diverged the attention from how to implement them in a successful way and guarantee their uptake into routine practice. Although the evidence on the effectiveness of TCIs is encouraging, knowledge on how to implement them in practice to enhance their effectiveness is still scarce.

IMPLEMENTATION OF TRANSITIONAL CARE INNOVATIONS

TCIs are valuable innovations, however, their implementation in long-term care practice remains complex and prone to failure. Establishing the effectiveness of a TCI, like any innovation in a certain healthcare setting, can create evidence to encourage its uptake by individuals yet does not necessarily ensure its continuous and widespread usage (31). Moreover, the successful implementation of a TCI in one setting might not bring about the same result in a different setting. Therefore, the effectiveness of a TCI and its success in avoiding or improving care transitions for older persons is determined by the particular context where it is being implemented. Understanding why TCIs' implementation is often challenging and how to make it successful is still lacking. This merits exploring the context in which the TCIs' implementation is taking place and determining the relevant influencing factors (32).

Context refers to the setting, environment, organization, system, place, or circumstances in which the implementation of an innovation happens. Moreover, some denote context

GENERAL INTRODUCTION

as a "bundle of stimuli" and can signify how easy it is to implement an innovation (33, 34). Therefore, the context is a dynamic medium whereby multiple factors exist and play out as barriers or facilitators to the implementation of an innovation. In addition, the context has several dimensions or domains at which these factors occur such as, but not limited to, the inner setting (organizational structures, culture, processes) or the wider environment (policies, regulations, mandates) (34).

Specifically, the context of transitional care is unique in several ways. First, most TCIs are designed in a way to involve multiple care transition points and organizations, which makes them multifaceted complex innovations (35). Second, TCIs involve at least two care settings (i.e. organizational contexts) which can be at different levels of readiness for implementing innovations. Third, older persons, who are at the core of each TCI, have heterogeneous care needs. For instance, care transitions of older persons with dementia differ from those who suffer from the consequences of heart failure, which in turn, adds to the intricacy of implementing TCIs (36, 37). Some studies explored the factors that influence the implementation of individual TCIs (38, 39). However, insight is lacking on the most common and prominent barriers and facilitators that influence the implementation of various types of TCIs across settings, as well as whether these factors are mostly linked to the TCI's features, characteristics of individuals implementing the TCI, organizational setting, or another contextual dimension, and if any interrelationships between the factors exist. Moreover, there is no consensus on the level of importance of each factor in influencing the implementation of TCIs nor which are the most important ones. Besides, literature generally reports on factors post-implementing TCIs with limited effort to assess the context beforehand and to understand the critical barriers and facilitators within it prior to implementation.

Given the significance of accounting for the context and the innovation's characteristics, several implementation frameworks, models, and theories were developed as useful tools to explain how and why the implementation of innovations into practice succeeds or not (40-42). Accordingly, throughout this dissertation, we opted to use the Consolidated Framework for Implementation Research (CFIR) as a frame of reference to examine the implementation of TCIs and to identify the barriers and facilitators (43), in addition to including other relevant implementation science methods.

In light of the complexity of implementing healthcare innovations such as TCIs, various taxonomies and overviews of implementation strategies described as "methods used to improve adoption, implementation, and sustainment of interventions in healthcare practice" were developed (44). Examples of such strategies include assessing for implementation readiness and identifying contextual factors, involving executive boards, obtaining formal commitments, revising professional roles, conducting training sessions, or using an

implementation specialist (45, 46). Hence, utilizing implementation strategies can potentially address the influencing factors and improve the implementation of TCIs. However, few studies in the literature reported on the selection and use of strategies to implement TCIs.

Though many strategies were proposed and described for implementing healthcare innovations, selecting specific strategies relevant and applicable to TCIs can be more useful. A selection of implementation strategies to address particularly the important factors that influence the implementation of TCIs is still lacking. Moreover, guidance on indications for evidence on the effectiveness of these strategies to enhance implementation is necessary. This contributes to closing the gap existing between developing TCIs and moving them into practice successfully, by increasing the awareness of implementers on prospective challenges in implementing TCIs in advance and offering ways to tackle them.

OBJECTIVES AND OUTLINE OF THE THESIS

This dissertation aims to provide an overarching and thorough insight into the implementation of TCIs. More specifically, to identify the most important factors (barriers, facilitators) that influence the implementation of TCIs and to determine the implementation strategies necessary to address these influencing factors in order to improve the implementation of TCIs in long-term care practice.

Chapter 2 presents a scoping review of the factors that influence the implementation of TCIs for older persons in long-term care settings. **Chapter 3** provides the results of a modified Delphi study conducted with international scientific and practice-based experts to obtain consensus on the relative importance level of factors that influence the implementation of TCIs and the feasibility of addressing them with implementation strategies. Chapter 4 provides the results of a retrospective qualitative collective case study, which explored the implementation of four transitional care innovations for older persons in Belgium, by assessing three implementation aspects: implementation factors, strategies, and outcomes. Chapter 5 provides the results of a qualitative interview study that examined the stakeholders' experiences with and perceptions on four transitional care innovations (the same innovations as studied in chapter 4) implemented within an integrated care project in Belgium. Chapter 6 describes in detail the systematic development of a set of theory and evidence-based implementation strategies selected for TCIs by using Implementation Mapping methodology. A selection of 40 implementation strategies was formulated, and each strategy was presented with a summary of effectiveness supported by theory and/or evidence, practical applications, and the target person/entity. Chapter 7 summarizes the main findings, discusses the methodological strengths and limitations of this research, presents some theoretical considerations, and concludes with implications for both research and practice.

REFERENCES

- 1. WHO. World report on ageing and health [Internet]. Luxembourg: World Health Organization; 2015.
- European Commission, Directorate-General for Employment, Social Affairs and Inclusion. Long-term care report: trends, challenges and opportunities in an ageing society. Volume I. Luxembourg: Publications Office of the European Union; 2021.
- European Commission, Directorate-General for Employment, Social Affairs and Inclusion. Adequate social protection for long-term care needs in an ageing society: report jointly prepared by the Social Protection Committee and the European Commission. Luxembourg: Publications Office of the European Union; 2014.
- 4. Marengoni A, Angleman S, Melis R, Mangialasche F, Karp A, Garmen A, et al. Aging with multimorbidity: a systematic review of the literature. Ageing Res Rev. 2011;10(4):430-9.
- Damiani G, Farelli V, Anselmi A, Sicuro L, Solipaca A, Burgio A, et al. Patterns of long term care in 29 European countries: evidence from an exploratory study. BMC Health Serv Res. 2011;11:316.
- 6. Allen J, Hutchinson AM, Brown R, Livingston PM. Quality care outcomes following transitional care interventions for older people from hospital to home: a systematic review. BMC Health Serv Res. 2014;14:346.
- Coleman EA. Falling through the cracks: challenges and opportunities for improving transitional care for persons with continuous complex care needs. J Am Geriatr Soc. 2003;51(4):549-55.
- 8. Abraham S, Menec V. Transitions between care settings at the end of life among older homecare recipients: a population-based study. Gerontol Geriatr Med. 2016;2:2333721416684400-.
- 9. Callahan CM, Arling G, Tu W, Rosenman MB, Counsell SR, Stump TE, et al. Transitions in care for older adults with and without dementia. J Am Geriatr Soc. 2012;60(5):813-20.
- 10. Coleman EA, Boult C, American Geriatrics Society Health Care Systems C. Improving the quality of transitional care for persons with complex care needs. J Am Geriatr Soc. 2003;51(4):556-7.
- 11. Mansukhani RP, Bridgeman MB, Candelario D, Eckert LJ. Exploring transitional care: evidence-based strategies for improving provider communication and reducing readmissions. Pharm Ther. 2015;40(10):690-4.
- Greysen SR, Hoi-Cheung D, Garcia V, Kessell E, Sarkar U, Goldman L, et al. "Missing pieces"-functional, social, and environmental barriers to recovery for vulnerable older adults transitioning from hospital to home. J Am Geriatr Soc. 2014;62(8):1556-61.
- 13. Hestevik CH, Molin M, Debesay J, Bergland A, Bye A. Older persons' experiences of adapting to daily life at home after hospital discharge: a qualitative metasummary. BMC Health Serv Res. 2019;19(1):224.
- 14. Hirschman KB, Hodgson NA. Evidence-based interventions for transitions in care for individuals living with dementia. Gerontologist. 2018;58(suppl_1):S129-S40.
- Naylor MD, Aiken LH, Kurtzman ET, Olds DM, Hirschman KB. The care span: The importance of transitional care in achieving health reform. Health Aff (Millwood). 2011;30(4):746-54.
- Länsisalmi H, Kivimäki M, Aalto P, Ruoranen R. Innovation in healthcare: a systematic review of recent research. Nurs Sci Q. 2006;19(1):66-72; discussion 65.
- 17. Thoma-Lürken T, Bleijlevens MH, Lexis MA, Hamers JP, de Witte LP. An overview of potential labor-saving and quality-improving innovations in long-term care for older people. J Am Med Dir Assoc. 2015;16(6):482-9.
- Buist Y, Verbeek H, de Boer B, de Bruin SR. Innovating dementia care; implementing characteristics of green care farms in other long-term care settings. Int Psychogeriatr. 2018;30(7):1057-68.
- 19. Naylor MD, Shaid EC, Carpenter D, Gass B, Levine C, Li J, et al. Components of comprehensive and effective transitional care. J Am Geriatr Soc. 2017;65(6):1119-25.
- Hwang U, Dresden SM, Rosenberg MS, Garrido MM, Loo G, Sze J, et al. Geriatric emergency department innovations: transitional care nurses and hospital use. J Am Geriatr Soc. 2018;66(3):459-66.
- 21. Naylor MD, Feldman PH, Keating S, Koren MJ, Kurtzman ET, Maccoy MC, et al. Translating research into practice: transitional care for older adults. J Eval Clin Pract. 2009;15(6):1164-70.
- 22. Couture M, Sasseville M, Gascon V. Facilitators and barriers to implementing transitional care managers within a public health care system. J Gerontol Soc Work. 2016;59(4):364-77.
- Everink IH, van Haastregt JC, Maessen JM, Schols JM, Kempen GI. Process evaluation of an integrated care
 pathway in geriatric rehabilitation for people with complex health problems. BMC Health Serv Res. 2017;17(1):34.
- Plochg T, Delnoij DM, van der Kruk TF, Janmaat TA, Klazinga NS. Intermediate care: for better or worse? Process evaluation of an intermediate care model between a university hospital and a residential home. BMC Health Serv Res. 2005;5:38.
- 25. Enderlin CA, McLeskey N, Rooker JL, Steinhauser C, D'Avolio D, Gusewelle R, et al. Review of current conceptual models and frameworks to guide transitions of care in older adults. Geriatr Nurs. 2013;34(1):47-52.
- 26. Fønss Rasmussen L, Grode LB, Lange J, Barat I, Gregersen M. Impact of transitional care interventions on hospital readmissions in older medical patients: a systematic review. BMJ Open. 2021;11(1):e040057.
- Sezgin D, O'Caoimh R, Liew A, O'Donovan MR, Illario M, Salem MA, et al. The effectiveness of intermediate care including transitional care interventions on function, healthcare utilisation and costs: a scoping review. Eur Geriatr Med. 2020;11(6):961-74.

- Dresden SM, Hwang U, Garrido MM, Sze J, Kang R, Vargas-Torres C, et al. Geriatric Emergency Department Innovations: The Impact of Transitional Care Nurses on 30-day Readmissions for Older Adults. Academic emergency medicine : official journal of the Society for Academic Emergency Medicine. 2020;27(1):43-53.
- Hendrix C, Tepfer S, Forest S, Ziegler K, Fox V, Stein J, et al. Transitional Care Partners: a hospital-to-home support for older adults and their caregivers. J Am Assoc Nurse Pract. 2013;25(8):407-14.
- Lee JY, Yang YS, Cho E. Transitional care from hospital to home for frail older adults: a systematic review and meta-analysis. Geriatr Nurs. 2022;43:64-76.
- 31. Bauer MS, Kirchner J. Implementation science: What is it and why should I care? Psychiatry research. 2020;283:112376.
- Pfadenhauer LM, Gerhardus A, Mozygemba K, Lysdahl KB, Booth A, Hofmann B, et al. Making sense of complexity in context and implementation: the context and implementation of complex interventions (CICI) framework. Implement Sci. 2017;12(1):21.
- Ovretveit JC, Shekelle PG, Dy SM, McDonald KM, Hempel S, Pronovost P, et al. How does context affect interventions to improve patient safety? An assessment of evidence from studies of five patient safety practices and proposals for research. BMJ Qual Saf. 2011;20(7):604-10.
- Nilsen P, Bernhardsson S. Context matters in implementation science: a scoping review of determinant frameworks that describe contextual determinants for implementation outcomes. BMC Health Serv Res. 2019;19(1):189.
- Naylor MD, Bowles KH, McCauley KM, Maccoy MC, Maislin G, Pauly MV, et al. High-value transitional care: translation of research into practice. J Eval Clin Pract. 2013;19(5):727-33.
- Feltner C, Jones CD, Cené CW, Zheng ZJ, Sueta CA, Coker-Schwimmer EJ, et al. Transitional care interventions to prevent readmissions for persons with heart failure: a systematic review and meta-analysis. Ann Intern Med. 2014;160(11):774-84.
- 37. Prusaczyk B, Olsen MA, Carpenter CR, Proctor E. Differences in transitional care provided to patients with and without dementia. J Gerontol Nurs 2019;45(8):15-22.
- 38. Yip O, Huber E, Stenz S, Zullig LL, Zeller A, De Geest SM, et al. A contextual analysis and logic model for integrated care for frail older adults living at home: the INSPIRE project. Int J Integr Care. 2021;21(2):9.
- McAiney C, Markle-Reid M, Ganann R, Whitmore C, Valaitis R, Urajnik DJ, et al. Implementation of the community assets supporting transitions (CAST) transitional care intervention for older adults with multimorbidity and depressive symptoms: a qualitative descriptive study. PLoS One. 2022;17(8):e0271500.
- 40. Grol R, Wensing M. What drives change? Barriers to and incentives for achieving evidence-based practice. Med J Aust. 2004;180:S57-S60.
- Grol R, Wensing M, Bosch M, Hulscher M, Eccles M. Theories on implementation of change in healthcare. In: Grol R, Wensing M, Eccles M, Davis D, editors. Improving patient care: the implementation of change in health care 2nd ed. United Kingdom2013. p. 18-39.
- 42. Nilsen P. Making sense of implementation theories, models and frameworks. Implement Sci. 2015;10:53.
- Damschroder LJ, Aron DC, Keith RE, Kirsh SR, Alexander JA, Lowery JC. Fostering implementation of health services research findings into practice: a consolidated framework for advancing implementation science. Implement Sci. 2009;4:50.
- Powell BJ, Fernandez ME, Williams NJ, Aarons GA, Beidas RS, Lewis CC, et al. Enhancing the impact of implementation strategies in healthcare: a research agenda. Front Public Health. 2019;7.
- Cochrane. Effective Practice and Organisation of Care (EPOC). EPOC Taxonomy; 2015 [Available from: epoc. cochrane.org/epoc-taxonomy]
- Powell BJ, Waltz TJ, Chinman MJ, Damschroder LJ, Smith JL, Matthieu MM, et al. A refined compilation of implementation strategies: results from the Expert Recommendations for Implementing Change (ERIC) project. Implement Sci. 2015;10:21.



2

CHAPTER

A Myriad of Factors Influencing the Implementation of Transitional Care Innovations: A Scoping Review

Published as:

Fakha A, Groenvynck L, de Boer B, van Achterberg T, Hamers J, Verbeek H. A myriad of factors influencing the implementation of transitional care innovations: a scoping review. Implement Sci. 2021; 16(1):21. https://doi.org/10.1186/s13012-021-01087-2

ABSTRACT

BACKGROUND

Care transitions of older persons between multiple care settings are frequently hampered by various issues such as discontinuous care delivery or poor information transfer among healthcare providers. Therefore, several innovations have been developed to optimize transitional care (TC). This review aims to identify which factors influence the implementation of TC innovations.

METHODS

As part of TRANS-SENIOR, an international innovative training and research network focusing on enhancing or avoiding care transitions, a scoping review was conducted. The five stages of the Arksey and O'Malley framework were followed. PubMed/MEDLINE, EMBASE, and CINAHL were searched, and eligible studies published between years 2000 and 2020 were retrieved. Data were extracted from the included studies and mapped to the domains and constructs of the Consolidated Framework for Implementation Research (CFIR) and Care Transitions Framework (CTF).

RESULTS

Of 1,537 studies identified, 21 were included. Twenty different TC innovations were covered, and aimed at improving or preventing transitions between multiple care settings, the majority focused on transitions from hospital to home. Key components of the innovations encompassed transition nurses, teach-back methods, follow-up home visits, partnerships with community services, and transfer units. Twenty-five prominent implementation factors (seven barriers, seven facilitators, and eleven factors with equivalent hindering/facilitating influence) were shown to affect the implementation of TC innovations. Low organizational readiness for implementation, and the overall implementation climate were topmost hindering factors. Similarly, failing to target the right population group was commonly reported as a major barrier. Moreover, the presence of skilled users but with restricted knowledge and mixed attitudes about the innovation impeded its implementation. Among the eminent enabling factors, a high-perceived advantage of the innovation by staff, along with encouraging transition roles, and a continuous monitoring process facilitated the implementation of several innovations. Other important factors were a high degree of organizational networks, engaging activities, and culture; these factors had an almost equivalent hindering/facilitating influence.

CONCLUSIONS

Addressing the right target population and instituting transition roles in care settings appear to be specific factors to consider during the implementation of TC innovations. Long-term care settings should simultaneously emphasize their organizational readiness for implementation and change, in order to improve transitional care through innovations.

KEYWORDS

Implementation, innovation, care transitions, transitional care, long-term care, factors, older persons.

Contributions to the literature

- Our study identifies a set of significant factors that influence the implementation of innovations specific to transitional care, which diminishes the existing gap in implementation literature and offers guidance to long-term care organizations in future endeavors for enhancing this type of care for older persons.
- The current findings provide a dynamic and different perspective by addressing the interorganizational aspect of implementing transitional care innovations across multiple long-term care settings.
- The methodology used illustrates the possibility of combining multiple implementation research frameworks to enable a rich and comprehensive study of the influencing factors on implementing transitional care innovations.

BACKGROUND

Innovations in transitional care (TC) are often implemented to ensure an optimal continuity of healthcare delivery for older persons who transfer between multiple care settings. Older persons aged 65 years and above are at high risk of adverse events during care transitions due to the prevalence of chronic diseases and multimorbidity (1-7). Care transitions of older persons are frequently hampered by a diversity of issues, such as, but not limited to, fragmented care, medication errors, or poor communication among healthcare providers (7,8). Consequently, the delivery of proper TC for the older population is not always achieved.

There appears to be an urgent need to innovate in order to alleviate the augmented demand for long-term care (LTC) services and promote better and safer care transitions. Based on the World Health Organization's concept of LTC, we adapted its definition to fit the use throughout this article as "LTC refers to the provision of continuous care activities performed by formal and/or informal/family caregivers to ensure that older persons with or at risk of a significant ongoing loss of intrinsic capacity can maintain a level of functional ability consistent with their basic rights, fundamental freedoms, and human dignity; also it can be achieved through: (a) optimizing the older person's trajectory of intrinsic capacity, (b) compensating for a loss of capacity by providing the environmental support and care necessary to maintain functional ability at a level that ensures well-being; and can be provided in settings, such as but not limited to: nursing and residential care facilities, assisted living facilities, or private/own home" (9). To that end, multiple evidence-based TC interventions, models, or programs also referred to as "innovations" have been developed with the goal to improve or prevent transitions between different settings (2). According to existing literature, we defined the following terms to be used throughout this article: 'improve care transitions' - to provide and enhance the transitional care and services delivered during physical relocations of older persons from one care setting to another, with a view to creating optimal benefit as a result of the care transition; 'prevent care transitions' - to provide the care and services needed in order to avert an unnecessary or avoidable physical movement of older persons between two care settings or more (2, 5, 7). The Transitional Care Model and Coleman's Care Transitions Intervention, are examples of interventions designed to improve care transitions from hospital to home (2). Key components of these interventions include appointing a transition coach or nurse, encouraging patient self-management, and planning hospital discharge (10-12). While other interventions (13) aim to prevent care transitions from nursing home to hospital through the use of specific advanced care planning tools, alternative interventions focus on providing acute care at home to prevent transitions from home to hospital (13, 14). The successful implementation of these interventions has been shown to enhance the quality of care, control costs, reduce hospital readmission rates, and ultimately meet patient needs (2, 15). However, while innovation in TC is encouraged as a solution, its implementation is often difficult and unsuccessful.

The success or failure of the implementation of any innovation within a healthcare setting is usually influenced by multiple factors recognized as either barriers or facilitators (16). These factors can be linked to either the innovation characteristics, individual professionals, patients and caregivers, organizational structure, or the environmental context (16, 17). Nevertheless, other factors related to the actual process and activities undertaken to implement an innovation such as the planning, execution, and evaluation methods are as crucial (17). Similarly, attempts to implement innovations in TC are frequently affected by multiple factors. Amongst the barriers are limited organizational resources, absence of an implementation climate, complexity of the innovations, and low leadership engagement (18, 19). Conversely, facilitators include the adaptability of innovations, a high relative advantage of the innovation as perceived by users, and the existence of robust external organizational partnerships (14, 19).

However, to the best of our knowledge, no overview exists on barriers and facilitators that influence the implementation of innovations for preventing or improving care transitions for older persons. Thus, there is a need to explore and map the available evidence on these implementation factors. The main research question of the current study is: What are the barriers and facilitators that influence the implementation of TC innovations for older persons in long-term care settings? A secondary question is whether the literature captured the perspectives of older persons and informal or family caregivers on the innovation's implementation and overall experience; and if so, what was reported as feedback.

METHODS

This scoping review follows the Preferred Reporting Items for Systematic Reviews and Meta-Analyses extension for Scoping Reviews (PRISMA-ScR) checklist (20). The review was conducted according to the five stages described by the Arksey and O'Malley framework (21) and the enhancements proposed by Levac *et al.* (22).

STAGE 1: IDENTIFYING THE RESEARCH QUESTION

This scoping review is guided by the question: "What are the barriers and facilitators that influence the implementation of TC innovations for older persons in long-term care settings?

STAGE 2: IDENTIFYING RELEVANT STUDIES

Initially on July 25, 2019, a systematic search of three databases was conducted: PubMed/ MEDLINE, EMBASE, and CINAHL; an update was run on March 10, 2020. Four main concept terms were used in the search: implementation; care transition; innovation; and older persons. To formulate the search strings, relevant keywords and synonyms were identified for each concept term in addition to the controlled vocabulary terms (such as MeSH headings in MEDLINE/PubMed). The search strategy was discussed by the authors as well as reviewed by an information specialist. Reference lists of articles that fulfilled the inclusion criteria were searched to identify additional papers. The final search strategy is available (see Appendix 2A).

STAGE 3: STUDY SELECTION

Literature published in any language between January 1, 2000 and March 10, 2020 was retrieved. Original research studies were included. Articles were eligible for inclusion if : a) target population (participants or receiver of care) were all, or if the majority were older persons aged 65 years and above (also referred to as patients, older adults, frail older adults, elderly) with long-term care needs and at risk of care transitions; b) focused on the transfer and physical movement of older persons between two or more care settings with at least one setting providing long-term care; c) implemented an innovation within a care setting to prevent or improve care transitions; d) reported on the barriers and facilitators that influenced the implementation process of the innovation; e) stated the perspectives of the older persons, family, informal caregivers, and/or healthcare providers on the innovation. After the removal of duplicates, the first author (AF) screened the titles and abstracts for eligibility. In order to increase reliability, the second author (LG) screened a random selection of 10% of the total records for titles and abstracts (23). Both reviewers then compared their assessment decisions and resolved any differences through discussion and when necessary through consultation with the author (BdB). In the next phase, the two authors (AF; LG) independently screened and discussed 100% of the full texts of those articles deemed eligible (23, 24). The Preferred Reporting Items for Systematic Reviews and Meta-analyses (PRISMA) flowchart (25) was used to report the study selection process.

STAGE 4: CHARTING THE DATA

a. <u>Development of the data charting forms:</u>

A data charting form consisting of two parts was developed. *Data charting form – part 1* comprised the following: title; authors; year; country; study aim; design and methodology; population; setting; innovation description; duration and phase of implementation; presence of barriers and/or facilitators to innovation implementation; reported themes of barriers and/ or facilitators to the implementation of the innovation; perspectives of older persons, family, or informal caregivers and/or providers on the innovation; and reported implications of the innovation. *Data charting form – part 2* was devised to map barriers and facilitators as identified in the studies to the Consolidated Framework for Implementation Research (CFIR) (26) and the Care Transitions Framework (CTF) (27).

The CFIR is composed of five domains: *i) intervention characteristics; ii) outer setting; iii) inner setting; iv) characteristics of individuals; v) process;* and 39 standardized constructs and subconstructs (26). This framework helps researchers identify the factors (i.e., barriers and facilitators) that influence the implementation of innovations (28). Moreover, specific constructs from the CTF (27) were selected and used in supplement to the CFIR (see

Appendix 2B). The CTF is an adaptation of the CFIR, whereby it incorporates all the CFIR constructs in addition to new ones, which are mostly relevant to transitional care.

b. <u>Testing of data charting forms and the charting process:</u>

Both forms were tested initially on two articles, and then results were discussed critically within the research team. It was agreed to include additional elements to describe further the innovations' features in data charting form 1. In the data charting, the implementation factors and themes were extracted from the included articles and then mapped to the CFIR's relevant domains, constructs, and the selected CTF constructs using the CFIR codebook (29). Subsequently, the CFIR rating rules were used to determine each factor's influence as negative: a barrier, or positive: a facilitator (30). Two authors (AF; TvA) charted data independently from five randomly selected articles. Disagreements on mapping factors to CFIR/CTF constructs were resolved between the two authors leading to a consensus. Afterwards, author AF completed the full data charting for all the included articles.

STAGE 5: COLLATING, SUMMARIZING, AND REPORTING THE RESULTS

The data charted were synthesized as follows:

- a. Description of included studies: classification of the studies into four groups according to the care transition pathways of each TC innovation; included the author(s), year of publication, country, objective, population, design, and methods.
- b. Description of the TC innovations: classification of the innovations into four groups according to the specific care transition pathways; included the target population, key components, and the CFIR domains influencing their implementation.
- c. Barriers and facilitators to implementation of TC innovations: the frequency of the reported factors identified as barriers and/or facilitators to the implementation was calculated based on their presence in the number of studies.
- d. Perspectives of older persons, family, or informal caregivers: a narrative description of the feedback on the overall experience, satisfaction with, or views on the implementation of the TC innovation.

RESULTS

STUDY SELECTION

Initially, 1,537 studies were identified, and 21 were included in the final stage. The flowchart for the selection process is depicted in Figure 1.



Figure 1. PRISMA flowchart of study selection process

STUDY CHARACTERISTICS

The 21 studies included described the implementation of 20 different TC innovations (see Table 1). Almost half of the studies (N=11, 52%) originated from the USA, and five were from Europe. The majority of the studies were process evaluations and were performed during or post the implementation of a TC innovation to examine the influencing factors. Most studies used qualitative research methods, and 11 utilized a preselected evaluation, implementation, or quality-related framework, tool, model, or instrument to guide data collection such as interviews and/or data analysis.

Study populations across all studies were comprised of multiple healthcare professionals and providers. Only six studies included older persons or family/informal caregivers and explored their perspectives on the TC innovations (36, 37, 41, 44, 46, 47).

KEY FEATURES OF THE TC INNOVATIONS

Sixteen innovations focused on improving care transitions for older persons, while four focused on preventing transitions. TC innovations were classified into groups according to the care transition pathways (see Table 2).

DESCRIPTION OF THE FOUR GROUPS OF TC INNOVATIONS

Care transitions from hospital to home settings were the focus of ten TC innovations. Improving care transitions was the main aim of these innovations with common goals to reduce hospital readmissions, lower healthcare costs (31, 34-36, 39, 40), enhance older persons' quality of life (18, 32, 33) and satisfaction (34, 39), and scale down the need for institutional care (38). Mostly, these innovations targeted older persons with chronic and complex conditions discharged from hospital, requiring long-term care at home, and who were at higher risk of readmission. The common component across the innovations was the existence of a healthcare professional with a "transition role," such as a transition nurse, health coach, care coordinator, social worker, or community nurse. The role served to ameliorate the transition journey from hospital to home by primarily providing follow-up, developing individualized care plans, and coordinating care.

Care transitions from hospital to intermediary care places to a final destination were the focus of four TC innovations. These innovations aimed to improve care transitions with common objectives, such as reducing the length of hospital stays, relieving hospital bedblocking, and preventing inappropriate admission to residential aged care (42, 43). All four innovations were designed for older persons who concluded an episode of acute care at hospital but were unfit to transfer to home or another final long-term care destination. The creation of "transition intermediary care places," such as transfer beds hosted within a residential care facility or community setting was the notable component across these innovations (41-44). Hence, the four TC innovations allowed extra time to organize a more personalized arrangement for the long-term care final destination for older persons.

Care transitions from hospital or home to nursing/residential care facility were the focus of two TC innovations. The goal of these innovations was to improve care transitions with the objective to enhance information transfer between hospitals and nursing facilities and promote continuity of care. The essential aspect of both innovations was the provision of "transition advice & support" to nursing facility staff. This was enabled through the arrangement of community geriatric services and a psychiatric community nurse (45, 46).

Table 1. Characteris	Lable 1. Characteristics of the ZI included studies	TC innovation name	Ctudu nanulation tatal campledal	Docion 8 mothods
Autnor(s), year or Objective publication, country collection	Objective & timing of data collection		study population, total sample(n)	Design & methods
Group I – Studies fo	Group I – Studies focused on TC innovations to improve care transitions from hospital to home	care transitions from hospital to	home	
Bradway et al. 2012, USA (31)	To describe the barriers and facilitators to implementing a transitional care intervention for cognitively impaired older adults & their caregivers led by advanced	APN-directed TCM Advanced Practice Nurse- Directed Transitional Care Model	Healthcare professionals (n=3): • Advanced Practice Nurses	 Exploratory qualitative: Case summaries for patient-caregiver dyads completed by APNs at end of intervention – 15 randomly selected Field notes taken by study investigators
	practice nurses post-implementation			during biweekly APNs' case conferences
Couture et al. 2016, Canada (18)	To evaluate and explain the barriers and facilitators to	TCM Role	Healthcare professionals (n=29): • Transitional care managers	Process evaluation: Assessed the fidelity, acceptability, &
	the implementation of a pilot intervention – introducing the role of Transitional Care Managers within a public healthcare system	Transitional Care Manager Role	 Hospital workers Social workers Staff of health and social service centers 	appropriateness (contextual factors) • Focus groups • Direct observations • Direct observations • Meeting minutes & documents of
	 during implementation 			coordinating and implementation committee
Hung et al. 2015, USA (32)	To examine and describe the successes and challenges of the implementation of a pilot	Community- based TCP	Healthcare professionals & program's management team (n=7): • Interprofessional team of program	 Qualitative: Semi-structured interviews guided by the Organizational Readiness to Change
	community-based transitional care program	Community-based Transitional Care Program	 staff (nurses, social workers) Members of program's steering committee 	Assessment instrument Analysis using PARIHS framework (contextual factors, evidence, facilitation)
	post-implementation			techniques)
Hung et al. 2018, USA (33)	To examine the key contextual features enabling the	Community-based TCP	Healthcare professionals & program's management team (n=17):	Qualitative: Semi-structured interviews guided by
	implementation and hospital-wide scaling of a community-based transitional care program	Community-based Transitional Care Program	 Program director and manager Program staff (nurses, social workers, wellness coaches) 	the Organizational Readiness to Change Assessment instrument • Analysis using Care Transitions Framework
	V post-implementation		Members of program's steering committee	(domains – intervention, organizational, & patient characteristics, implementation process, measures, & outcomes)

Table 1. Characteristics of the 21 included studies

Table 1. Continued				
Author(s), year of publication, country	Objective & timing of data collection	TC innovation name	Study population, total sample(n)	Design & methods
McNeil et al. 2016, Canada (34)	To evaluate the effectiveness and identify barriers to and facilitators of the implementation of an intervention involving patient handoff between a hospital-based care transitions nurse and a community-based response nurse	CTI-Handoff Care Transition Intervention Handoff	 Care transitions & rapid response nurses (n= not specified) Managers & executives (n= 4) 	Qualitative:Focus group discussionsIndividual interviews
Naylor et al. 2009, USA (35)	 post-implementation To identify the major facilitators and barriers to implementing the Transitional Care Model in an insurance organization over the phases of start-up & roll-out P pre- & during implementation 	TCM Transitional Care Model	Healthcare professionals and project team (n=19). Implementation staff Case managers Transitional care nurses	 Qualitative: Semi-structured interviews guided by Everett Roger's framework for diffusion of innovations (focus on: staff involvement, cinture, communication channels, model integration within organization, ease/ difficulty of start-up phase)
Nurjono et al. 2019, Singapore (36)	To evaluate the implementation fidelity of a transitional care program	NUHS-RHS TCP National University Health System-Regional Health Systems Transitional Care Program	Healthcare professionals (n=25): • Care coordinators, program managers, physicians • Family caregivers (n=45)	 Realist evaluation: Using the Conceptual Framework of Implementation Fide/lity (moderating factors: context, participant responseveness, program complexity, facilitating strategies, recruitment) Interviews Observations Reviews of medical records & program databases
Parrish et al. 2009, USA (37)	To identify factors that promote the sustainability of the implementation of a care transition intervention pilot post-implementation	cfi Care Transitions Intervention	 Plidt project team members for 10 sites (n= 30-40)', including Transition coaches (n= at least 10)' Patients (n=791, out of which 69.4% are aged 66+) *estimations 	 Mixed methods: Surveys Surveys Interviews Interviews Final project narrative reports, data reports Comparison of pre-& post-project sustainability plans Svariables used to assess sustainability factors (leadership, transition coaches staff, project management, team commitment, sustainability plan) Care transition measure (CTM) to assess quality of assess level of patient activation in the 4 pillars of the CTI

Table 1. Continued

σ
(۱)
Ť
2
. <u> </u>
Ę
5
U
<u>-</u>
Ð
P
Ца

Author(s), year of publication, country	Objective & timing of data collection	TC innovation name	Study population, total sample(n)	Design & methods
Rosstad et al. 2015, Norway (38)	To investigate the implementation process of a care pathway for elderly patients into the daily practice of healthcare professionals a during & post-implementation	Patient Trajectory for Home- dwelling elders	Healthcare professionals (n=60): • Home care managers • Nurses • Nursing assistants	 Comparative qualitative process evaluation: Semi-structured interviews guided by questions focused on staff involvement & expectations, care pathway introduction & efforts to use it, challenges, promoting factors, benefits, sustainability Focus group discussions Field notes (by study investigator on overall implementation process) Meeting minutes (conference calls with head nurses & home care managers performed by study investigator) Analysis guided by the Normalization Process Theory core constructs (coherence, cognitive porticipation, reflexive monitoring)
Williams et al. 2014, USA (39)	To examine the successes and failures experienced by implementing Project BOOST aiming to enhance transition from hospital to home P during implementation in cohorts & post-implementation in pilot sites	Better Outcomes by Optimizing Safe Transitions	 Local team leaders of (n= 6 pilot hospital sites) Hospital team leaders of (n= 27 cohort sites) 	 Evaluation: Telephone interviews guided by basic implementation questions & opinions on intervention elements in pilot sites Survey with hospital cohorts on: implementation occurrence, when, and how
Xiang et al. 2018, USA (40)	To examine the experiences of community-based organizations implementing the Bridge Model of Transitional care, and identify facilitators and barriers associated with the implementation and sustainability of the model V during & post-implementation	The Bridge Model	Healthcare professionals (n=13): • Clinical supervisors • Program coordinators	 Qualitative case study: Semi-structured interviews by telephone, guided by 3 domains of successful implementation of the PARIHS framework fevidence, context, leadership, evaluation, facilitation)

Author(s), year of Objective publication, country collection	Objective & timing of data collection	TC innovation name	Study population, total sample(n)	Design & methods
Group II – Studies fo destination	cused on TC innovations to <u>improve</u>	care transitions from <u>hospital</u>	to intermediary care places (residenti	Group II – Studies focused on TC innovations to improve care transitions from hospital to intermediary care places (residential care or rehabilitation facility) to a final destination
Everink et al. 2017, the Netherlands (41)	To evaluate the feasibility of implementing an Integrated Care	ICP Geri-Rehab	Healthcare professionals (n=19): • Elderly care physicians	 Process evaluation: using Saunders & colleagues framework
	Pathway in Geriatric Rehabilitation	Integrated Care Pathway in Geriatric Rehabilitation for	 Nurses (specialized, discharge) Physiotherapists 	(fidelity, dose delivered, satisfaction, contextual & external factors)
	V post-implementation	People with Complex Health Problems	 Professionals from home care organizations 	 Semi-structured group interviews Face-to-face interviews Self-administered questionnaires
			 Informal caregivers (n=37) Patients (n=113) 	Patient files Meeting minutes
Masters et al. 2008,	To examine reports from providers	TC Places	Organizations providing transition	Qualitative:
(42)	to reveal enablers and partiels to compliance with the key requirements of a transition care program	Transition Care Places	care services (n=23 organizations)	 Content analysis of quality sett-reports
	 post-implementation 			
Plochg et al. 2005, the Netherlands (43)	To assess the functioning and implementation of an intermediate care model between a hospital and a residential home	ICM Intermediate Care Model	Healthcare professionals and management staff from the residential home & medical center (n=21);	 Process evaluation: Semi-structured interviews Analysis using a typology of quality systems based on 5 elements (structural
	V post-implementation		 Leadership (general manager, director, head of care department, chair board of directors, chair of medical specialists) Physicians Nurses (liaison, geriatric nurse specialist, liaison nurse-head of discharge unit, registered nurse- head of transfer unit) Occupational / physical therapists 	assets, allocation of responsibilities, protocols, information transfer, and monitoring/feedback cycles), and Grol's model on effective implementation

;

hor(s), year of lication, country ehan et al. 2013, tralia	Objective & timing of data	TC innovation name	Study population, total sample(n)	Design & methods
Renehan et al. 2013, Australia			A	
(44)	To evaluate the implementation and effectiveness of a Transition Care Cognitive Assessment and Management Pilot > post-implementation	TC CAMP Transition Care Cognitive Assessment and Management Pilot	 Healthcare professionals (n=17): TC CAMP facility and health services staff (nursing, management, allied health, team management, allied health, team consultant) Unit managers of final destination facility 	Process and outcome evaluation: • Structured interviews • Focus group discussions • Medical records file audits
Ground III Churding formand on TC	ourond on TC innotations to innervo	a latinad martinad transformed	o Family caregivers (n=/) incorretions to immove constituent from from the most of musical constant and for illev	cities.
ol oup III - aluais				
Sutton et al. 2016, UK (45)	To characterize the challenges experienced in a quality improvement project aiming to improve transitions for older people > post-implementation	alP-TC Quality Improvement Project to Improve Transitions of Care for Older People	 Healthcare professionals & project team (n=12): Care home staff (managers, owners, assistants, frontline) Hospital-based staff Project staff 	Ethnographic process evaluation: • Observations • Semi-structured interviews • Project documents (progress reports, meeting minutes)
Van Mierlo et al. 2015, the Netherlands (46)	To evaluate a mental healthcare transfer intervention after the movement of a person with dementia into a nursing home, and to investigate factors that influence its successful implementation > post-implementation	CPN Follow-up visit by Community Psychiatric Nurse	 Healthcare professionals (n=28): Professional nursing home carers Community psychiatric nurses Nursing home managers Outpatient clinic managers General practitioners Team manager – center for people with dementia Healthcare insurer Family caregivers (n=5) 	 Evaluation: Semi-structured interviews based on the Theoretical Model of Adoptive the Theoretical Model of Adoptive function (external factors, different phases & levels of implementation process)
Group IV – Studies focused on TC	cused on TC innovations to prevent	innovations to prevent care transitions from nursing facility or home to hospital	ncility or home to hospital	
Brody et al. 2019, USA (14)	To examine the barriers to and facilitators of the implementation of Hospital at Home Plus 30 days of transitional care program during its first year of operation	HaH Plus program Hospital at Home Plus Program	Healthcare professionals (n=27): • Team physicians • Nurse practitioners/leaders • Social workers • Leadership staff • Executives • Home health agency staff	Qualitative: • Primers to help recall of key events/factors & develop interview guide • Semi-structured interviews • Focus group discussions

Table 1. Continued

Table 1. Continued				
Author(s), year of publication, country	Objective & timing of data collection	TC innovation name	Study population, total sample(n)	Design & methods
Ersek et al. 2018, USA (47)	To explore the stakeholders' perspectives on the implementation of OPTIMISTIC program, which aims to reduce hospitalizations from nursing facility, specifically the program's effective components, facilitating adoption features, and barriers to implementation	OPTIMISTIC Optimizing Patient Transfers, Impacting Medical Quality, and Improving Symptoms: Transforming Institutional Care project	 Healthcare professionals (n=53): Primary care providers Nursing home staff & leadership OPTIMISTIC clinical staff Pamily members of nursing home residents (n=10) 	 Evaluation: Using Statler framework of formative evaluation Semi-structured group and individual interviews
Hirschman et al. 2017, USA (48)	• during implementation To describe the experiences of healthcare providers involved in adapting and testing the feasibility of implementing a care innovation by combining two models: the patient-centered medical home and the transitional care model	PCMH+TCM Patient-Centered Medical Home + Transitional Care Model	 Transitional care nurses (n=2) Clinicians (n= 2.4/site', 5 sites) *estimations 	Qualitative: • Surveys (open-ended questions)
Rask et al. 2017, USA (49)	 post-implementation To identify contextual and implementation factors impacting the effectiveness of an organizational-level intervention to reduce preventable hospital readmissions from affiliated skilled nursing facilities (SNFs) post-implementation 	INTERACT II Interventions to Reduce Acute Care Transfers	 Quality Improvement Organization Evaluation: staff (n= 4)* Leaders & nurses of SNF Leaders & nurses of SNF Leaders & nurses of SNF Ne staff (n= 2-3/facility)* in Quality organizations *estimations 	Evaluation: Interviews with open-ended questions based on contextual factors' domains of the Model for Understanding Success in Quality tool (external environment, organization, QI support and capacity, microsystem, miscellaneous)

Table 1. Continued

TC innovation name	Target population	TC innovation – Aims & Key components	CFIR domains associated
			with the TC innovation's implementation
Group I: 10 TC innovations to impro	<u>ove care transitions from hospital to home</u>		
APN-directed TCM Advanced Practice Nurse-directed Transitional Care Model	 265yr older adults, hospitalized, cognitively impaired Presence of a family careoiver (CG) 	Aims: to improve patient outcomes and ensure a safe & timely transition	 Intervention characteristics Outer setting Characteristics of individuals
Bradway et al. 2012, USA (31)		 Advanced Practice Nurse, role: daily hospital visits to patient–CG dyad home (or SNF) visits* within 24 hrs post-discharge, a 	<pre> Process </pre>
		minimum of 4 o telephone follow-up & support o development of individualized care plans, patient-	
		CG goals o implementation of risk reduction strategies to minimized of constitution immediates	
		 coordination with a multidisciplinary local team of healthcare experts 	
		 building CG ability to identify early symptoms & apply strategies to prevent poor outcomes 	
		*sometimes patients are admitted to (SNF) skilled nursing facility prior to going home, they receive visits in both settings	
TCM Role Transitional Care Manager Role	 ≥ 70yr older adults, and/or chronically ill Being discharged from hospital, or end of 	Aims: to improve existing discharge planning practices Intervention characteristics Inner setting	 Intervention characteristics Inner setting
Couture et al. 2016, Canada (18)	acuté care is predictable	 Transitional Care Manager (social worker, or any other healthcare professional, except nurses), a liaison agent role:)
		 o improvement of discharge planning by management of environmental & community barriers o exchange of patient information between providers o coordination of care & problem-solving of transitional care 	

Table 2. Description and key features of the 20 TC innovations
TC innovation name	Target population	TC innovation – Aims & Key components	CFIR domains associated with the TC innovation's implementation
Community-based TCP Community-based Transitional Care Program	 265yr older adults About to be discharged from hospital At high risk of readmission 	Aims: to reduce preventable hospital readmissions and improve patient's quality of life at home & in the community	 Intervention characteristics Outer setting Inner setting Construction of individuals
Hung et al. 2015/2018, USA (32, 33)		 Health coach (nurse or social worker), role: o home visits (within 24-48hrs) post-discharge, follow- up phone calls & appointments with primary care providers Discharge planning using 'teach-back' methods resources Support system network Augment date planning Wellness coach up to 6 months 	 Characteristics of individuals Process
CTI-Handoff Care Transition Intervention Handoff	 Frail older adults with complex conditions Discharged from hospital & require home 	 Aims: to reduce redmissions, improve information transfer, & enhance patient satisfaction 	 Intervention characteristics Inner setting Characteristics of inclividuals
McNeil et al. 2016, Canada (34)		 Patient care handoff between Hospital Care Transition Nurse & Community Rapid Response Nurse Home care & follow-up period up to 30 days Referral to hospital-based chronic disease management clinics 	 Process
TCM Transitional Care Model	 Chronically ill, high-risk older adults Hospitalized with multiple chronic conditions 	 Alms: to improve patient outcomes, reduce readmissions, & reduce healthcare costs 	 Intervention characteristics Outer setting Intervention
Naylor et al. 2009, USA (35)		 Transitional Care Nurse, role: o primary care coordinator among providers & ensuring a multidisciplinary approach with open communication o in-hospital patient case assessment & development of care plan o regular home visits & ongoing telephone support (7 days/week over 2 months post-discharge) o continuity of medical care with hospital/primary care & accompanying patients on follow-up visits Early identification & response to health risks Active engagement of patients & their family/informal caregivers by focusing on education and support 	

Tc innovation name Target population Tc innovation - Aims & Key components NUHS-RHS TCP Older adults, and/or with complex National University Health System-Regional Health Nurjono et al. 2019, Singapore (36) A dims: to improve quality of care, reduce hospital transitions intervention A dims: to endorator, an integrator role: o promotion of self-care o robital vist. CrI Carl Carl Carl Carl Strequining long-term carls A dims: to endorator with a network of medical & social o robital vist. A dims: to endorator of personalized care o robital vist. A dims: to endorator of personalized care o robital vist. A dims: to endorator of the social o robital vist. A dims: to endorator of the social o robital vist. A dims: to endorator of the social o robital vist. CrI Carl Carl Carl Carl Carl Carl Carl Carl				
 Older adults, and/or with complex healthcare needs Frequent admitters to hospital Have limited ambulation & caregivers at home Oder adults, in hospital for chronic disease, by & requiring long-term care 	TC innovation name	Target population	TC innovation – Aims & Key components	CFIR domains associated with the TC innovation's implementation
 Oder adults, in hospital for chronic disease, & requiring long-term care 	NUHS-RHS TCP National University Health System-Regional Health Systems Transitional Care Program Nurjono et al. 2019, Singapore (36)	 Older adults, and/or with complex healthcare needs Frequent admitters to hospital Have limited ambulation & caregivers at home 	 A Aims: to improve quality of care, reduce hospital utilizations, & reduce healthcare related costs utilizations, & reduce healthcare related costs Care Coordinator, an integrator role: Care Coordinator, an integrator role: home visits, telephone monitoring home visits, telephone monitoring needs & home environment assessment o development of personalized care o promotion of self-care care coordination with a network of medical & social care providers in/out of hospital 	 Intervention characteristics Outer setting Inner setting Characteristics of individuals Process
appointments	CTI Care Transitions Intervention Parrish et al. 2009, USA (37)	 Oder adults, in hospital for chronic disease, & requiring long-term care 	 A lims: to enhance patient safety during transitions 4-week intervention Transition Coach (nurse or social worker), role: o hospital visit o hospital visit o 1 home visit (24–72hrs post-discharge) o 3 follow-up phone calls Improvement of patient's capacity: o medication self-management o using a patient-centered health record o knowledge of "read flags" o making primary care provider/specialist 	 Inner setting Process

TC innovation name	Target population	TC innovation – Aims & Key components	CFIR domains associated with the TC innovation's implementation
Patient Trajectory for Home- dwelling elders Rosstad et al. 2015, Norway (38)	 Elderly patients requiring home care services after discharge from the hospital 	 A Aims: to improve continuity of care & reduce the need of institutional care Continuity of care from hospital & follow-up of home care recipients Exchange of patient discharge information between the hospital, local health care allocations (municipality-level), & home care services; o local health care allocations office evaluate & decide on care assistance o home care services comprehensive patient assessment within 3 days upon discharge o general practitioner consults patient 14 days post-discharge o district nurse/nursing assistant performs extended assessment during the first 4 weeks Communication among services through a patient daily 	 Intervention characteristics Inner setting Characteristics of individuals Process
BOOST Better Outcomes by Optimizing Safe Transitions	 Older adults At high-risk of adverse events post-hospital discharge 	Aims: to improve patient's discharge & reduce errors, reduce 30 day readmission rates, & improve patient satisfaction	 Intervention characteristics Inner setting Process
Williams et al. 2014, USA (39)		 Comprehensive intervention toolkit for clinical teams: risk assessment risk assessment patient/caregiver education tools teach back o discharge summary o follow-up phone call within 72 hrs follow-up phone call within 72 hrs 	

TC innovation name	Target population	TC innovation – Aims & Key components	CFIR domains associated with the TC innovation's implementation
The Bridge Model Xiang et al. 2018, USA (40)	 Older adults with complex care needs Discharged from an inpatient hospital stay At risk of readmission due to psychosocial 	Aims: to improve care transition and prevent readmission by addressing the psychosocial determinants	 Intervention characteristics Outer setting Inner setting
	determinants	 Bridge Care Coordinator (social worker), role: hospital visits hospychosocial needs assessment & development of a care plan care coordination & follow-up in person or by telephone throughout 30 days post-discharge Collaboration of hospital & community-based organizations for aging services 	- Todess
Group II: 4 TC innovations to impro	<u>ove care transitions from hospital to intermedia</u>	ove care transitions from hospital to intermediary care places (residential care or rehabilitation facility) to a final destination	o a final destination
ICP Geri-Rehab Integrated Care Pathway in Geriatric Rehabilitation for People with Complex Health Problems	 ≥ 65yr frail older adults with complex health problems Previously admitted to hospital & geriatric rehabilitation care 	Aims: to improve communication between healthcare providers & enhance the triage process during transitions	 Intervention characteristics Outer setting Inner setting Process
Everink et al. 2017, the Netherlands (41)	- 1	 Triage instrument for intermediary Geriatric Rehabilitation Facility: a assessment of patient need for admission before movement to home setting Care pathway coordinator, role: o comunication between professionals & across settings coordination & continuity of care Acrive involvement of patients & informal caregivers Patient discharge summaries Evaluation meetings & open communication across providers 	

TC innovation name	Target population	TC innovation – Aims & Key components	CFIR domains associated with the TC innovation's implementation
TC Places Transition Care Places Masters et al. 2008, Australia (42)	 Older adults Concluded an acute hospital episode Requiring more time & support in a non- acute setting to complete their restorative process and optimize their functional 	A Aims: to minimize inappropriate extended hospital length of stay, prevent inappropriate admission to residential aged care, & optimize patient's independence /functional capacity	 Outer setting Inner setting Process
	process and optimize their instruction capacity	 TC intermediary places located in a residential care facility or a community setting Delivery of transition care in TC places: o goal-orientated, individualized o time-limited care o low-intensity therapies & services o case management Finalization of long-term care arrangements 	
ICM Intermediate Care Model	 Frail older adults, chronically ill Completed medical treatment at hospital but under to no home 	Aims: to reduce length of hospital stays, prevent hospital readmissions, retain patient's independence	 Intervention characteristics Inner setting Characteristics of individuals
Plochg et al. 2005, the Netherlands (43)	•	 Transfer Unit (beds) located in a residential home: low-intensity early discharge care model provision of services bridging the acute, primary, & social care Coordination of transitions by Hospital Liaison Nurse 	 Process
TC CAMP Transition Care Cognitive	 > 65yr older adults With cognitive impairment (dementia) 	 Aims: to reduce readmissions 	 Intervention characteristics Inner setting
Assessment and Management Pilot Renehan et al. 2013, Australia (44)	At conclusion of an episode of hospital care	 TC CAMP intermediary restorative care places located in a residential care facility Clinical Nurse Consultant (CNC), role: case management individualized care plan behavioural management Kev to Me, patient information tool 	 Characteristics of individuals Process

TC innovation name	Target population	TC innovation – Aims & Key components	CFIR domains associated with the TC innovation's implementation
Group III: 2 TC innovations to impr	<u>rove</u> care transitions from <u>hospital or home to nursing/residential care facility</u>	nursing/residential care facility	
QIP-TC Quality Improvement Project to Improve Transitions of Care for Older People	 Older people In transition between hospital and residential care settings during period of acute illness 	 Aims: to improve communication & information transfer and reduce readmissions Community geriatric service: 	 Intervention characteristics Outer setting Characteristics of individuals Process
Sutton et al. 2016, UK (45)		 Operational & community nurse Operational & community nurse Eaclifty staff Patient information summary form 	
CPN Follow-up visit by Community Pevichiatric Nurse	 Older people with dementia behavioral disturbances Expected to be admitted or are advised to 	Aims: to promote continuity of care & improve quality of care	 Intervention characteristics Outer setting Innar setting
Van Mierlo et al. 2015, the Netherlands (46)	move from home into a nursing facility	 Community Psychiatric Nurse (CPN), role: o follow-up visit 6 weeks after placement in a nursing home 	 Characteristics of individuals Process
		 clinical & behavioral assessment support & advice to facility nurse support to family caregiver 	
Group IV: 4 TC innovations to prev	Group IV: 4 TC innovations to <u>prevent</u> care transitions from <u>nursing facility or home to hospital</u>	me to hospital	
HaH Plus program Hospital at Home Plus Program	 ≥ 65yr older adults, requiring inpatient admission 	 Aims: to reduce mortality, readmission rates, costs, and achieve better patient/caregiver satisfaction 	 Intervention characteristics Outer setting Inner setting
Brody et al. 2019, USA (14)		 Acute-level care services provision at home as a substitute for hospital admission, <i>plus</i> A 30-day post-acute period of transitional care bundle (self-management, care coordination) 	 Characteristics of individuals Process
OPTIMISTIC Ontimizing Patient Transfers	Frail older residents of nursing facility	Aims: to reduce hospitalizations	 Intervention characteristics Inner setting
Detinizing reacts managed, Impacting Medical Quality, and Improving Symptoms: Transforming Institutional Care project		OPTIMISTIC RNs' & NPs' role: o identification, assessment, & management of acute conditions in nursing home condition of NUTEDACT (Association to Doduce)	
Ersek et al. 2018, USA (47)		 promoted for an experimentation of the promoted of the provided o	

TC innovation name	Target population	TC innovation – Aims & Key components	CFIR domains associated with the TC innovation's implementation
PCMH+TCM Patient-Centered Medical Home + Transitional Care Model	 ≥ 65yr older adults with multiple chronic conditions In community settings 	 Aims: to prevent avoidable emergency room visits & hospitalizations, and provide a continuous care management 	 Outer setting Inner setting Characteristics of individuals
Hirschman et al. 2017, USA (48)		 Patient-centered holistic approach Combination of disease management in primary care settings & home care: coordination of care during an episode of acute illness across settings, facilitated by: Transitional Care Nurse (TCN), role: home visits, telephone support active engagement of patient, family caregivers, & collaoration with primary care providers coordination of education & community services to develop self-management skills 	
INTERACT II Interventions to Reduce Acute Care Transfers	Residents of long-term care settings	A kims: to reduce the frequency of transfers to hospital, and improve quality of care for residents	 Outer setting Inner setting Characteristics of individuals
Rask et al. 2017, USA (49)		 Identification, evaluation, & communication of resident status changes Use of 4 practice tools: o quality improvement o communication o decision support o advance care planning 	<pre> Process </pre>
Notes: 'improve care transitions' - to with a view to creating optimal bene avoidable physical movement of olde	Notes: 'improve care transitions' - to provide and enhance the transitional care and services delivered during physical relocation with a view to creating optimal benefit as a result of the care transition; 'prevent care transitions' - to provide the care and se avoidable physical movement of older persons between two care settings or more, * RN: registered nurse, NP: nurse practitioner.	Notes: 'improve care transitions' - to provide and enhance the transitional care and services delivered during physical relocations of older persons from one care setting to another, with a view to creating optimal benefit as a result of the care transition; 'prevent care transitions' - to provide the care and services needed in order to avert an unnecessary or avoidable physical movement of older persons between two care settings or more; * RN: registered nurse, NP: nurse practitioner.	is from one care setting to another i order to avert an unnecessary o

Care transitions from nursing facility or home to hospital were the focus of four TC innovations. These innovations aimed to prevent care transitions. Hence, the main objectives were the provision of a value-based and patient-centered high-quality care (14), as well as the reduction and prevention of avoidable hospitalizations (47, 48), and reducing the frequency of transfers to acute hospital care (49). The unique component of all four innovations was "transition care management in place."

BARRIERS AND FACILITATORS TO THE IMPLEMENTATION OF TC INNOVATIONS

Factors reported in the 21 studies could be mapped to 61 CFIR&CTF constructs, out of which 19 were reported as barriers only, 8 as facilitators only, and 34 as both barriers and facilitators. Among these 34 factors, 15 were reported as having both influences concurrently in the same study. The reporting frequency, presented as number of studies, for the barriers and facilitators influencing the implementation of the transitional care innovations as mapped to the CFIR&CTF constructs is shown in Figure 2.

The most commonly reported domains impacting implementation were process (20 studies) and inner setting (19 studies), while factors in the outer setting were least reported (12 studies). Twenty-five factors were reported by at least five studies (25%), and therefore were considered the most prominent ones. Among these factors, we distinguished seven factors as predominant barriers and seven as predominant facilitators. The remaining 11 factors showed a nearly equivalent direction of influence as impeding and facilitating (i.e., indistinguishable). Here we use "predominant" when a factor was clearly and more frequently reported as either a barrier or facilitator, judged by whether at least two thirds of the total number of studies reporting this factor reported it as a barrier or facilitator. Nevertheless, this does not directly imply that these factors are the most important, but it conveys that they are very likely to affect the implementation of TC innovations in either direction of influence. The main findings describing the most prominent factors are presented below, and Figure 3 provides an overall summary.

Factors – Predominantly Barriers

Targeted groups – A mismatch between the TC innovation components and the intended profile of the recipients, older persons, was evident to affect its implementation as indicated in nine studies (14, 18, 31, 32, 34-36, 43, 47). Five studies reported that unclear eligibility criteria of the TC innovation often impeded the identification of older persons that could benefit from it (14, 18, 32, 35, 47). Another four studies stated that TC innovations were unable to meet the specific care needs of the targeted older persons due to the high frailty and complex conditions of the recipients, confirming an incompatible fit (31, 34, 36, 43).

Figure 2. Frequency of reported barriers and facilitators to TC innovations implementation, mapped to CFIR&CTF (61 constructs)



*represents factors cited by at least 5 studies (25%) as a barrier and/or facilitator; ^ denotes factor as a predominant barrier or facilitator; total number of studies is 21.

Complexity – The intricacy of the TC innovation design and the difficulty of putting it into action were reported mutually in five studies (14, 31, 36, 38, 39). Two studies cited that the necessity to involve multiple homecare service providers (14) and informal caregivers (31), and the absence of bundled care payment methods (14) led to difficulty in implementing TC innovations in home settings. Healthcare providers perceived that TC innovations with complex and extensive processes (39), unstandardized or detailed protocols (36), and hard to understand and use tools and checklists (38) affected the implementation negatively.

Readiness for implementation: available resources – Low staffing levels (43, 44, 46) and a lack of dedicated staff (14) were common impeding factors to the implementation of TC innovations. Similarly, staff turnover (38, 47, 49), plus losing key team members (39) and major program staff and contact persons (40) affected the implementation negatively. This led to increased costs and weakened relationships between organizations involved in implementing a TC innovation (40). Heavy workloads (38, 47, 49), time constraints (39, 46), and work schedule pressure (46) also hindered implementation, and sometimes led to less staff engagement (38). Limited availability of needed resources such as equipment and care service provisions (18), as well as financial constraints (47) or a lack of funding (37, 40) were notable barriers to implementation. Moreover, three studies indicated that an inadequate training and education offered to providers and staff hindered their ability to implement new TC innovations (36, 42, 47).

Continuity – A disrupted information flow, communication, or relationship between multiple healthcare providers and across organizations was described as cumbersome and impeding to the implementation of TC innovations (14, 18, 32, 34, 42, 46). Four studies reported that an insufficient, inconsistent, or discontinuous patient information exchange between different organizations often led to delays in coordination of services and care planning, which was the essence of some TC innovations (14, 18, 34, 42). Furthermore, the inefficient communication and difficulties in maintaining steady working relationships among the TC innovation program staff and, for example, the hospital or nursing home staff were barriers to the implementation (32, 46).

Implementation climate: relative priority – The existence of multiple quality improvement initiatives and projects within the organization often hindered the efforts to implement new TC innovations concurrently (33). Moreover, alternate quality improvement projects posed competition to the introduction of new TC innovations (39), and sometimes a mix of confusion and doubts among the staff on their need (47). Overall, staff described change fatigue as a main barrier to endorsing new transitional care activities, as well as leadership sometimes failing to actively endorse new transitional care programs (32, 33). Two studies indicated that major organizational changes also created different priorities among staff, and a reduced capacity and motivation to implement new TC innovations (38, 49).

Knowledge & beliefs about the intervention – The older persons' misconceptions about the TC innovation together with a limited awareness of its specific services and goals, as well as a low perceived value affected the enrolment process and implementation (31, 33, 36). Moreover, some older persons expressed privacy concerns over aspects of the innovations, such as home visits by care providers, and hence viewed it as a disruption with a little value (33, 35, 36). Similarly, mixed knowledge and beliefs surrounding the innovation (38), confusion on the innovation's direction (36), and not knowing what is expected (44, 47) by healthcare providers were reported as hindrances to the implementation. One study cited that care home staff believed that the new intervention would make them highly liable and accountable (45); whereas in another study, staff saw that a mind-shift is required or implementation is impeded (46).

Planning – Two studies indicated that following a less organized implementation plan with a low-quality and feasibility vision impeded the execution of a TC innovation (33, 39). While another four studies cited that the lack of clear initiation workflows and specific protocols (14, 47), as well as an absence of early induction and explanation of the innovation (35, 44), led to early missteps and confusion in rolling out the TC innovations (35, 47).

Factors – Predominantly Facilitators

Relative advantage – Four studies reported that the benefits and usefulness offered by a TC innovation facilitated its implementation (35, 38, 44, 46). Healthcare providers stated that TC innovations with certain supportive tools, such as compiling an older person's information during transitions between settings, helped staff work more efficiently, and thereafter enhanced the implementation (44). In addition, an improved quality of information transfer and communication between community and nursing home settings offered by one TC innovation's features was perceived as advantageous by staff (46). Moreover, the implementation of a TC innovation was facilitated when managers observed incremental benefits such as improved healthcare staff practice and skills (38).

Evidence strength & quality – Proven effectiveness and solid evidence on the TC innovation's ability to ensure positive outcomes enabled its implementation (35, 39, 40). Outcomes such as low readmission rates (40) and better patient satisfaction (39) resulting from a TC innovation led to a high buy-in from the healthcare providers (40) and a convinced leadership (35), which consequently supported the implementation.

Information Technology (IT) and Health Information Technology (HIT) resources: HIT systems – The presence of supportive electronic health information systems enhanced the implementation of TC innovations by enabling better communication, shared information documentation, and patient care management across settings (18, 36, 38, 48). Notably, the incorporation in patients' electronic files of either a TC innovation-specific checklist (38)

or signaling the involvement of a TC manager in the care management (18) facilitated the adoption.

Role – Defining clear roles and responsibilities for the key TC innovation implementing team members facilitated the implementation (35). Three studies reported that key staff played a critical role in implementation, through adhering to the application of the innovation's specific activities (31, 32), providing regular support, and serving as a liaison and communication channel between different care settings and caregivers (47).

Transition roles: frontline staff – Five studies reported that the presence of frontline staff with a designated transition role facilitated the execution of a TC innovation (31, 35, 42, 44, 47). A role directly attached to the innovation, such as transition care staff (42), advanced practice nurses (31), or a clinical nurse consultant (44) was vital to implement the core components of the innovation by being in direct contact with older persons, and able to identify and manage their transition care needs.

Reflecting & evaluating – Measurement capability and data availability

Regular communication and feedback between staff on the progress of implementing TC innovations, such as sharing successful outcome measures, fostered more leadership support for continuing the implementation (38-40). Furthermore, ensuring a continuous monitoring of the innovation's effectiveness, overall performance, as well as quality and safety for patients allowed for timely adaptations in the implementation process, together with maintaining its continuity (14, 35, 38).

Factors – Indistinguishable Barriers/Facilitators

Eleven factors across four domains were highly reported, however with an overall nearly equivalent influence as both impeding and facilitating the implementation of a TC innovation.

Cosmopolitanism – Although five studies reported that pre-existing partnerships, the establishment of new external networks, or sharing of practices between various healthcare organizations enabled a faster and better implementation of TC innovations (14, 40-42, 46), four studies indicated poor interorganizational relationships and unwillingness to collaborate as evident barriers (33, 40, 45, 48).

External policy & incentives – The presence of external unsupportive laws and regulations, as well as the discontinuity of national funding schemes, showed a negative influence on the implementation of TC innovations in four studies (14, 35, 40, 46). Conversely, another four studies cited that favorable extrinsic legislative changes (41, 49) or the availability of governmental sponsorship for new TC innovations were facilitators (36, 42).

Networks & communications – A challenging team formation with an absence of regular, effective, and clear communication among the members impeded the implementation, as cited in seven studies (14, 32, 34, 43, 46-48). In contrast, suggested facilitators included established interdisciplinary teams (39), strong coordination (33), or cooperative working relationships across team members (35, 36, 48).

Culture – Progressive (33), innovative (32), flexible (40), or problem-solving (35, 49) organizational norms and values with emphasis on patient-centered care (32), fostered implementing new TC innovations. In contrast, a mismatch in cultures between healthcare organizations or the presence of traditional and resistant to change values was shown to hinder the implementation (35, 43, 46, 47, 49).

Readiness for implementation (leadership engagement; access to knowledge & information) – Insufficient involvement and a limited support from existing leadership along with a lack of interest in implementing a new TC innovation affected the process negatively (32, 37, 39, 43, 47). Likewise, failing to provide the required information and initial training to staff on a new TC innovation hindered its implementation (14, 37, 44). In contrast, a high organizational commitment and sustained leadership (35, 38, 41, 49), and ensuring the access to knowledge and mentoring on the TC innovation, facilitated the implementation (35, 38, 42).

Skills, competencies, and other personal attributes – Six studies indicated that a lack of staff expertise, knowledge capacity, and skills, along with insufficient educational levels often delayed or ultimately hindered the implementation of TC innovations (14, 36, 38, 43, 45, 46). Conversely, another six studies suggested that staff with a wide experience in long-term care and possessing clinical and technical skills (31, 32, 35, 44, 47, 48), as well as high critical attributes (47) were a great source of implementation facilitation. Similarly, low motivation levels and frustration among the staff (36, 38) or patient's poor health literacy (34) and no acknowledgement of care needs (31) impeded implementation; yet a high motivation for change had a positive influence (31, 46, 49).

Engaging: key stakeholders, innovation participants, organizations and external context – The challenge to involve actively and early on the key healthcare professionals, patients, family, and external providers in addition to low levels of training and induction activities impeded the implementation of various TC innovations (14, 31, 33, 39-45). However, a continuous engagement of healthcare providers (36) and the patient (39, 44), alongside stimulating external collaborations (46), or ensuring family inclusion in care goals setting (42) fostered the implementation. Similarly, exercising team-building efforts (14, 39), gaining an early buy-in and support from key staff (32, 38, 48), and advertising the TC innovation well (35) were essential facilitators.



Figure 3. Overview of the factors influencing the implementation of TC innovations

PERSPECTIVES OF OLDER PERSONS, FAMILY, OR INFORMAL CAREGIVERS ON TC INNOVATIONS

Six studies reported on the overall perception of the older persons and/or their caregivers regarding the transitional care innovation being implemented. Often the feedback was not specific to the implementation aspect, but rather on the innovation's components, benefits, and satisfaction. Some components of the TC innovations, such as medication management, were perceived as a challenge for patients (37); whereas a transition role, such as a care coordinator (36), clinical nurse consultant (44), or community psychiatric nurse (46) was perceived as highly valuable and beneficial. In addition, the provision of clear information and expectations from the TC innovation was seen as highly satisfactory (44, 47). Three studies reported that older persons and their caregivers had a mixed experience with the innovation as either satisfying or devaluing, thus sometimes feeling that the components do not fit or meet their care transition needs or wishes (36, 41, 44).

DISCUSSION

Our study identified an interplay of 25 main factors that acted as barriers and facilitators during the implementation of diverse transitional care innovations. Fourteen factors presented with a predominant direction of influence. The important barriers were linked to the organization's implementation readiness and climate, targeted older populations, process planning, and users' knowledge. The significant enabling factors were the innovation's high relative advantage, transition roles of professionals, and evaluation of the implementation process. Furthermore, we could not distinguish a clear-cut direction for the influence of other key factors. By large, the current findings are in line with previous research and theories suggesting that a range of interrelated factors existing at multiple levels determine the success of the implementation of innovations (50, 51).

Our results indicate that certain factors related to the implementation process and intervention characteristics appear to be specific to transitional care innovations. Whilst the roles of middle managers (52, 53) and champions (54, 55) were indicated as facilitators to implementing general healthcare or long-term care (LTC) innovations, transition roles of frontline staff in LTC were key in facilitating the adoption and execution of TC innovations. Moreover, awareness of existing barriers in designing and tailoring TC innovations to the target population was seen as lacking across many of the studies we reviewed. This could be explained by the specific profile and care transition needs of older persons that seem to be overlooked when developing innovations. Even though the components of some TC innovations entailed the involvement of both older persons and caregivers in the development of care plans, a mismatch of needs occurred. As presented elsewhere, it is highly important to ensure patient engagement in codesigning processes or evaluations of care improvement initiatives such as TC innovations (56, 57). Moreover and in our attempt to answer the second research question, this review found only few studies that took the perspectives of transitional care recipients into account, while examining the implementation of TC innovations. The role of the older persons and thereby the consideration of their wishes and needs in the implementation process appear to be limited. Hence, the older persons and/or their informal or family caregivers' reflection on the actual implementation challenges are understudied, since the providers' perspectives are often those sought after.

Furthermore, the specific context and characteristics of LTC organizations play an integral role in implementing innovations (58-62). Correspondingly, our results indicated that the LTC organizational culture, implementation climate, readiness for implementation, implementation process, the individuals' skills and attributes, and internal communication dynamics have a major impact on the uptake of several TC innovations. This provides further evidence regarding the theory on organizational readiness for change (ORC) by Weiner (63), which explains that fostering the organization's capacity, commitment, and efficacy to

CHAPTER 2

change are notable drivers in creating readiness and ultimately enhance implementation. Similarly, our results affirm the work of Attieh et al. (64), in which five core theoretical components of ORC were identified including the organizational dynamics, change process. innovation readiness, institutional readiness, and personal readiness. Our results indicate that lacking resources often hindered the implementation of various TC innovations, and that the organizational culture had a prominent yet mixed influence on bringing about a change. According to Weiner (63), organizational resources and culture are among the contextual factors that can affect the organizational capacity and readiness for change. This review also identified that the individuals' skills, knowledge, perceived attitudes, and designated roles were prominent factors in implementing an innovation. This is evident as per Holt's et al. (65) and Weiner's (63) concepts of change efficacy, which explain that individuals in an organization with a high shared collective capability and confidence to implement new tasks successfully can enhance the organizational readiness for change. In addition, our findings on the importance of implementation climate explained by the individuals' relative priority to implement a TC innovation within an organization as well as their motivation levels relate to the organizational change commitment (63, 65, 66). Lastly, the literature indicated that organizational leadership and internal communication dynamics are instrumental in generating readiness for change, as was mirrored in our results (63, 66).

FUTURE RECOMMENDATIONS

Research

Prospective studies on the degree of influence of each identified barrier and facilitator on the implementation of a TC innovation are needed. This will enable the development of tailored implementation strategies by addressing the prioritized factors. Furthermore, focusing on the older person's perspective when studying the implementation process of TC innovations is required. This will alleviate the discontinuous and problematic care transitions for the older population.

Policy and practice in transitional care

Future implementation of TC innovations can benefit from a preassessment of the key components that underpin an LTC organization's readiness for change by using established ORC measurement instruments (67). Overall, these measures can offer an initial support for LTC organizations to better prepare for implementing innovations by reducing blinded change efforts. Simultaneously, LTC organizations can leverage their readiness for implementing change by, for example, adopting the concept of innovation management as reflected in A.T. Kearney's House of Innovation (68). This framework invites organizations to start with an innovation strategy and build an innovative and open culture. In addition, organizations must manage the innovation's process in an integrated and continuous manner from idea conception to implementation, as a way to avoid inefficiencies and ensure timely positive outcomes. Bates *et al.* (58) emphasized the power to create successful innovative healthcare

52

environments by making innovation a strategic priority. Henceforth, we recommend LTC organizations bolster their innovation readiness and management, whereby they encourage among professionals an incessant mind-set of "change is the norm." Nevertheless, this readiness should be fostered across the continuum of care spanning multiple LTC settings, given the nature of TC. In addition, transition roles or implementation support practitioners (69) should be instituted to better operationalize innovations in TC.

STRENGTHS AND LIMITATIONS

We consider the combined use of CFIR and CTF a methodological asset for conducting this review, especially in the process of data extraction and mapping of factors. The CFIR provided an intricate yet systematic way to understand the interconnectedness of the numerous factors. The inclusion of constructs from the CTF was found vital in detecting factors specific to care transitions. On the other hand, we acknowledge that different or additional factors could have been found had we chosen to use another framework.

This review has some limitations. First, it is subject to publication bias, since we only included articles published in peer-reviewed journals and excluded grey literature, pre-registries, and policy documents. Second, even though we used an extensive search strategy to identify relevant studies on implementing TC innovations, we might have missed some potentially relevant papers, as the aim of innovations in LTC is not always clearly described. Third, not all records were screened by two persons; only a random selection of 10% of the initial total records was screened by a second reviewer for titles and abstracts. Though agreement seemed satisfactory, we cannot fully rule out that some relevant sources could have been missed. Fourth, we did not perform critical appraisal for the included studies, even though it is not mandatory in scoping reviews' methodology, it could have added to the interpretability of the findings.

CONCLUSIONS

A diversity of factors impact the implementation of TC innovations; these include the innovation's complexity, relative advantage and evidence strength, organizational readiness for implementation, individuals' knowledge and beliefs, and the implementation process planning and evaluation. To ensure implementation potential, TC innovations need to address the right older target population; and transition roles for staff should be developed as key steps. LTC organizations can benefit from collaborating and leveraging concurrently their readiness for change along with adopting innovation management in order to succeed in implementing TC innovations. Furthermore, minimizing the confusion around how implementing innovation works, holds the potential to improve care transitions for older persons.

REFERENCES

- Naylor MD, Aiken LH, Kurtzman ET, Olds DM, Hirschman KB. The care span: the importance of transitional care in achieving health reform. Health Aff (Millwood). 2011;30(4):746–54.
- Enderlin CA, McLeskey N, Rooker JL, Steinhauser C, D'Avolio D, Gusewelle R, et al. Review of current conceptual models and frameworks to guide transitions of care in older adults. Geriatr Nurs. 2013;34(1):47–52.
- Marengoni A, Angleman S, Melis R, Mangialasche F, Karp A, Garmen A, et al. Aging with multimorbidity: a systematic review of the literature. Ageing Res Rev. 2011;10(4):430–9.
- 4. Allen J, Hutchinson AM, Brown R, Livingston PM. Quality care outcomes following transitional care interventions for older people from hospital to home: a systematic review. BMC Health Serv Res. 2014;14:346.
- Coleman EA. Falling through the cracks: challenges and opportunities for improving transitional care for persons with continuous complex care needs. J Am Geriatr Soc. 2003;51(4):549–55.
- Coleman EA, Boult C. American Geriatrics Society Health Care Systems C. Improving the quality of transitional care for persons with complex care needs. J Am Geriatr Soc. 2003;51(4):556–7.
- Hirschman KB, Hodgson NA. Evidence-based interventions for transitions in care for individuals living with dementia. Gerontologist. 2018;58(Suppl 1): S129–40.
- Britton MC, Ouellet GM, Minges KE, Gawel M, Hodshon B, Chaudhry SI. Care transitions between hospitals and skilled nursing facilities: perspectives of sending and receiving providers. Jt Comm J Qual Patient Saf. 2017;43(11): 565–72.
- 9. WHO. World report on ageing and health. Luxembourg: World Health Organization; 2015.
- Naylor MD, Brooten DA, Campbell RL, Maislin G, McCauley KM, Schwartz JS. Transitional care of older adults hospitalized with heart failure: a randomized, controlled trial. J Am Geriatr Soc. 2004;52(5):675–84.
- 11. Coleman EA, Parry C, Chalmers S, Min SJ. The care transitions intervention: results of a randomized controlled trial. Arch Intern Med. 2006;166(17):1822–8.
- 12. Jack BW, Chetty VK, Anthony D, Greenwald JL, Sanchez GM, Johnson AE, et al. A reengineered hospital discharge program to decrease rehospitalization: a randomized trial. Ann Intern Med. 2009;150(3):178–87.
- Ouslander JG, Bonner A, Herndon L, Shutes J. The interventions to reduce acute care transfers (INTERACT) quality improvement program: an overview for medical directors and primary care clinicians in long term care. J Am Med Dir Assoc. 2014;15(3):162–70.
- Brody AA, Arbaje AI, DeCherrie LV, Federman AD, Leff B, Siu AL. Starting up a hospital at home program: facilitators and barriers to implementation. J Am Geriatr Soc. 2019;67(3):588–95.
- Parker E, Zimmerman S, Rodriguez S, Lee T. Exploring best practices in home health care. Home Health Care Manag Pract. 2013;26(1):17–33.
- Grol R, Wensing M. What drives change? barriers to and incentives for achieving evidence-based practice. Med J Aust. 2004;180:S57–60.
- Grol R, Wensing M, Bosch M, Hulscher M, Eccles M. Theories on implementation of change in healthcare. In: Grol R, Wensing M, Eccles M, Davis D, editors. Improving patient care: the implementation of change in health care. 2nd ed. Oxford: Wiley; 2013. p. 18–39.
- Couture M, Sasseville M, Gascon V. Facilitators and barriers to implementing transitional care managers within a public health care system. J Gerontol Soc Work. 2016;59(4):364–77.
- 19. Zamora Z, McCall B, Patel L, Biese K, Lamantia M, Platts-Mills T, et al. Implementation of a web-based system to improve the transitional care of older adults. J Nurs Care Qual. 2012;27(2):182–9.
- Tricco AC, Lillie E, Zarin W, O'Brien KK, Colquhoun H, Levac D, et al. PRISMA extension for scoping reviews (PRISMA-ScR): checklist and explanation. Ann Intern Med. 2018;169(7):467–73.
- Arksey H, O'Malley L. Scoping studies: towards a methodological framework. Int J Soc Res Methodol. 2005;8(1):19– 32.
- 22. Levac D, Colquhoun H, O'Brien KK. Scoping studies: advancing the methodology. Implement Sci. 2010;5:69.
- McDonagh M, Peterson K, Raina P, Chang S, Shekelle P. Avoiding bias in selecting studies. In: Methods guide for comparative effectiveness reviews. Agency for Healthcare Research and Quality; 2013. https://effectivehealthca re.ahrq.gov/sites/default/files/pdf/methods-guidance-bias_methods.pdf. Accessed 20 Aug 2019.
- 24. Stoll CRT, Izadi S, Fowler S, Green P, Suls J, Colditz GA. The value of a second reviewer for study selection in systematic reviews. Res Synth Methods. 2019;10(4):539–45.
- Moher D, Liberati A, Tetzlaff J, Altman DG, Group P. Preferred reporting items for systematic reviews and metaanalyses: the PRISMA statement. J Clin Epidemiol. 2009;62(10):1006–12.
- Damschroder LJ, Aron DC, Keith RE, Kirsh SR, Alexander JA, Lowery JC. Fostering implementation of health services research findings into practice: a consolidated framework for advancing implementation science. Implement Sci. 2009;4:50.
- 27. Dy SM, Ashok M, Wines RC, Rojas SL. A framework to guide implementation research for care transitions interventions. J Healthc Qual. 2015;37(1):41–54.
- 28. Kirk MA, Kelley C, Yankey N, Birken SA, Abadie B, Damschroder L. A systematic review of the use of the

consolidated framework for implementation research. Implement Sci. 2016;11:72.

- 29. CFIR data analysis tools, codebook template. CFIR Research Team Center for Clinical Management Research. 2020. https://cfirguide.org/tools/tools-andtemplates/. Accessed 15 Sept 2019.
- CFIR data analysis tools, rating rules. CFIR Research Team Center for Clinical Management Research. 2020. <u>https://</u> cfirguide.org/tools/tools-and-templa tes/. Accessed 01 Sept 2019.
- 31. Bradway C, Trotta R, Bixby MB, McPartland E, Wollman MC, Kapustka H, et al. A qualitative analysis of an advanced practice nurse-directed transitional care model intervention. Gerontologist. 2012;52(3):394–407.
- Hung D, Leidig RC. Implementing a transitional care program to reduce hospital readmissions among older adults. J Nurs Care Qual. 2015;30(2):121–9.
- Hung D, Truong Q, Yakir M, Nicosia F. Hospital-community partnerships to aid transitions for older adults: applying the care transitions framework. J Nurs Care Qual. 2018;33(3):221–8.
- 34. McNeil D, Strasser R, Lightfoot N, Pong R. A "simple" evidence-based intervention to improve care transitions for frail patients with complex health conditions: why didn't it work as expected? Healthc Q. 2016; 19(2):67–72.
- Naylor MD, Feldman PH, Keating S, Koren MJ, Kurtzman ET, Maccoy MC, et al. Translating research into practice: transitional care for older adults. J Eval Clin Pract. 2009;15(6):1164–70.
- Nurjono M, Shrestha P, Ang IYH, Shiraz F, Yoong JS, Toh SES, et al. Implementation fidelity of a strategy to integrate service delivery: learnings from a transitional care program for individuals with complex needs in Singapore. BMC Health Serv Res. 2019;19(1):177.
- Parrish MM, O'Malley K, Adams RI, Adams SR, Coleman EA. Implementation of the care transitions intervention: sustainability and lessons learned. Prof Case Manag. 2009;14(6):282–93.
- Rosstad T, Garasen H, Steinsbekk A, Haland E, Kristoffersen L, Grimsmo A. Implementing a care pathway for elderly patients, a comparative qualitative process evaluation in primary care. BMC Health Serv Res. 2015;15:86.
- Williams MV, Li J, Hansen LO, Forth V, Budnitz T, Greenwald JL, et al. Project BOOST implementation: lessons learned. South Med J. 2014;107(7):455–65.
- 40. Xiang X, Robinson-Lane SG, Rosenberg W, Alvarez R. Implementing and sustaining evidence-based practice in health care: the Bridge Model experience. J Gerontol Social Work. 2018;61(3):280–94.
- Everink IH, van Haastregt JC, Maessen JM, Schols JM, Kempen GI. Process evaluation of an integrated care pathway in geriatric rehabilitation for people with complex health problems. BMC Health Serv Res. 2017;17(1):34.
- 42. Masters S, Halbert J, Crotty M, Cheney F. What are the first quality reports from the transition care program in australia telling us? Australas J Ageing. 2008;27(2):97–102.
- Plochg T, Delnoij DM, van der Kruk TF, Janmaat TA, Klazinga NS. Intermediate care: for better or worse? process evaluation of an intermediate care model between a university hospital and a residential home. BMC Health Serv Res. 2005;5:38.
- 44. Renehan E, Haralambous B, Galvin P, Kotis M, Dow B. Evaluation of a transition care cognitive assessment and management pilot. Contemp Nurse. 2013;43(2):134–45.
- 45. Sutton E, Dixon-Woods M, Tarrant C. Ethnographic process evaluation of a quality improvement project to improve transitions of care for older people. BMJ Open. 2016;6(8):e010988.
- 46. Van Mierlo LD, Bootsma-Van der Wiel A, Meiland FJ, Van Hout HP, Stek ML, Droes RM. Tailored mental health care after nursing home admission: improving transfers of people with dementia with behavioral problems. An explorative study. Aging Ment Health. 2015;19(10):902–11.
- Ersek M, Hickman SE, Thomas AC, Bernard B, Unroe KT. Stakeholder perspectives on the optimizing patient transfers, impacting medical quality, and improving symptoms: transforming institutional care (OPTIMISTIC) project. Gerontologist. 2018;58(6):1177–87.
- Hirschman KB, Shaid E, Bixby MB, Badolato DJ, Barg R, Byrnes MB, et al. Transitional care in the patient-centered medical home: lessons in adaptation. J Healthc Qual. 2017;39(2):67–77.
- Rask KJ, Hodge J, Kluge L. Impact of contextual factors on interventions to reduce acute care transfers II implementation and hospital readmission rates. J Am Med Dir Assoc. 2017;18(11):991.e11–5.
- 50. Geerligs L, Rankin NM, Shepherd HL, Butow P. Hospital-based interventions: a systematic review of staff-reported barriers and facilitators to implementation processes. Implement Sci. 2018;13(1):36.
- 51. Nilsen P. Making sense of implementation theories, models and frameworks. Implement Sci. 2015;10:53.
- Birken SA, DiMartino LD, Kirk MA, Lee SY, McClelland M, Albert NM. Elaborating on theory with middle managers' experience implementing healthcare innovations in practice. Implement Sci. 2016;11:2.
- Birken SA, Lee SY, Weiner BJ, Chin MH, Chiu M, Schaefer CT. From strategy to action: how top managers' support increases middle managers' commitment to innovation implementation in health care organizations. Health Care Manage Rev. 2015;40(2):159–68.
- 54. Bonawitz K, Wetmore M, Heisler M, Dalton VK, Damschroder LJ, Forman J, et al. Champions in context: which attributes matter for change efforts in healthcare? Implement Sci. 2020;15(1):62.
- Palmer JA, Parker VA, Barre LR, Mor V, Volandes AE, Belanger E, et al. Understanding implementation fidelity in a pragmatic randomized clinical trial in the nursing home setting: a mixed-methods examination. Trials. 2019; 20(1):656.
- Boaz A, Robert G, Locock L, Sturmey G, Gager M, Vougioukalou S, et al. What patients do and their impact on implementation. J Health Organ Manag. 2016;30(2):258–78.

- 57. Bombard Y, Baker GR, Orlando E, Fancott C, Bhatia P, Casalino S, et al. Engaging patients to improve quality of care: a systematic review. Implement Sci. 2018;13(1):98.
- Bates DW, Sheikh A, Asch DA. Innovative environments in health care: where and how new approaches to care are succeeding. Health Aff (Millwood). 2017;36(3):400–7.
- Bunn F, Goodman C, Corazzini K, Sharpe R, Handley M, Lynch J, et al. Setting priorities to inform assessment of care homes' readiness to participate in healthcare innovation: a systematic mapping review and consensus process. Int J Environ Res Public Health. 2020;17(3).
- Nieboer AP, Strating MM. Innovative culture in long-term care settings: the influence of organizational characteristics. Health Care Manage Rev. 2012; 37(2):165–74.
- 61. Verleye K, Gemmel P. Innovation in the elderly care sector: at the edge of chaos. J Manage Marketing Healthc. 2011;4(2):122–8.
- 62. Williams I. Organizational readiness for innovation in health care: some lessons from the recent literature. Health Serv Manage Res. 2011;24(4):213–8.
- 63. Weiner BJ. A theory of organizational readiness for change. Implement Sci. 2009;4:67.
- 64. Attieh R, Gagnon MP, Estabrooks CA, Legare F, Ouimet M, Roch G, et al. Organizational readiness for knowledge translation in chronic care: a review of theoretical components. Implement Sci. 2013;8:138.
- 65. Holt DT, Helfrich CD, Hall CG, Weiner BJ. Are you ready? How health professionals can comprehensively conceptualize readiness for change. J Gen Intern Med. 2010;25(Suppl 1):50–5.
- 66. Attieh R, Gagnon MP, Estabrooks CA, Legare F, Ouimet M, Vazquez P, et al. Organizational readiness for knowledge translation in chronic care: a Delphi study. BMC Health Serv Res. 2014;14:534.
- 67. Gagnon MP, Attieh R, Ghandourel K, Legare F, Ouimet M, Estabrooks CA, et al. A systematic review of instruments to assess organizational readiness for knowledge translation in health care. PLoS One. 2014;9(12):e114338.
- 68. Riel A. Innovation managers 2.0: which competencies? European Conference on Software Process Improvement 2011.
- 69. Albers B, Metz A, Burke K. Implementation support practitioners—a proposal for consolidating a diverse evidence base. BMC Health Serv Res. 2020;20(1):368

APPENDICES

Appendix 2A. Search strategy for electronic databases

Appendix 2B. Consolidated Framework for Implementation Research & Care Transitions Framework constructs description. This file provides the description and definitions of the CFIR constructs and the constructs selected and used from the CTF.

Appendix 2A. Search strategy for electronic databases

PubMed/MEDLINE Concept 1: Implementation

MeSH terms	Keywords
Implementation Science[MeSH]	implement"[tiab] OR adopt"[tiab] OR integrat"[tiab] OR disseminat"[tiab] OR promot"[tiab]

Concept 2: Innovation

MeSH terms	Keywords
Diffusion of Innovation[MeSH] OR	program*[tiab] OR model*[tiab] OR intervention*[tiab] OR system*[tiab]
Organizational Innovation[MeSH]	OR practice*[tiab] OR tool*[tiab] OR approach*[tiab] OR pathway*[tiab] OR
OR Inventions[MeSH] OR Change	change*[tiab] OR innovat*[tiab] OR invention*[tiab]
Management[MeSH]	

Concept 3: Care Transition

MeSH terms	Keywords
Patient Transfer[MeSH] OR Transitional Care[MeSH] OR Patient Handoff[MeSH]	care transition*[tiab] OR "transition of care"[tiab] OR "transitions of care"[tiab] OR care transfer*[tiab] OR "transfer of care"[tiab] OR patient transition*[tiab] OR patient transfer*[tiab] OR "transfer of patient"[tiab] OR patient relocat*[tiab] OR patient handover[tiab] OR patient hand-over[tiab] OR patient handoff*[tiab]

Concept 4: Older Persons

MeSH terms	Keywords
Aged[MeSH] OR Geriatrics[MeSH] OR Frail Elderly[MeSH]	older*[tiab] OR elder*[tiab] OR frail*[tiab] OR geriatri*[tiab] OR old age*[tiab] OR oldest old*[tiab] OR senior*[tiab] OR very old*[tiab] OR older people[tiab] OR older patient*[tiab] OR older age*[tiab] OR older adult*[tiab] OR older population*[tiab] OR older person*[tiab] OR geriatric*[tiab]

EMBASE

Concept 1: Implementation

Subject Headings	Keywords
	(implement* OR promot* OR adopt* OR integrat* OR disseminat*).ti,ab,kw.

Concept 2: Innovation

Subject Headings	Keywords
exp Organization/ OR	(program* OR model* OR intervention* OR system* OR practice* OR tool* OR approach* OR pathway* OR change* OR innovat* OR invention*). ti,ab,kw.

Concept 3: Care transition

Subject Headings	Keywords
	(care transition* OR "transition of care" OR "transitions of care" OR care transfer* OR "transfer of care" OR patient transition* OR patient transfer* OR "transfer of patient" OR patient relocat* OR patient handover OR patient hand-over OR patient handoff*).ti,ab,kw.

Concept 4: Older persons

Subject Headings	Keywords
exp Aged/ OR	(older* OR elder* OR frail* OR geriatri* OR old age* OR oldest old* OR senior* OR very old* OR older people OR older patient* OR older age* OR older adult* OR older population* OR older person* OR geriatric*).ti,ab,kw.

<u>CINAHL</u>

Concept 1: Implementation

Subject Headings	Keywords
(MH "Implementation Science")	TI implement* OR TI promot* OR TI adopt* OR TI integrat* OR TI disseminat* OR
	AB implement* OR AB promot* OR AB adopt* OR AB integrat* AB disseminat*

Concept 2: Innovation

Subject Headings	Keywords	
(MH "Diffusion of Innovation+")	TI innovat* OR TI change* OR TI invention* OR TI model* OR TI program* OR TI intervention* OR TI system* OR TI practice* OR TI tool* OR TI approach* OR TI pathway* OR	
	AB innovat* OR AB change* OR AB invention* OR AB model* OR AB program* OR AB intervention* OR AB system* OR AB practice* OR AB tool* OR AB approach* OR AB pathway*	

Concept 3: Care Transition

Subject Headings	Keywords		
(MH "Transitional Care")	TI "care transition*" OR TI "transition of care" OR TI "transitions of care" OR TI "care transfer*" OR TI "transfer of care" OR TI "patient transition*" OR TI "patient transfer*" OR TI "transfer of patient" OR TI "patient relocat*" OR TI "patient handover" OR TI "patient hand-over" OR TI "patient handoff*" OR		
	AB "care transition*" OR AB "transition of care" OR AB "transitions of care" OR AB "care transfer*" OR AB "transfer of care" OR AB "patient transition*" OR AB "patient transfer*" OR AB "transfer of patient" OR AB "patient relocat*" OR AB "patient handover" OR AB "patient hand-over" OR AB "patient handoff*"		

Concept 4: Older persons

Subject Headings	Keywords
(MH "Aged+")	TI elder* OR TI older* OR TI frail* OR TI geriatri* OR TI old age* OR TI oldest old* OR TI senior* OR TI very old* OR TI older people OR TI older patient* OR TI older age* OR TI older adult* OR TI older population* OR TI older person* OR TI geriatric* OR
	AB elder* OR AB older* OR AB frail* OR AB geriatri* OR AB old age* OR AB oldest old* OR AB senior* OR AB very old* OR AB older people OR AB older patient* OR AB older age* OR AB older adult* OR AB older population* OR AB older person* OR AB geriatric*

Appendix 2B. Consolidated Framework for Implementation Research & Care Transitions Framework constructs description. This file provides the description and definitions of the CFIR constructs and the constructs selected and used from the CTF.

	constructs / CTF ted constructs	Short Description			
I. Do	main: INTERVENTION CH	IARACTERISTICS			
A	Intervention Source	Perception of key stakeholders about whether the intervention is externally or internally developed.			
в	Evidence Strength & Quality	Stakeholders' perceptions of the quality and validity of evidence supporting the belief that the intervention will have desired outcomes.			
с	Relative Advantage	Stakeholders' perception of the advantage of implementing the intervention versus an alternative solution.			
D	Adaptability	The degree to which an intervention can be adapted, tailored, refined, or reinvented to meet local needs.			
E	Trialability	The ability to test the intervention on a small scale in the organization, and to be able to reverse course (undo implementation) if warranted.			
F	Complexity	Perceived difficulty of implementation, reflected by duration, scope, radicalness, disruptiveness, centrality, and intricacy and number of steps required to implement.			
G	Design Quality & Packaging	Perceived excellence in how the intervention is bundled, presented, and assembled.			
н	Cost	Costs of the intervention and costs associated with implementing the intervention including investment, supply, and opportunity costs.			
CTF	Vision & Change Strategy	The proposed changes envisioned by the intervention and the theory of change how the intervention is supposed to work, what it is meant to achieve or do. May be explicated in logic models, goals, outcomes, performance measures.			
CTF	Targeted Groups	Staff and others (vendors, patients) who are the intended recipients or beneficiaries of the intervention.			
CTF	Feasibility	Target group and other stakeholders' perceptions of the extent to which the intervention can be successfully used or carried out within the organization(s).			
CTF	Compatibility	Target group and stakeholder perception of the alignment of the meaning, values, and norms attached to care transitions with those held by members of the organization(s).			
CTF	Radicaliness	Target group and other stakeholder perceptions of the degree of difference between the change envisioned and the current state of care transitions			
CTF	User Control	The degree to which the intervention relies on the end-users' authority/skill to implement the intervention on their own vs. reliance on experts.			
CTF	Location of intervention activity	Components of the intervention conducted outside the hospital/clinic/office setting using external service providers and organizations.			
CTF	Workflows	Tasks and workflows, including interdependencies between them that are the focus of the intervention or will be affected by it.			
CTF	Task/Process standardization	Degree to which the intervention seeks to standardize tasks and processes that require iterative consultation.			
CTF	History	Experiences with similar interventions within the organizations or within the target groups.			
II. Do	main: OUTER SETTING				
Α	Patient Needs & Resources	The extent to which patient needs, as well as barriers and facilitators to meet those needs, are accurately known and prioritized by the organization.			
в	Cosmopolitanism	The degree to which an organization is networked with other external organizations.			
с	Peer Pressure	Mimetic or competitive pressure to implement an intervention; typically because most or other key peer or competing organizations have already implemented or are in a bid for a competitive edge.			
D	External Policy & Incentives	A broad construct that includes external strategies to spread interventions, including policy and regulations (governmental or other central entity), external mandates, recommendations and guidelines, pay-for-performance, collaboratives, and public or benchmark reporting.			

Appendix	2B.	Continued
----------	-----	-----------

	constructs / CTF cted constructs	Short Description				
CTF	Technological Environment	The technological trends and movements and the availability of technological innovations that may affect the intervention and its context.				
CTF	Population Needs and Resources	Prevalence of conditions and disease in the population served and the characteristics of the community that are determinants of health status.				
CTF	Community Resources	Availability and access of service providers, aging resources, and multiple levels of community services and supports not directly involved in the intervention.				
III. D	omain: INNER SETTING					
Α	Structural Characteristics	The social architecture, age, maturity, and size of an organization.				
в	Networks & Communications	The nature and quality of webs of social networks and the nature and quality of formal and informal communications within an organization.				
с	Culture	Norms, values, and basic assumptions of a given organization.				
D	Implementation Climate:	The absorptive capacity for change, shared receptivity of involved individuals to an intervention, and the extent to which use of that intervention will be rewarded, supported, and expected within their organization.				
D.1	Tension for Change	The degree to which stakeholders perceive the current situation as intolerable or needing change.				
D.2	Compatibility	The degree of tangible fit between meaning and values attached to the intervention by involved individuals, how those align with individuals' own norms, values, and perceived risks and needs, and how the intervention fits with existing workflows and systems.				
D.3	Relative Priority	Individuals' shared perception of the importance of the implementation within the organization.				
D.4	Organizational Incentives & Rewards	Extrinsic incentives such as goal-sharing awards, performance reviews, promotions, and raises in salary, and less tangible incentives such as increased stature or respect.				
D.5	Goals and Feedback	The degree to which goals are clearly communicated, acted upon, and fed back to staff, and alignment of that feedback with goals.				
D.6	Learning Climate	A climate in which: a) leaders express their own fallibility and need for team members' assistance and input; b) team members feel that they are essential, valued, and knowledgeable partners in the change process; c) individuals feel psychologically safe to try new methods; and d) there is sufficient time and space for reflective thinking and evaluation.				
CTF	Mandate	Whether compliance with the intervention is expected within the organization.				
CTF	Accountability	Whether entities are subject to tangible consequences for noncompliance.				
E	Readiness for Implementation:	Tangible and immediate indicators of organizational commitment to its decision to implement an intervention.				
E.1	Leadership Engagement	Commitment, involvement, and accountability of leaders and managers with the implementation.				
E.2	Available Resources	The level of resources dedicated for implementation and on-going operations, including money, training, education, physical space, and time.				
E.3	Access to Knowledge & Information	Ease of access to digestible information and knowledge about the intervention and how to incorporate it into work tasks.				
CTF	Staff Commitment	The degree of clinician, transitional, and community care staff, patient, and caregiver involvement in transition planning				
CTF	IT and HIT Resources:	Technological infrastructure in place to support electronic information management, including IT that crosses organizations.				
CTF	HIT Systems	Electronic information management infrastructure and technologies available to clinicians to manage patient care, data, and communications.				
CTF	IT Systems	Technological systems and capabilities to support care transitions.				
CTF	HIT/IT Accessibility	Includes features of the physical, technical, and social environment in the organization that determine the use, accessibility, and acceptability of technology in patient care.				

Appendix 2B. Continued

	constructs / CTF ted constructs	Short Description				
CTF	Other Resources	Resources for implementation and ongoing operations to support change and innovation, including grant or other funding specific to care transitions.				
CTF	Patient Self- management Infrastructure	Training, counseling, and education available to patients prior to the intervention within the hospital and ambulatory setting.				
CTF	Continuity	Information continuity (exchange of information) and relationship continuity, both with providers and patients/caregivers and across organizations.				
CTF	Patient/caregiver- centeredness	Extent to which the organization(s) knows and prioritizes patient and caregiver goals, needs, and preferences, and has the resources and services to meet them				
IV. D	omain: CHARACTERISTI	CS OF INDIVIDUALS				
Α	Knowledge & Beliefs about the Intervention	Individuals' attitudes toward and value placed on the intervention as well as familiarity with facts, truths, and principles related to the intervention.				
в	Self-efficacy	Individual belief in their own capabilities to execute courses of action to achieve implementation goals.				
с	Individual Stage of Change	Characterization of the phase an individual is in, as he or she progresses toward skilled, enthusiastic, and sustained use of the intervention.				
D	Individual Identification with Organization	A broad construct related to how individuals perceive the organization, and their relationship and degree of commitment with that organization.				
E	Other Personal Attributes	A broad construct to include other personal traits such as tolerance of ambiguity, intellectual ability, motivation, values, competence, capacity, and learning style.				
CTF	Collective Efficacy	Conviction of individuals and teams involved that the intervention can be carried out in cooperation with each other.				
CTF	Skills and Competencies	Degree of relevant subject matter expertise, skills, and competencies within the implementing team, unit, and organization.				
CTF	Role	Individual's role and responsibility for the intervention. The degree of multiple or shared roles.				
CTF	Authority	Individual provider's perceived and actual degree of authority to make decisions and act autonomously.				
CTF	Socioeconomic Demographics	Characteristics related to the individual's socioeconomic status.				
CTF	Patient Needs and Resources	Patient priorities for health and health care priorities and the social and economic capital to address those priorities.				
CTF	Caregiver Needs and Resources	Caregiver priorities for health and health care, and the social and economic capital to address those priorities.				
V. Do	main: PROCESS					
Α	Planning:	The degree to which a scheme or method of behavior and tasks for implementing an intervention are developed in advance, and the quality of those schemes or methods.				
CTF	Assessing	Formal assessment of care transitions issues; the needs of the users; barriers to change; the timing of these activities relative to implementation.				
CTF	Contingency Planning	Plans for adaptation in response to various scenarios and outcomes.				
CTF	Acquiring and Allocating Resources	Resources dedicated to implementing the intervention; the adequacy of those allocations.				
CTF	Process Ownership	The diversity of transition roles involved in processes of implementation; authority and accountability for these activities.				
CTF	Transition Roles:	Roles of individuals involved in the decision to adopt, execute, and facilitate the intervention.				
CTF	Organizational Leaders	Managers and others with the authority to dedicate resources and make decisions to adopt, maintain, or abandon the implementation.				
CTF	Frontline Staff	Administrative staff, providers (within and outside the organization) who will carry out the intervention or be affected by it.				

Append	dix	2B.	Continued
--------	-----	-----	-----------

	constructs / CTF ted constructs	Short Description
CTF	Integrators	Individuals who build relationships between organizations and create linkages to facilitate the intervention.
CTF	Patients, Caregivers, and Other Stakeholders	Patient and his/her family members, and members of the family's support network.
В	Engaging:	Attracting and involving appropriate individuals in the implementation and use of the intervention through a combined strategy of social marketing, education, role modeling, training, and other similar activities.
B.1	Opinion Leaders	Individuals in an organization who have formal or informal influence on the attitudes and beliefs of their colleagues with respect to implementing the intervention.
B.2	Formally Appointed Internal Implementation Leaders	Individuals from within the organization who have been formally appointed with responsibility for implementing an intervention as coordinator, project manager, team leader, or other similar role.
B.3	Champions	"Individuals who dedicate themselves to supporting, marketing, and 'driving through' an [implementation]", overcoming indifference or resistance that the intervention may provoke in an organization.
В.4	External Change Agents	Individuals who are affiliated with an outside entity who formally influence or facilitate intervention decisions in a desirable direction.
B.5	Key Stakeholders*	Individuals from within the organization that are directly impacted by the innovation, e.g., staff responsible for making referrals to a new program or using a new work process.
B.6	Innovation Participants*	Individuals served by the organization that participate in the innovation, e.g., patients in a prevention program in a hospital.
CTF	Engaging Organizations, External Context	Developing and capitalizing on relationships with providers, leaders, and frontline staff in the implementing organizations, and to external providers, resources, funders.
с	Executing:	Carrying out or accomplishing the implementation according to plan.
CTF	Decision-making	Frequency, duration, and timing of the activities involved in making decisions. The directionality of these activities.
CTF	Staging and Iteration	Degree to which the care transition is carried out in iterative, incremental steps or implemented in its entirety within a specified period.
D	Reflecting & Evaluating:	Quantitative and qualitative feedback about the progress and quality of implementation accompanied with regular personal and team debriefing about progress and experience.
CTF	Measurement Capability and Data Availability	Availability of timely data. Capacity for monitoring, evaluation, and process improvement. Includes measurement differences; accountability for collection, documentation, and analysis.

*Two additional constructs (engaging: key stakeholders, innovation participants) under engaging in the process domain were added as per CFIR research group (<u>https://cfirguide.org/</u>).

REFERENCES

- Damschroder LJ, Aron DC, Keith RE, Kirsh SR, Alexander JA, Lowery JC. Fostering implementation of health services research findings into practice: a consolidated framework for advancing implementation science. Implement Sci. 2009;4:50.
- Dy SM, Ashok M, Wines RC, Rojas Smith L. A framework to guide implementation research for care transitions interventions. J Healthc Qual. 2015;37(1):41-54.



3

CHAPTER

Fostering the Implementation of Transitional Care Innovations for Older Persons: Prioritizing the Influencing Key Factors Using a Modified Delphi Technique

Published as:

Fakha A, de Boer B, van Achterberg T, Hamers J, Verbeek H. Fostering the implementation of transitional care innovations for older persons: prioritizing the influencing key factors using a modified Delphi technique. BMC Geriatr. 2022; 22(1):131. https://doi. org/10.1186/s12877-021-02672-2

ABSTRACT

BACKGROUND

Transitions in care for older persons requiring long-term care are common and often problematic. Therefore, the implementation of transitional care innovations (TCIs) aims to improve necessary or avert avoidable care transitions. Various factors were recognized as influencers to the implementation of TCIs. This study aims to gain consensus on the relative importance level and the feasibility of addressing these factors with implementation strategies from the perspectives of experts. This work is within TRANS-SENIOR, an innovative research network focusing on care transitions.

METHODS

A modified Delphi study was conducted with international scientific and practice-based experts, recruited using purposive and snowballing methods, from multiple disciplinary backgrounds, including implementation science, transitional care, long-term care, and healthcare innovations. This study was built on the findings of a previously conducted scoping review, whereby 25 factors (barriers, facilitators) influencing the implementation of TCIs were selected for the first Delphi round. Two sequential rounds of anonymous online surveys using an *a priori* consensus level of >70% and a final expert consultation session were performed to determine the implementation factors': i) direction of influence, ii) importance, and iii) feasibility to address with implementation strategies. The survey design was guided by the Consolidated Framework for Implementation Research (CFIR). Data were collected using Qualtrics software and analyzed with descriptive statistics and thematic analysis.

RESULTS

Twenty-nine experts from 10 countries participated in the study. Eleven factors were ranked as of the highest importance among those that reached consensus. Notably, organizational and process-related factors, including engagement of leadership and key stakeholders, availability of resources, sense of urgency, and relative priority, showed to be imperative for the implementation of TCIs. Nineteen factors reached consensus for feasibility of addressing them with implementation strategies; however, the majority were rated as difficult to address. Experts indicated that it was hard to rate the direction of influence for all factors.

CONCLUSIONS

Priority factors influencing the implementation of TCIs were mostly at the organizational and process levels. The feasibility to address these factors remains difficult. Alternative strategies considering the interaction between the organizational context and the outer setting holds a potential for enhancing the implementation of TCIs.

KEYWORDS

Implementation, innovation, transitional care, Delphi technique, leadership, engagement, older persons, factors, strategies.

BACKGROUND

Transitions in care are common among older persons and entail the movement between different settings and healthcare providers (1, 2). Research shows that older persons have at least one transition towards their end of life, and one in five experience an adverse event in common transition from hospital to home (1, 3). Transitional care innovations (TCIs) are emerging evidence-based interventions (EBIs) designed to enhance the continuity and coordination of care for older persons when transferring between multiple care settings (4-6). Numerous TCIs demonstrated promising evidence for their effectiveness, such as in relation to reducing hospital readmissions, shortening hospital stay, preventing unnecessary admission to a nursing facility, averting hospitalization during an emergency department visit, or improving quality of life (2, 6-10).

While the positive outcomes of TCIs are encouraging, the successful translation of these innovations from trials into "real-world settings" is a main challenge (4, 11). The implementation of TCIs in practice remains slow and ambiguous with a lack of rigorous evidence on how to best achieve translation (11). The key components of most TCIs cross the care continuum and involve multiple care settings, which render them intricate and multifaceted (6, 12). Therefore, integrating TCIs into an existing healthcare system with specific regulations, reimbursement, and funding mechanisms is a starting point of an onerous roadmap to their implementation (11, 13). Moreover, TCIs normally involve two or more care settings or organizations that can be at different levels of readiness for implementing new interventions. Hence, misalignment of the different organizations' readiness for change, which encompasses factors such as staff commitment, receptive context for innovation, priority setting, change agents, or dedicated resources, is a basic difficulty in implementation of TCIs (4, 13). Correspondingly, while the older persons remain the core and common element across various TCIs, the heterogeneity of their care needs prevails. For instance, transitions in care for older persons with dementia (14) differ from those who suffer from the consequences of a stroke (15), which in turn, adds to the complexity of implementing TCIs.

Understanding these challenges and the various interacting constituents of TCIs illuminates the realm of implementing such complex healthcare interventions (16). Consequently, several research efforts identified factors (barriers, facilitators) influencing the implementation of TCIs in order to better inform implementers and enhance the process (6, 13, 17). Failure to target the right older population, discontinuous information exchange among care providers, and a lack of organizational resources with low priority given to innovation were among the prominent factors reported to hinder the implementation of TCIs (6). In contrast, predominant facilitators included a demonstrated advantage of the TCIs for the stakeholders, the presence of frontline staff with a care transition role, as well as a continuous evaluation and monitoring process (6). However, other factors such as leadership engagement and

external policies and incentives were highly reported in the literature, but with a mixed (occasionally enabling/occasionally hindering) influence (6).

Although an awareness of these common factors helps to overcome the challenges of implementing TCIs, this compilation results in multiple and diverse factors, which are highly variable across multiple contexts. Moreover, there is a lack of prioritization based on the importance of the influencing factors, and there is a dearth in evidence on the feasibility of addressing the barriers and leveraging the facilitators with implementation strategies.

This study builds from a scoping review by the research team that identified 25 prominent factors influencing the implementation of TCIs (6). The study aims to achieve expert consensus on the i) direction of influence (hindering, facilitating) of the known factors relevant to the implementation of TCIs that were predetermined from the literature, ii) the relative degree of importance for each factor in the implementation of TCIs, and iii) the feasibility of addressing each factor with implementation strategies for TCIs.

The overall objective is to derive a prioritized list by degree of importance and feasibility of the factors influencing the implementation of TCIs.

METHODS

All methods used to carry out this study are in accordance with relevant published guidelines and regulations for the Delphi technique, and this report of the study followed the guidelines for reporting Delphi studies (18-21).

ETHICAL APPROVAL AND CONSENT TO PARTICIPATE

Ethical approval was granted by the Maastricht University Faculty of Health, Medicine, & Life Sciences Ethics Committee (approval no. FHML-REC/2020/088). Informed consent was obtained from all participants prior to the study and by including a consent statement in each survey round as the initial question, whereby participants needed to agree in order to progress.

MODIFIED DELPHI STUDY APPROACH

A three-step modified Delphi study was conducted with a panel of international experts in the fields of implementation of innovations, transitional care, and long-term care (LTC) between July 2020 and March 2021. The method consisted of two sequential rounds of an online survey and a final group discussion session performed through an online video-conferencing platform.

In this study, the Delphi technique was used as a practical and iterative method to obtain broad perspectives of an experienced mix of experts (in this case, from different countries and, therefore, long-term healthcare system backgrounds) and to achieve consensus in an area where there is not enough evidence (19, 22, 23). Specifically, a modified Delphi method was chosen. This approach differs from the classical one, because first the content for round one was based on pre-determined items derived from data collected from other resources prior to the Delphi study (in this case, a scoping review) and hence utilized close-ended questions (22-25). Second, the final round was held as a face-to-face group discussion session with the experts (22, 26-29). A rating approach was used, whereby a panel of experts anonymously took part in surveys in different rounds. The findings and feedback of round one led the development of round two, and the final expert consultation session was based on input from the previous two rounds (22).

PARTICIPANTS

The expert profile was defined as individuals with extensive research and/or real-life experience in development, implementation, and evaluation of transitional care innovations (programs, models, and interventions) in LTC settings; healthcare innovations; LTC; or implementation science. Purposive and snowball sampling methods were used to recruit experts internationally.

An initial list of potential experts was developed based on professional contacts of the research team, authors of 21 published studies on the implementation of TCIs (from a previously published scoping review; Fakha et al. 2021) (6), and established contacts from the 3rd UK Implementation Science Conference – July 2020. We aimed for a minimum of 20 participants, as generally recommended (30). Initially, 62 experts were purposively contacted, and an additional three potential experts were contacted as a result of the snowball technique. All experts were invited to participate in the study by sending them individual recruitment emails along with an invitation letter describing the overall study background, aims, and methodology.

DATA COLLECTION

Survey design and development

Qualtrics software, an online survey tool, was used to develop and conduct the survey. This entailed sending the different questionnaires of each round to the participants. A total of 25 factors identified from the results of a previously published scoping review study conducted by the research team (6) were used to develop the survey content. Three of these factors were split into two subparts for description clarity, and a final list of 28 factors was thus used in round one (see Table 1). The factors were grouped into the five domains of the Consolidated Framework for Implementation Research (CFIR): *intervention characteristics, outer setting, inner setting, characteristics of individuals, and process* (31). The survey consisted of three sections and explored the following for each of the 28 factors: *i) direction*

of influence as hindering or facilitating to the implementation of TCIs; ii) importance of influence to the implementation of TCIs; and iii) feasibility (easiness/difficulty) of addressing the factor with implementation strategies for TCIs (see Appendix 3A for survey round 1). The survey was piloted among the research team and amended accordingly.

Table 1. Predetermined factors	(n=28) for Delphi ro	und one grouped into	CFIR domains

CFIR Domain	Factors
Intervention (TCIs) Characteristics	 Targeted groups* (older persons as recipients of the TCIs) Complexity (perceived difficulty of TCIs' implementation) Relative advantage (perceived advantage and usefulness of the TCIs by stakeholders) Evidence strength and quality (evidence for TCIs' effects on older persons' outcomes)
Outer Setting	 Cosmopolitanism (degree to which an organization is networked with other external organizations) External policy (e.g., mandates and regulations supporting the implementation of TCIs) External incentives (e.g., national funding schemes or sponsorship supporting the implementation of TCIs)
Inner Setting†	 Networks and communications (communications within an organization, e.g., interdisciplinary teams) Culture (organizational norms, values) Relative priority (healthcare professionals'/staff's perception of the importance of the implementation of TCls) Leadership engagement (commitment and involvement of leaders with the implementation of TCls) Available resources (resources dedicated to the implementation of TCls) Access to knowledge and information on the TCls Information continuity* (exchange of medical data on the older person among caregivers and across organizations) Health information technology (HIT) systems* (e.g., electronic medical records to manage care)
Characteristics of Individuals	 Knowledge and beliefs of healthcare professionals about the TCIs Knowledge and beliefs of older persons about the TCIs Role* of healthcare professionals/staff in implementing the TCIs Skills and competencies* of healthcare professionals/staff involved in implementation of TCIs Other personal attributes of healthcare professionals (values, motivation) Other personal attributes of older persons (values, health literacy)
Process	 Planning for the implementation of TCIs in advance Transition roles of frontline staff* (e.g., transition nurses who will implement the TCIs) Reflecting and evaluating the feedback and progress in the implementation of TCIs Measurement capability/data availability* (capacity for the implementation process monitoring, evaluation, and improvement) Engaging key stakeholders (individuals within the organization directly impacted by the TCIs) Engaging organizations, external context* (collaborations among various organizations and care providers involved in the implementation of TCIs) Engaging innovation participants (older persons, family, and informal caregivers who participate in the implementation of TCIs)

^{*}These factors are constructs from the Care Transitions Framework (CTF), which were added within the CFIR relative domains for the purpose of this study, check scoping review by Fakha et al. 2021 (6) for further details; †Inner setting is also referred to as the organizational context.

Round one – Personal links to the survey were sent in individual emails to the participants. A consent statement was the initial question, and participants needed to agree in order to progress. Participants were asked to rate each factor on a five-point Likert scale, either CHAPTER 3

in ascending order or from negative to positive (18). Ratings used per section were as follows: For direction of influence: 1) strongly hindering, 2) hindering, 3) neither hindering nor facilitating, 4) facilitating, 5) strongly facilitating. For importance of influence: 1) not important, 2) slightly important, 3) moderately important, 4) very important, 5) extremely important. For feasibility: 1) very difficult, 2) difficult, 3) neither difficult nor easy, 4) easy, 5) very easy. Moreover, participants were requested to provide additional comments in freetext boxes provided per each section, as well as overall suggestions for additional factors relevant to the implementation of TCIs. The survey required approximately 20 minutes to complete. Instructions were sent to participants on how to complete the survey, and they were given two weeks to respond. A reminder was sent to participants who did not complete the survey within the two-week period.

Round two - Survey round two was conducted online and in a manner similar to that for round one. Only participants who completed round one of the survey were invited to round two. This survey included factors that did not reach consensus from round one and additional factors that were suggested by participants, an established approach using the Delphi technique (22, 23). The definitions of a few factors were revised based on comments of participants in round one (see Appendix 3B for survey round 2). Participants were asked to again rate the factors using the same method as round one, but with knowledge of their individual ratings and the group ratings for each factor from the first round. In addition, a summary report on the results of round one was provided to all participants prior to the second round.

Final round: expert consultation session

The final round was comprised of two online video call meetings lasting two hours each, and performed in the same manner and using the same content, through a data-protected, web-based conferencing platform. All participants of round two were invited to join and were assigned to either of the two meetings according to their availability. In order to limit bias, AF and BdB facilitated both sessions, and TvA was an observant who also intervened when necessary to ensure participation from all experts. The sessions' discussions were recorded and later transcribed. The goal of these sessions was to allow interaction between the experts and provide further clarifications on the overall results from previous rounds. The specific aim, determined by the earlier results, was to *i*) *narrow down the important factors to the "must have" factors, ii*) obtain further insights on the feasibility of addressing the important factors, and *iii*) receive suggestions for potentially relevant implementation strategies. Initially, the results of rounds one and two were presented, and then participants were used to guide the discussion in order to allow for further deliberations, as follows:

- a) What are your views on the top factors? And if you were to choose 5 "must-have" factors, what would they be?
- b) Please can you explain why the majority of factors that reached consensus on
feasibility (including top important ones) were rated as difficult to address when developing implementation strategies? Any advice on how to approach this? Which strategies would you suggest to tackle each factor?

DATA ANALYSIS

Responses were analyzed after the completion of each survey round, and rating scores were calculated as percentages using SPSS Statistics 25 software. Consensus was determined as reached if over 70% of the respondents rated the *i*) direction of influence of each factor as either 'strongly hindering'/'hindering' (combined score), 'strongly facilitating'/facilitating' (combined score), or neither. Similarly, consensus on the *ii*) importance of factors was reached if either combined scores for 'extremely important'/'very important' or combined scores for 'moderately important'/'slightly important' or 'not important' were over 70%, and for *iii*) feasibility 'very easy'/'easy', 'very difficult'/'difficult', or 'neither'. This level of agreement was used and considered appropriate in previous Delphi studies (23, 24, 32-34). Free-text comments from rounds one and two were analyzed thematically and discussed among the research team to identify any additional factors or to rephrase and clarify the definitions.

Thematic analysis: final expert consultation session

The transcripts of the two final meetings were compiled together and analyzed thematically by three researchers (AF, BdB, TvA) following the six-step method described by Braun and Clarke (2006) (35, 36). An inductive form of thematic analysis was performed, and the codes created were data-driven. Following data coding, themes were developed and then reviewed iteratively. A thematic map was developed, and a clear delineation of the final themes was discussed and agreed upon by the research team.

RESULTS

The overall study design, number of participants, and results per round are summarized in Figure 1.

ROUND ONE

Initially, 32 experts agreed to participate in the study (round one), out of which 29 responded to the first survey (45% response rate, based on initial number of invited participants). Table 2 presents the participant demographics and professional backgrounds. More than half of the participants had at least 10 years of experience. The majority had a current role in research, mainly in implementation science and transitional care.





'Final number of factors that reached consensus is a cumulative build up between the consecutive rounds.

	Frequency
Country	
Australia	1
Belgium	1
Canada	2
Germany	1
Netherlands	11
Singapore	1
Sweden	1
Switzerland	1
UK	5
USA	5
Education level	
Master's	4
PhD	25
Current role	
Academia/research	26
Practice	5
Both	2
Area of expertise	
Transitional care	12
Long-term care	11
Healthcare innovations	10
Implementation science	14
Years of experience	
3 to 5 years	2
5 to 10 years	10
10 years and above	17

Table 2. Participant demographics (n=29)

*Some participants are experts in more than one area.

Eighteen factors out of 28 reached consensus on the direction of influence; however, only one factor (complexity) was generally seen as a barrier, while 17 were seen as facilitators (see Appendix 3C). Fourteen factors reached consensus on importance (see Table 3), with all but one factor, rated as very/extremely important in influencing the implementation of TCIs. Engaging key stakeholders ranked as the most important influencing factor. Nine factors reached consensus on feasibility, with only one factor (planning) rated as easy/very easy to address with implementation strategies (see Table 4). The organization's culture surpassed the other factors as most difficult.

Thematic analysis of the free-text comments in the first round indicated that the direction of influence for the factors was very difficult to assess. The participants mentioned that factors can behave differently according to the specific context where TCIs are implemented. Therefore, it was hard to judge the ultimate influence of each factor. *"Factors that hinder can paradoxically also be factors that facilitate and vice versa" (Expert 8, round 1). "Whether these factors are hindering or facilitating depends highly on the specific nature of this factor in the organization, so culture can be facilitating if there is an innovative culture, but hindering if there is a conservative culture without openness to innovation" (Expert 7, round 1).*

Moreover, the experts identified an additional seven factors to explore for consensus in the consecutive round. These factors were recognized across the five CFIR domains and included power of decision-makers, sense of urgency, adoption of change in work processes, financing of TCIs' implementation, inter-organizational collaborations, previous experiences with implementation of innovations, and co-design of the TCIs (see Appendix 3B for survey round 2). According to the experts' comments, the definitions of some factors were revised.

ROUND TWO

Twenty-eight of the 29 participants, who completed round one, completed this round (97% response rate). In this round, rating the factors' direction of influence was omitted. The consensus results from round one were skewed mostly to one direction (facilitating) and hence were ruled as of low relevance and non-conclusive by the research team.

A further nine factors reached consensus as very/extremely important and one as slightly/ moderately. Additionally, six out of the seven newly added factors reached consensus as very/extremely important to the implementation of TCIs, with financing of TCIs' implementation ranked as highest (see Table 3). Ten additional factors achieved consensus on feasibility in this round, with leadership engagement as the most difficult and transition roles as a neutral factor (see Table 4). A further two of the seven new factors (financing of TCIs' implementation, adoption of change in work processes) reached consensus as difficult/very difficult.

Table 3. Factors that reached consensus on importance of influence on the implementation of TCIs, in
order of ranking

Factor	Rating: Very/Extremely Important (Consensus level in %)	CFIR Domain
Round one		
Engaging key stakeholders	97	Process
Leadership engagement	93	Inner setting
Available resources	93	Inner setting
Relative priority	86	Inner setting
Relative advantage	79	Intervention
		characteristics
External incentives	79	Outer setting
Transition roles – frontline staff	76	Process
Skills and competencies	72	Characteristics of individuals
Role	72	Characteristics of individuals
Planning	72	Process
Knowledge and beliefs of healthcare professionals about the TCIs	72	Characteristics of individuals
Culture	72	Inner setting
Complexity	72	Intervention characteristics
	Rating: Slightly/Moderately Important (Consensus level in %)	
Other personal attributes of older persons	72	Characteristics of individuals
	Rating: Very/Extremely Important (Consensus level in %)	
Leadership engagement*	100	Inner setting
Information continuity	96	Inner setting
Financing of TCIs' implementation	96	Inner/outer setting
HIT systems	93	Inner setting
Access to knowledge and information	89	Inner setting
Engaging organizations, external context	89	Process
Sense of urgency	89	Inner setting
Reflecting and evaluating	86	Process
Other personal attributes of healthcare professionals	82	Characteristics of individuals
Adoption of change in work processes	82	Inner setting
Networks and communications	79	Inner setting
Inter-organizational collaborations	79	Inner/outer setting
Codesign of TCIs	79	Intervention
3	-	characteristics
Power of decision-makers	75	Inner/outer setting
Measurement capability/data availability	75	Process
External policy	71	Outer setting
External policy		0
	Rating: Slightly/Moderately Important (Consensus level in %)	

 $^{*}\mbox{Definition}$ revised for round two, and therefore rating for this factor was repeated.

Factor	Rating: Difficult/Very Difficult (Consensus level in %)	CFIR Domain
Round one		
Culture	100	Inner setting
HIT systems	93	Inner setting
Complexity	86	Intervention characteristics
External incentives	83	Outer setting
Networks and communications	76	Inner setting
External policy	76	Outer setting
Available resources	76	Inner setting
Other personal attributes of healthcare professionals	72	Characteristics of individuals
	Rating: Easy/Very Easy (Consensus level in %)	
Planning	76	Process
Round two		
	Rating: Difficult/Very Difficult (Consensus level in %)	
Leadership engagement	93	Inner setting
Engaging organizations, external context	93	Process
Relative priority	86	Inner setting
Information continuity	86	Inner setting
Other personal attributes of older persons	89	Characteristics of individuals
Financing of TCIs' implementation	89	Inner/outer setting
Cosmopolitanism	82	Outer setting
Adoption of change in work processes	82	Inner setting
	Rating: Easy/Very Easy (Consensus level in %)	
Access to knowledge and information	89	Inner setting
	Neither Easy nor Difficult (Consensus level in %)	
Transition roles – frontline staff	75	Process

 Table 4. Factors that reached consensus on feasibility (easy/difficult to address with implementation strategies), in order of ranking

FACTORS OF HIGHEST IMPORTANCE

Of the total 30 factors that reached consensus on the importance of influence following rounds one and two, the majority were linked to the inner setting. Within this domain, leadership engagement, availability of resources including HIT systems, and information continuity between care providers had the highest consensus levels on importance as compared to other factors such as the organizational culture. Whereas the engagement of stakeholders and organizations/external context was of highest importance within the implementation process and exceeded other factors, such as planning, reflecting and evaluating, and transition roles. In comparison, factors (skills and competencies, role, knowledge and beliefs) related to the characteristics of individuals had a lower level of consensus on their importance. Moreover, the personal attributes of older persons such as their motivation, values, or intellectual ability were rated as slightly/moderately important,

CHAPTER 3

while factors relating to healthcare professionals were seen as very/extremely important. As for the intervention characteristics, the relative advantage and benefits of the innovation as perceived by stakeholders (older persons and healthcare providers) as well as the degree of involvement of these stakeholders in its design prior to implementation were the most important factors. Alternatively, the demonstrated evidence strength and quality of the TCIs appear to be of least importance to influence the implementation. In contrast, external incentives and policy supporting the TCIs' implementation and national financing structures, such as a healthcare services reimbursement system, were the important factors within the outer setting.

A final list of the 11 factors that ranked as most important with consensus of 85% and above is presented in Table 5. Once more, these key factors were predominantly related to the inner setting, and only three were linked to the implementation process. The engagement of leaders and key stakeholders was confirmed by the experts as essential in influencing the implementation of TCIs. "If key stakeholders are in favor of an innovation, good communication can really help, but when they are not in favor, it can really hinder an implementation process" (Expert 7, round 1). Nevertheless, the continuity of information and communication across multiple organizations came in third place, which could be explained by the nature of transitional care involving several care settings. "It is difficult to coordinate care that goes beyond the boundaries of a specific organization" (Expert 7, round 1). Moreover, the availability of organizational resources as well as the existing financing structures to support the implementation of TCIs were important influencers. "In transitional care also the reimbursement system in healthcare can play an important role. If an intervention does not fit the current system, this can be a real challenge for the implementation process" (Expert 7, round 1). "Lack of money and lack of management support can stop efforts very quickly" (Expert 5, round 1).

FEASIBILITY

Around only half (54%) of the total number of factors reached consensus on feasibility, with the majority rated as difficult to address with implementation strategies and repeatedly linked to the organizational context. An attempt to address the organizational culture was regarded by experts as topmost difficult and as the least feasible approach to take. "And since we're talking about implementing a new model or some sort of a change, it's always a culture change conversation, and there are a lot of things involved in changing culture..." (Expert 12, consultation session). Moreover, experts indicated that it was challenging to assess the feasibility for each factor, since it depends on the context where the TCIs' implementation is happening. "The difficulty to address these items in practice can vary a lot from project to project" (Expert 5, round 1). "Ideally, each site should identify the areas that are strengths and challenges in relation to the intervention, and from there identify which strengths they can leverage" (Expert 18, round 1). In addition, consensus on the feasibility

of specific factors revealed that, while these factors are very important in influencing the implementation of TCIs, it is most likely difficult to consider, control, or change them with strategies (see Table 5).

Pric	rity Factors*	Feasibility	CFIR Domain
1.	Leadership engagement	Difficult/very difficult	Inner setting
2.	Engaging key stakeholders	No consensus	Process
3.	Information continuity	Difficult/very difficult	Inner setting
4.	Financing of TCIs' implementation	Difficult/very difficult	Inner/outer setting
5.	Available resources	Difficult/very difficult	Inner setting
6.	HIT systems	Difficult/very difficult	Inner setting
7.	Access to knowledge and information	Easy/very easy	Inner setting
8.	Engaging organizations, external context	Difficult/very difficult	Process
9.	Sense of urgency	No consensus	Inner setting
10.	Relative priority	Difficult/very difficult	Inner setting
11.	Reflecting and evaluating	No consensus	Process

Table 5. Final list of most important factors and indication of feasibility⁺

*Factors are listed in descending ranking order; *factors with a consensus level \geq 85% were considered as most important, i.e., priority.

FINAL ROUND: EXPERT CONSULTATION SESSION

Fourteen experts participated in this session and two overarching themes emerged, which are described as follows:

Theme one: 'The Catalysts'

This theme describes a combination of temporal and interconnected factors that were seen as essential prerequisites for starting the implementation of TCIs. From the 11 key important factors from the previous rounds, the experts identified five factors that are the catalysts to launch any effort to implement TCIs. These factors were the sense of urgency, relative priority, financing and resources, leadership engagement, and engagement of key stakeholders across the continuum of transitional care. Sense of urgency was identified as a primary factor to induce any change within organizations and even to create priorities and allocate resources needed for the implementation of TCIs. Whereas relative priority was regarded as a *"stop/go"* flag for the implementation of TCIs, it also depends on from whom or where this priority is coming.

"First to get that sense of urgency and then it realigns priorities." (Expert 12, consultation session)

"And since we're talking about implementing a new model or some sort of a change. It is always a culture change conversation, and there are a lot of things involved in changing culture. Having people feel like this is important; maybe you realize the priorities in their head, but I think the urgency comes first, in my mind." (Expert 12, consultation session)

"So this is the sort of stop/go regardless of leadership engagement, regardless of HIT systems, regardless of result. You are not going to have resources, you are not going to have engagement unless something is the priority, so for me this is almost like a step before." (Expert 4, consultation session)

The experts reconfirmed that the engagement of leaders, key stakeholders such as representatives of multiple organizations, older persons, and family caregivers is the backbone for implementing TCIs. However, it is crucial to first identify the role and responsibility of each stakeholder in the implementation process and then to create the right engaging activity and sustain it.

"Engaging key stakeholders is a means to an end is kind of an initial, you know piece of it is a catalyst for all of the other things that happen up and down those levels." (Expert 9, consultation session)

"...that you have a successful implementation, and that is stakeholder engagement and the leadership engagement." (Expert 6, consultation session)

Furthermore, it was indicated that resources including HIT systems and funds would only be made available if leadership and key stakeholders are involved early on. As for the factor of financing the implementation of TCIs, it was discussed that reimbursement or financing structures could be varied in transitional care, especially when several organizations are involved. Therefore, the key element is to demonstrate the TCIs' value for care and the return on investment in order to feedback into the loop of leadership engagement and prioritizing its implementation. The proposed interrelationships among these catalyst factors are depicted in Figure 2.



Figure 2. Depiction of the interrelationships among the catalyst factors influencing the implementation of TCIs

Theme two: Alignment – 'The Driver'

This theme highlights the importance of aligning the motivation for change across various organizations and levels in the healthcare system. Experts implied that regardless of *'catalysts'* being present, the alignment of forces to drive the implementation of TCIs across the continuum of transitional care is a key issue, yet often the *"blind spot"*. Aligning the different priorities, interests, motivations, innovation readiness, financial incentives, and agendas of the organizations involved can help drive the implementation of TCIs. Moreover, experts agreed on the significance of considering at which level the implementation for implementing TCIs. Lastly, the experts believed that it is a crucial driver to ensure that the TCIs are in alignment with the older person's care needs.

Feasibility – The experts concurred that developing implementation strategies to address the important factors is largely dependent on the context, individuals involved, and the care continuum. Specifically, trying to overcome hindering factors linked to the organizational context was seen as a known challenge and hard to successfully address with implementation strategies. "Organizational inertia, culture, its also infrastructure, its processes are inert, is because what they've done has worked so far. If they're surviving it's because they've done something that, for whatever reason, has worked" (Expert 9, consultation session). However, experts highlighted that the focus could be shifted to creating strategies to induce change at the individual level, which may ultimately improve the organization's willingness to innovate.

DISCUSSION

Experts in this study prioritized 11 factors as the most important in the implementation of TCIs. Amongst these factors, the majority were organizational factors, primarily the leadership engagement, availability of resources, information continuity, sense of urgency, and relative priority. Moreover, engagement of stakeholders linked to the implementation process was seen as another priority factor. However, the study results demonstrated a prevalent agreement among experts on the difficulty to address these priority factors with implementation strategies. Nevertheless, ensuring the alignment of the organizations' interests, agendas, incentives, and priorities was established as a crucial "*stepping stone*" in implementing TCIs across the transitional care continuum. The current findings are congruent with earlier research indicating that organizational factors, chiefly leadership, resources, and communication, significantly influence the implementation of EBIs in healthcare settings (37-40).

In this study, experts concurred strongly that a high commitment of organizational leadership is the dominant factor in initiating the implementation of TCIs and supporting the overall change process. Furthermore, leadership has the ability to respond to a sense of urgency to innovate within an organization and keep it as a priority. Therefore, this suggested a versatile nature of leadership influence on the implementation of TCIs by being not only a precursor but also a probable moderator or mediator. This resonates with recent studies that recognized a contingent relationship between leadership influence and other implementation factors (37, 38, 41). The other two priority factors — resource allocation and engagement of key stakeholders (i.e., healthcare professionals and staff required for implementation of TCIs) — were acceded by our Delphi panel as dependent on the existence of a supportive leadership influence. Correspondingly, Gifford et al. describe the potential effective contribution of leadership to promote a successful implementation of evidence in healthcare practice (42).

Our panel agreed that engagement of key stakeholders was significant in the implementation of TCIs at all levels of the transitional care continuum. Similarly, engaging multidisciplinary healthcare teams and key staff in various care settings has been widely reported as an integral element and a necessary process activity for implementing innovations in transitional care and LTC in general (6, 38, 39, 43-46). Nevertheless, despite the well-known importance of stakeholders' engagement in implementation, there is still vagueness and limited evidence on its definition, who qualifies as a stakeholder, and which best practices to employ (47, 48).

In relation to this, and to our surprise, the importance of engaging the older persons and their family or informal caregivers in the implementation of TCIs was not something our

experts reached consensus on. We would have expected that factors related to the older persons including their knowledge, perceptions, attitudes toward and value placed on the TCIs' services, personal care needs, and an overall consideration of *"what matters to them?"* would be prioritized as very important by the experts. Acknowledging the characteristics and interests of the older persons was revealed as instrumental in other studies describing the process of uptake and implementation of interventions in transitional care (38, 49). Although person-centered transitional care, whereby TCIs are tailored to older persons' needs and preferences, is generally seen as important (50); our results indicate that involving the target group in the implementation of TCIs is less evident. Likewise, Olsen et al. highlight that engaging the older persons and listening to their needs and wishes are fundamental factors in delivering transitional care interventions, yet there appears to be other significant and overlooked constraints at the organizational and system levels (51).

The feasibility of addressing the agreed upon priority factors with implementation strategies was concurred by the experts as mostly challenging, particularly for the organizational factors. Contrary to our expectations, these results do not inform the development of strategies for implementation of TCIs. Notably, a number of taxonomies and compilations of strategies were developed to aide in implementing EBIs in healthcare (52-55). Moreover, some of these strategies were matched to the relative influencing factors in general healthcare settings (54, 56). In addition, although organizational leadership was rated as difficult to address in this Delphi study, there is evidence on an emerging strategy: the leadership and organizational change for implementation (LOCI) (57). Among its aspects, LOCI focuses on leveraging the leaders' readiness for implementing EBIs, training them on how to overcome implementation barriers, and promoting them to be proactive and create a shared vision within the organization (57). On the other hand, the body of research on implementation strategies is not specific to transitional care, although it is starting to be utilized in implementing certain TCIs (43). Therefore, our findings add to the evolving literature by indicating that practitioners and researchers in the field of implementing TCIs perceive that strategies should be tailored to the specific settings involved.

IMPLICATIONS FOR IMPLEMENTING INNOVATIONS IN TRANSITIONAL CARE

Practice – In light of our findings, we ask healthcare practitioners (leaders, managers, frontline staff, and other professionals) wanting to implement a TCI, to start by conducting a local needs assessment. It is crucial to know the inner settings of LTC organizations, as well as the inter-organizational differences and dynamics, as this is also a key message from previous studies (39, 58). To better assess the organizational readiness for putting a TCI into practice, we recommend utilizing our list of 'priority factors' as a starting point. Exploring these factors locally will provide an early essential awareness and knowledge of what will most likely help or hinder the implementation of a TCI. Based on the existing literature, we hereby provide hands-on suggestions for addressing the priority factors

when implementing a TCI (53, 54, 56). For example, having both intra/inter-organizational discussions among care providers and key stakeholders can help identify existing problems in care transitions of older persons between different settings, and hence create a sense of urgency or prioritize the need for a TCI as a solution (53, 56). This, in turn can be used to build a case and to present it to the leaders of LTC organizations, as to obtain support for implementation. Furthermore, engaged leaders can help secure required resources through practical ways by looking for funding options for initial implementation, such as responding for governmental calls to fund implementation of innovations in practice or restructuring organizational incentives. In addition, the creation of inter-organizational working groups of key stakeholders can assist in following through the implementation process of a TCI and making necessary decisions and adjustments (53).

Correspondingly and given the big role of organizational factors, we also highlight the potential value of using insights from four prominent organizational theories *(transaction cost economics, institutional theory, contingency theory, and resource dependency theory)* in implementing TCIs (59). For example, with keeping in mind the disparities across different healthcare systems, a healthcare manager can assess the transaction cost of implementing an intervention and consider the possibility of outsourcing a TCI's components or services to another institution. Otherwise as denoted by resource dependency theory, healthcare managers and leaders can establish inter-organizational partnerships and alliances to acquire resources needed. Moreover, as per both institutional and contingency theories, healthcare managers and leaders can promote the adoption and implementation of a TCI within their organization by copying successful innovative behaviours of other organizations in the environment, and boost the organization's agility to respond to external factors.

Future research – Further investigation of the prioritized factors in the actual implementation of TCIs in practice can provide a better understanding of how they exist and interact. In addition, the development and testing of a set of tailored, effective, and feasible strategies to target these priority factors influencing the implementation of TCIs is required.

STRENGTHS AND LIMITATIONS

This study gathered consensus by drawing on an international panel of experts from the fields of innovation, implementation, and transitional care, which allowed to obtain focused perspectives. Moreover, the use of an online survey permitted a high response, and the final session with the experts was instrumental to understanding the consensus results and enriched the study.

Alternatively, there were some limitations. First, selection bias could play a role. A majority of the panel were scientists rather than professionals from practice, which may have led to an under-representation of insights from real-life context. This could also explain the panel

members' difficulties in assessing implementation factors' direction of influence and their focus on organizational factors, as their direct care experience might have been limited and/ or mainly in the past. Furthermore, the majority of experts came from European healthcare systems. Insights from other alternative models of healthcare systems (e.g. the USA) were under-represented in our panel of experts, thus limiting the applicability of the findings. Second, repeating the survey with other panel experts might have led to other results. However, to improve the reliability, we aimed for a large sample of experts from various backgrounds and countries. Also, we performed the two survey rounds in a consistent manner. Third, we provided the group ratings from round one in the consecutive round, which could be viewed as a pressure to obtain consensus. However, one can also argue that panel members are entitled to receiving core results, besides we followed the Delphi methodology carefully, used findings of a previous literature study to inform the survey, and based our work on an established implementation framework. Lastly, the final round being a gualitative group discussion, holds the limitation pertaining to typical group dynamics and power/confidence of expression. However, the session also enriched our study and facilitators for the session made sure that all participants were heard, and encouraged an open discussion.

CONCLUSIONS

Though many factors are relevant in the implementation of TCIs, experts conceded that the priority factors in the implementation of TCIs are leadership engagement, sense of urgency to innovate, relative priority given to a TCI, availability of organizational resources, and engagement of key stakeholders. Results from our study enable the selection of relevant strategies for implementation of TCIs, yet special attention should be given to inter-organizational factors, which are seen as difficult to address, as well as the interrelationships revealed between these factors. This study provides novel guidance for healthcare researchers and practitioners, opting to improve transitional care for older persons, to better navigate the implementation process of innovations, and deter efforts based on intuition rather than evidence.

REFERENCES

- Baxter R, Shannon R, Murray J, O'Hara JK, Sheard L, Cracknell A, et al. Delivering exceptionally safe transitions of care to older people: a qualitative study of multidisciplinary staff perspectives. BMC Health Serv Res. 2020;20(1):780.
- Hirschman KB, Hodgson NA. Evidence-Based Interventions for Transitions in Care for Individuals Living With Dementia. Gerontologist. 2018;58(suppl_1):S129-s40.
- 3. Abraham S, Menec V. Transitions Between Care Settings at the End of Life Among Older Homecare Recipients: A Population-Based Study. Gerontol Geriatr Med. 2016;2:2333721416684400-.
- Wee SL, Vrijhoef HJ. A conceptual framework for evaluating the conceptualization, implementation and performance of transitional care programmes. J Eval Clin Pract. 2015;21(2):221-8.
- Prusaczyk B, Fabbre V, Morrow-Howell N, Proctor E. Understanding transitional care provided to older adults with and without dementia: A mixed methods study. Int J Care Coord. 2020;23(1):14-23.
- 6. Fakha A, Groenvynck L, de Boer B, van Achterberg T, Hamers J, Verbeek H. A myriad of factors influencing the implementation of transitional care innovations: a scoping review. Implement Sci. 2021;16(1):21.
- Dresden SM, Hwang U, Garrido MM, Sze J, Kang R, Vargas-Torres C, et al. Geriatric Emergency Department Innovations: The Impact of Transitional Care Nurses on 30-day Readmissions for Older Adults. Acad Emerg Med. 2020;27(1):43-53.
- 8. Finlayson K, Chang AM, Courtney MD, Edwards HE, Parker AW, Hamilton K, et al. Transitional care interventions reduce unplanned hospital readmissions in high-risk older adults. BMC Health Serv Res. 2018;18(1):956.
- 9. Hwang U, Dresden SM, Rosenberg MS, Garrido MM, Loo G, Sze J, et al. Geriatric Emergency Department Innovations: Transitional Care Nurses and Hospital Use. J Am Geriatr Soc. 2018;66(3):459-66.
- Sezgin D, Hendry A, Liew A, O'Donovan M, Salem M, Carriazo AM, et al. Transitional palliative care interventions for older adults with advanced non-malignant diseases and frailty: a systematic review. J Integr Care. 2020;28(4):387-403.
- Naylor MD, Bowles KH, McCauley KM, Maccoy MC, Maislin G, Pauly MV, et al. High-value transitional care: translation of research into practice. J Eval Clin Pract. 2013;19(5):727-33.
- Naylor MD, Shaid EC, Carpenter D, Gass B, Levine C, Li J, et al. Components of Comprehensive and Effective Transitional Care. J Am Geriatr Soc. 2017;65(6):1119-25.
- Lutz BJ, Reimold AE, Coleman SW, Guzik AK, Russell LP, Radman MD, et al. Implementation of a Transitional Care Model for Stroke: Perspectives From Frontline Clinicians, Administrators, and COMPASS-TC Implementation Staff. Gerontologist. 2020;60(6):1071-84.
- 14. Prusaczyk B, Olsen MA, Carpenter CR, Proctor E. Differences in transitional care provided to patients with and without dementia. J Gerontol Nurs. 2019;45(8):15-22.
- Markle-Reid M, Valaitis R, Bartholomew A, Fisher K, Fleck R, Ploeg J, et al. An integrated hospital-to-home transitional care intervention for older adults with stroke and multimorbidity: A feasibility study. J Comorb. 2020;10:2235042X19900451.
- Khalil H, Kynoch K. Implementation of sustainable complex interventions in health care services: the triple C model. BMC Health Serv Res. 2021;21(1):143.
- 17. Weeks LE, Barber B, MacDougall ES, Macdonald M, Martin-Misener R, Warner G. An exploration of Canadian transitional care programs for older adults. Healthc Manage Forum. 2021;34(3):163-8.
- Toma C, Picioreanu I. The Delphi Technique: methodological considerations and the need for reporting guidelines in medical journals. Int J Public Health Res. 2016;4(6):47-59.
- Hasson F, Keeney S, McKenna H. Research guidelines for the Delphi survey technique. J Adv Nurs. 2000;32(4):1008-15.
- Diamond IR, Grant RC, Feldman BM, Pencharz PB, Ling SC, Moore AM, et al. Defining consensus: a systematic review recommends methodologic criteria for reporting of Delphi studies. J Clin Epidemiol. 2014;67(4):401-9.
- Junger S, Payne SA, Brine J, Radbruch L, Brearley SG. Guidance on Conducting and REporting DElphi Studies (CREDES) in palliative care: Recommendations based on a methodological systematic review. Palliat Med. 2017;31(8):684-706.
- 22. Keeney S, Hasson F, McKenna H. The Delphi Technique in Nursing and Health Research. first ed. UK: John Wiley & Sons; 2011.
- Havers SM, Martin E, Wilson A, Hall L. Implementation of government-directed policy in the hospital setting: a modified Delphi study. Health Res Policy Syst. 2019;17(1):91.
- Fergie L, Campbell KA, Coleman-Haynes T, Ussher M, Cooper S, Coleman T. Stop smoking practitioner consensus on barriers and facilitators to smoking cessation in pregnancy and how to address these: A modified Delphi survey. Addict Behav Rep. 2019;9:100164.
- Hsu C-C, Sandford BA. The Delphi technique: making sense of consensus. Pract Assess Res Evaluation. 2007;12(10).
- 26. Eubank BH, Mohtadi NG, Lafave MR, Wiley JP, Bois AJ, Boorman RS, et al. Using the modified Delphi method to

establish clinical consensus for the diagnosis and treatment of patients with rotator cuff pathology. BMC Med Res Methodol. 2016;16:56.

- van Vliet DC, van der Meij E, Bouwsma EV, Vonk Noordegraaf A, van den Heuvel B, Meijerink WJ, et al. A modified Delphi method toward multidisciplinary consensus on functional convalescence recommendations after abdominal surgery. Surg Endosc. 2016;30(12):5583-95.
- Schneider P, Evaniew N, Rendon JS, McKay P, Randall RL, Turcotte R, et al. Moving forward through consensus: protocol for a modified Delphi approach to determine the top research priorities in the field of orthopaedic oncology. BMJ Open. 2016;6(5):e011780.
- Woodcock T, Adeleke Y, Goeschel C, Pronovost P, Dixon-Woods M. A modified Delphi study to identify the features of high quality measurement plans for healthcare improvement projects. BMC Med Res Methodol. 2020;20(1):8.
- Okoli C, Pawlowski SD. The Delphi method as a research tool: an example, design considerations and applications. Information & Management. 2004;42(1):15-29.
- Damschroder LJ, Aron DC, Keith RE, Kirsh SR, Alexander JA, Lowery JC. Fostering implementation of health services research findings into practice: a consolidated framework for advancing implementation science. Implement Sci. 2009;4(1):50.
- Fisher RJ, Walker MF, Golton I, Jenkinson D. The implementation of evidence-based rehabilitation services for stroke survivors living in the community: the results of a Delphi consensus process. Clin Rehabil. 2013;27(8):741-9.
- Jorm AF. Using the Delphi expert consensus method in mental health research. Aust N Z J Psychiatry. 2015;49(10):887-97.
- Vogel C, Zwolinsky S, Griffiths C, Hobbs M, Henderson E, Wilkins E. A Delphi study to build consensus on the definition and use of big data in obesity research. Int J Obes 2019;43(12):2573-86.
- 35. Braun V, Clarke V. Using thematic analysis in psychology. Qual Res Psychol. 2006;3(2):77-101.
- Nowell LS, Norris JM, White DE, Moules NJ. Thematic Analysis: Striving to Meet the Trustworthiness Criteria. Int J Qual Methods. 2017;16(1):1609406917733847.
- 37. Li S-A, Jeffs L, Barwick M, Stevens B. Organizational contextual features that influence the implementation of evidence-based practices across healthcare settings: a systematic integrative review. Syst Rev. 2018;7(1):72.
- Sarkies M, Long JC, Pomare C, Wu W, Clay-Williams R, Nguyen HM, et al. Avoiding unnecessary hospitalisation for patients with chronic conditions: a systematic review of implementation determinants for hospital avoidance programmes. Implement Sci. 2020;15(1):91.
- Bunn F, Goodman C, Corazzini K, Sharpe R, Handley M, Lynch J, et al. Setting Priorities to Inform Assessment of Care Homes' Readiness to Participate in Healthcare Innovation: A Systematic Mapping Review and Consensus Process. Int J Environ Res Public Health. 2020;17(3):987.
- Nurjono M, Shrestha P, Ang IYH, Shiraz F, Yoong JS-Y, Toh S-AES, et al. Implementation fidelity of a strategy to integrate service delivery: learnings from a transitional care program for individuals with complex needs in Singapore. BMC Health Serv Res. 2019;19(1):177.
- 41. Aarons GA, Ehrhart MG, Farahnak LR, Sklar M. Aligning Leadership Across Systems and Organizations to Develop a Strategic Climate for Evidence-Based Practice Implementation. Annu Rev Public Health. 2014;35(1):255-74.
- 42. Gifford WA, Squires JE, Angus DE, Ashley LA, Brosseau L, Craik JM, et al. Managerial leadership for research use in nursing and allied health care professions: a systematic review. Implement Sci. 2018;13(1):127.
- 43. Yip O, Huber E, Stenz S, Zullig LL, Zeller A, De Geest SM, et al. A Contextual Analysis and Logic Model for Integrated Care for Frail Older Adults Living at Home: The INSPIRE Project. Int J Integr Care. 2021;21(2):9.
- 44. Yeager VA, Wharton MK, Monnette A, Price-Haywood EG, Nauman E, Angove RSM, et al. Non-Face-to-Face Chronic Care Management: A Qualitative Study Assessing the Implementation of a New CMS Reimbursement Strategy. Popul Health Manag. 2018;21(6):454-61.
- 45. Williams MV, Li J, Hansen LO, Forth V, Budnitz T, Greenwald JL, et al. Project BOOST implementation: lessons learned. South Med J. 2014;107(7):455-65.
- 46. Brody AA, Arbaje AI, DeCherrie LV, Federman AD, Leff B, Siu AL. Starting Up a Hospital at Home Program: Facilitators and Barriers to Implementation. J Am Geriatr Soc. 2019;67(3):588-95.
- 47. Goodman MS, Sanders Thompson VL. The science of stakeholder engagement in research: classification, implementation, and evaluation. Transl Behav Med. 2017;7(3):486-91.
- 48. Norris JM, White DE, Nowell L, Mrklas K, Stelfox HT. How do stakeholders from multiple hierarchical levels of a large provincial health system define engagement? A qualitative study. Implement Sci. 2017;12(1):98.
- 49. Gesell SB, Bushnell CD, Jones SB, Coleman SW, Levy SM, Xenakis JG, et al. Implementation of a billable transitional care model for stroke patients: the COMPASS study. BMC Health Serv Res. 2019;19(1):978.
- 50. Groenvynck L, Fakha A, de Boer B, Hamers JPH, van Achterberg T, van Rossum E, et al. Interventions to Improve the Transition from Home to a Nursing Home: A Scoping Review. Gerontologist. 2021.
- Olsen CF, Debesay J, Bergland A, Bye A, Langaas AG. What matters when asking, "what matters to you?" perceptions and experiences of health care providers on involving older people in transitional care. BMC Health Serv Res. 2020;20(1):317.
- 52. Cochrane. Effective Practice and Organisation of Care (EPOC). EPOC Taxonomy; 2015 [Available from: epoc.

cochrane.org/epoc-taxonomy]

- Powell BJ, Waltz TJ, Chinman MJ, Damschroder LJ, Smith JL, Matthieu MM, et al. A refined compilation of implementation strategies: results from the Expert Recommendations for Implementing Change (ERIC) project. Implement Sci. 2015;10:21.
- 54. Waltz TJ, Powell BJ, Fernandez ME, Abadie B, Damschroder LJ. Choosing implementation strategies to address contextual barriers: diversity in recommendations and future directions. Implement Sci. 2019;14(1):42.
- 55. Bartholomew LK, Parcel GS, Kok G, Gottlieb NH, Schaalma HC, Markham CC, et al. Planning health promotion programs: an intervention mapping approach: Jossey-Bass; 2006.
- 56. Waltz TJ, Powell BJ, Matthieu MM, Damschroder LJ, Chinman MJ, Smith JL, et al. Use of concept mapping to characterize relationships among implementation strategies and assess their feasibility and importance: results from the Expert Recommendations for Implementing Change (ERIC) study. Implement Sci. 2015;10:109.
- 57. Aarons GA, Ehrhart MG, Farahnak LR, Hurlburt MS. Leadership and organizational change for implementation (LOCI): a randomized mixed method pilot study of a leadership and organization development intervention for evidence-based practice implementation. Implement Sci. 2015;10:11.
- Cammer A, Morgan D, Stewart N, McGilton K, Rycroft-Malone J, Dopson S, et al. The Hidden Complexity of Long-Term Care: how context mediates knowledge translation and use of best practices. Gerontologist. 2014;54(6):1013-23.
- 59. Birken SA, Bunger AC, Powell BJ, Turner K, Clary AS, Klaman SL, et al. Organizational theory for dissemination and implementation research. Implement Sci. 2017;12(1):62.

APPENDICES

Appendix 3A. Modified Delphi survey - Round 1Appendix 3B. Modified Delphi survey - Round 2Appendix 3C. List of factors that reached consensus on direction of influence on the implementation of TCls, from round one

Appendix 3A. Modified Delphi survey - Round 1

Dear Expert,

Thank you for agreeing to participate in the current Modified Delphi study survey (round one) in order to help achieve consensus on the top priority barriers and facilitators that influence the implementation of innovations in transitional care.

We would like to remind you quickly of key points before starting the survey: >"Transitional care is defined as a set of actions designed to ensure the coordination and continuity of healthcare as patients transfer between different locations or different levels of care within the same location".

>We refer throughout this survey to Transitional Care (TC) Innovation, as any *intervention, model, or program* which has been developed with a goal to improve or prevent care transitions for the <u>older population/persons (</u>> 65 years old) between different long-term care settings.

- The survey is developed based on the five domains: (Interventions characteristics, Outer setting, Inner (organizational) setting, Characteristics of Individuals, Process) of the CFIR (Consolidated Framework for Implementation Research) and selected constructs (factors) from this framework and the CTF (Care Transitions Framework).
- The survey consists of three sections and will explore the following concepts for each of the 25 selected factors (3 factors are split into parts a/b), and definitions of factors adapted to transitional care are provided in the survey:

>Section A: to rate the direction of influence for each of the 25 factors as more a barrier or facilitator to the implementation of transitional care innovations
>Section B: to rate the importance of influence of each of the 25 factors on the implementation of transitional care innovations
>Section C - to indicate the feasibility (easiness / difficulty) to address each of the 25 factors in the development of implementation strategies for transitional care innovations

 General Instructions: The survey will take approximately 15-20 minutes to complete. Your participation is voluntary, and you can withdraw from the study during the survey round without comment or penalty by closing the survey link. By clicking (yes, I consent) below, you will indicate that you have fully read and understood the complete information document provided to you earlier regarding this study with the participant invitation/recruitment email. Your responses to the survey are automatically saved as you go through the questions, and at any moment you can close the survey link and continue it later by using your same personal link. Please complete the demographics section at the end of the survey. All data collected and processed will be kept anonymous, confidential, and stored on a password-protected database at Maastricht University. If you have any questions or technical issues please contact: Amal Fakha (main researcher, Department of Health Services Research, Faculty of Health Medicine and Life Sciences, Maastricht University, the Netherlands) by sending an email to a.fakha@maastrichtuniversity.nl. On the behalf of the research team we thank you greatly for your participation.

- I hereby agree to participate and undertake this survey.
- o Yes, I consent.

Section A - Direction of Influence: factors can either hinder or facilitate the implementation of an innovation; for example: i) a match between the Targeted Groups (factor) and the Transitional Care innovation features can be facilitating to the implementation, while a mismatch can have a hindering effect. ii) Accessibility to Information and Knowledge (factor) about a Transitional Care innovation can have a facilitating effect on its implementation, while inaccessibility can be hindering.

1) Rate for each of the 25 factors listed below if it acts more often as **hindering or facilitating** to the implementation of Transitional Care Innovations:

	Strongly hindering	Hindering	Neither hindering nor facilitating	Facilitating	Strongly facilitating
Domain I - Intervention (Transitional Care Innovation) Characteristics Factor 1 - Targeted Groups: patients/ older population who are the intended recipients or beneficiaries of the transitional care innovation.	0	0	0	0	0
Factor 2 - Complexity: perceived difficulty of implementation, reflected by duration, scope, radicalness, disruptiveness, centrality, and intricacy and number of steps required to implement.	0	0	0	0	0
Factor 3 - Relative Advantage: stakeholders' perception of the advantage (benefits and usefulness) of implementing the transitional care innovation versus an alternative solution.	0	0	0	0	0
Factor 4 - Evidence Strength & Quality: stakeholders' perceptions of the quality and validity of evidence (proven effectiveness) supporting the belief that the transitional care intervention will have desired outcomes (e.g. low readmission rates).	0	0	0	0	0
Domain II - Outer Setting Factor 5 - Cosmopolitanism: the degree to which an organization is networked with other external organizations.	0	0	0	0	0
Factor 6a - External Policy: a broad construct that includes external strategies (by government or other central entity) to spread transitional care innovations; including policy, regulations, laws, external mandates, legislative changes, recommendations, and guidelines.	0	0	0	0	0
Factor 6b - External Incentives: a broad construct that includes external strategies (by the government or other central entity) to spread transitional care innovations; including national funding schemes or governmental sponsorship.	0	0	0	0	0

	Strongly hindering	Hindering	Neither hindering nor facilitating	Facilitating	Strongly facilitating
Domain III - Inner (Organizational) Setting Factor 7 - Networks & Communications: the nature and quality of webs of social networks and of formal/ informal communications within an organization (e.g. interdisciplinary teams, coordination & communication among team members).	0	0	0	0	0
Factor 8 - Culture: norms, values, and basic assumptions of a given organization.	0	0	0	0	0
Factor 9 - Relative Priority: individuals' (healthcare professionals, staff within implementing team) shared perception of the importance of the implementation of a transitional care innovation within the organization.	0	0	0	0	0
Factor 10 - Leadership Engagement: commitment, involvement, and accountability of leaders and managers with the implementation of a transitional care innovation.	0	0	0	0	0
Factor 11 - Available Resources: the level of resources dedicated for the implementation and on-going operations of a transitional care innovation; including staffing levels, money, funding, training, education, physical space, equipment, and time.	0	0	0	0	0
Factor 12 - Access to Knowledge & Information: ease of access to digestible information and knowledge (e.g. mentoring, initial training) about the transitional care innovation and how to incorporate it into work tasks.	0	0	0	0	0
Factor 13 - Continuity: information continuity (e.g. patient information exchange, services & care planning) and relationship continuity, both with providers and patients/caregivers and across organizations.	0	0	0	0	0
Factor 14 - IT&HIT resources (HIT systems): electronic information management infrastructure and technologies (e.g. electronic health records) available to clinicians to manage patient care, data, and communications.	0	0	0	0	0
Domain IV - Characteristics of Individuals Factor 15a - Knowledge & Beliefs about the Intervention: healthcare professionals/staff within implementing team's beliefs, expectations, and familiarity with facts, truths, & principles related to the transitional care innovation.	0	0	0	0	0

	Strongly hindering	Hindering	Neither hindering nor facilitating	Facilitating	Strongly facilitating
Factor 15b - Knowledge & Beliefs about the Intervention: patients/older persons' attitudes toward and value placed on the transitional care innovation as well as awareness on its care services & goals.	0	0	0	0	0
Factor 16 - Role: individual's role (healthcare professionals, staff within implementing team) and responsibility for the transitional care innovation; including the degree of multiple or shared roles.	0	0	0	0	0
Factor 17 - Skills & Competencies: degree of relevant subject matter expertise, skills, and competencies within the implementing team, unit, and organization.	0	0	0	0	0
Factor 18a - Other Personal Attributes: healthcare professionals' other personal traits such as motivation levels, values, tolerance of ambiguity, critical attributes, intellectual ability, capacity, and learning style.	0	0	0	0	0
Factor 18b - Other Personal Attributes: patients/older persons' other personal traits such as health literacy, values, and acknowledgement of own care needs.	0	0	0	0	0
Domain V - (Implementation) Process Factor 19 - Planning: the degree to which a scheme or method of behavior and tasks for implementing an innovation are developed in advance, and the quality of those schemes or methods.	0	0	0	0	0
Factor 20 - Transition Roles (Frontline Staff): administrative staff, providers (within and outside the organization), e.g. frontline staff such as transition nurses or advanced practice nurses with designated transition roles who will carry out the innovation or be affected by it.	0	0	0	0	0
Factor 21 - Reflecting & Evaluating: quantitative and qualitative feedback about the progress and quality of implementation accompanied with regular personal and team debriefing about progress and experience.	0	0	0	0	0
Factor 22 - Measurement Capability/ Data Availability: availability of timely data. Capacity for monitoring, evaluation, and process improvement. Includes measurement differences; accountability for collection, documentation, and analysis.	0	0	0	0	0

	Strongly hindering	Hindering	Neither hindering nor facilitating	Facilitating	Strongly facilitating
Factor 23 - Engaging Key Stakeholders: individuals from within the organization that are directly impacted by the transitional care innovation, e.g. staff responsible for making referrals to a new program or using a new work process.	0	0	0	0	0
Factor 24 - Engaging Organizations, External Context: developing and capitalizing on relationships with healthcare professionals and frontline staff in the implementing organizations, and promoting external collaborations with outside care providers, and resources linked to the implementation of a transitional care innovation.	0	0	0	0	0
Factor 25 - Engaging Innovation Participants: individuals (patients/older persons, family, informal caregivers) served by the organization that participate in the transitional care innovation.	0	0	0	0	0

Q1) Please provide any further comments you might have on the direction of influence of the 25 factors in relation to implementing Transitional Care Innovations:

Section B - Importance of Influence:

2) Rate the **importance of influence** of each of the 25 factors listed below on the implementation of Transitional Care Innovations:

	Not important	Slightly important	Moderately important	Very important	Extremely important
Domain I - Intervention (Transitional Care Innovation) Characteristics Factor 1 - Targeted Groups: patients/older population who are the intended recipients or beneficiaries of the transitional care innovation.	0	0	0	0	0
Factor 2 - Complexity: perceived difficulty of implementation, reflected by duration, scope, radicalness, disruptiveness, centrality, and intricacy and number of steps required to implement.	0	0	0	0	0
Factor 3 - Relative Advantage: stakeholders' perception of the advantage (benefits and usefulness) of implementing the transitional care innovation versus an alternative solution.	0	0	0	0	0
Factor 4 - Evidence Strength & Quality: stakeholders' perceptions of the quality and validity of evidence (proven effectiveness) supporting the belief that the transitional care intervention will have desired outcomes (e.g. low readmission rates).	0	0	0	0	0
Domain II - Outer Setting Factor 5 - Cosmopolitanism: the degree to which an organization is networked with other external organizations.	0	0	0	0	0
Factor 6a - External Policy: a broad construct that includes external strategies (by government or other central entity) to spread transitional care innovations; including policy, regulations, laws, external mandates, legislative changes, recommendations, and guidelines.	0	0	0	0	0
Factor 6b - External Incentives: a broad construct that includes external strategies (by the government or other central entity) to spread transitional care innovations; including national funding schemes or governmental sponsorship.	0	0	0	0	0
Domain III - Inner (Organizational) Setting Factor 7 - Networks & Communications: the nature and quality of webs of social networks and of formal/ informal communications within an organization (e.g. interdisciplinary teams, coordination & communication among team members).	0	0	0	0	0
Factor 8 - Culture: norms, values, and basic assumptions of a given organization.	0	0	0	0	0

	Not important	Slightly important	Moderately important	Very important	Extremely important
Factor 9 - Relative Priority: individuals' (healthcare professionals, staff within implementing team) shared perception of the importance of the implementation of a transitional care innovation within the organization.	0	0	0	0	0
Factor 10 - Leadership Engagement: commitment, involvement, and accountability of leaders and managers with the implementation of a transitional care innovation.	0	0	0	0	0
Factor 11 - Available Resources: the level of resources dedicated for the implementation and on-going operations of a transitional care innovation; including staffing levels, money, funding, training, education, physical space, equipment, and time.	0	0	0	0	0
Factor 12 - Access to Knowledge & Information: ease of access to digestible information and knowledge (e.g. mentoring, initial training) about the transitional care innovation and how to incorporate it into work tasks.	0	0	0	0	0
Factor 13 - Continuity: information continuity (e.g. patient information exchange, services & care planning) and relationship continuity, both with providers and patients/caregivers and across organizations.	0	0	0	0	0
Factor 14 - IT&HIT resources (HIT systems): electronic information management infrastructure and technologies (e.g. electronic health records) available to clinicians to manage patient care, data, and communications.	0	0	0	0	0
Domain IV - Characteristics of Individuals Factor 15a - Knowledge & Beliefs about the Intervention: healthcare professionals/staff within implementing team's beliefs, expectations, and familiarity with facts, truths, & principles related to the transitional care innovation.	0	0	0	0	0
Factor 15b - Knowledge & Beliefs about the Intervention: patients/older persons' attitudes toward and value placed on the transitional care innovation as well as awareness on its care services & goals.	0	0	0	0	0
Factor 16 - Role: individual's role (healthcare professionals, staff within implementing team) and responsibility for the transitional care innovation; including the degree of multiple or shared roles.	0	0	0	0	0

	Not important	Slightly important	Moderately important	Very important	Extremely important
Factor 17 - Skills & Competencies: degree of relevant subject matter expertise, skills, and competencies within the implementing team, unit, and organization.	0	0	0	0	0
Factor 18a - Other Personal Attributes: healthcare professionals' other personal traits such as motivation levels, values, tolerance of ambiguity, critical attributes, intellectual ability, capacity, and learning style.	0	0	0	0	0
Factor 18b - Other Personal Attributes: patients/older persons' other personal traits such as health literacy, values, and acknowledgement of own care needs.	0	0	0	0	0
Domain V - (Implementation) Process Factor 19 - Planning: the degree to which a scheme or method of behavior and tasks for implementing an innovation are developed in advance, and the quality of those schemes or methods.	0	0	0	0	0
Factor 20 - Transition Roles (Frontline Staff): administrative staff, providers (within and outside the organization), e.g. frontline staff such as transition nurses or advanced practice nurses with designated transition roles who will carry out the innovation or be affected by it.	0	0	0	0	0
Factor 21 - Reflecting & Evaluating: quantitative and qualitative feedback about the progress and quality of implementation accompanied with regular personal and team debriefing about progress and experience.	0	0	0	0	0
Factor 22 - Measurement Capability/ Data Availability: availability of timely data. Capacity for monitoring, evaluation, and process improvement. Includes measurement differences; accountability for collection, documentation, and analysis.	0	0	0	0	0
Factor 23 - Engaging Key Stakeholders: individuals from within the organization that are directly impacted by the transitional care innovation, e.g. staff responsible for making referrals to a new program or using a new work process.	0	0	0	0	0
Factor 24 - Engaging Organizations, External Context: developing and capitalizing on relationships with healthcare professionals and frontline staff in the implementing organizations, and promoting external collaborations with outside care providers, and resources linked to the implementation of a transitional care innovation.	0	0	O	0	0

	Not important	Slightly important	Moderately important	Very important	Extremely important
Factor 25 - Engaging Innovation Participants: individuals (patients/older persons, family, informal caregivers) served by the organization that participate in the transitional care innovation.	0	0	0	0	0

Q2) Please provide any further comments you might have on the importance of influence of the 25 factors in relation to implementing Transitional Care Innovations:

Section C - Feasibility (easiness/difficulty):

3) How **easy/difficult** is it to address each of the 25 factors listed below in the development of implementation strategies for Transitional Care Innovations?

	Very difficult	Difficult	Neither difficult nor easy	Easy	Very easy
Domain I - Intervention (Transitional Care Innovation) Characteristics Factor 1 - Targeted Groups: patients/older population who are the intended recipients or beneficiaries of the transitional care innovation.	0	0	0	0	0
Factor 2 - Complexity: perceived difficulty of implementation, reflected by duration, scope, radicalness, disruptiveness, centrality, and intricacy and number of steps required to implement.	0	0	0	0	0
Factor 3 - Relative Advantage: stakeholders' perception of the advantage (benefits and usefulness) of implementing the transitional care innovation versus an alternative solution.	0	0	0	0	0
Factor 4 - Evidence Strength & Quality: stakeholders' perceptions of the quality and validity of evidence (proven effectiveness) supporting the belief that the transitional care intervention will have desired outcomes (e.g. low readmission rates).	0	0	0	0	0
Domain II - Outer Setting Factor 5 - Cosmopolitanism: the degree to which an organization is networked with other external organizations.	0	0	0	0	0
Factor 6a - External Policy: a broad construct that includes external strategies (by government or other central entity) to spread transitional care innovations; including policy, regulations, laws, external mandates, legislative changes, recommendations, and guidelines.	0	0	0	0	0
Factor 6b - External Incentives: a broad construct that includes external strategies (by the government or other central entity) to spread transitional care innovations; including national funding schemes or governmental sponsorship.	0	0	0	0	0
Domain III - Inner (Organizational) Setting Factor 7 - Networks & Communications: the nature and quality of webs of social networks and of formal/ informal communications within an organization (e.g. interdisciplinary teams, coordination & communication among team members).	0	0	0	0	0
Factor 8 - Culture: norms, values, and basic assumptions of a given organization.	0	0	0	0	0
Factor 9 - Relative Priority: individuals' (healthcare professionals, staff within implementing team) shared perception of the importance of the implementation of a transitional care innovation within the organization.	0	0	0	0	0
Factor 10 - Leadership Engagement: commitment, involvement, and accountability of leaders and managers with the implementation of a transitional care innovation.	0	0	0	0	0

	Very difficult	Difficult	Neither difficult nor easy	Easy	Very easy
Factor 11 - Available Resources: the level of resources dedicated for the implementation and on-going operations of a transitional care innovation; including staffing levels, money, funding, training, education, physical space, equipment, and time.	0	0	0	0	0
Factor 12 - Access to Knowledge & Information: ease of access to digestible information and knowledge (e.g. mentoring, initial training) about the transitional care innovation and how to incorporate it into work tasks.	0	0	0	0	0
Factor 13 - Continuity: information continuity (e.g. patient information exchange, services & care planning) and relationship continuity, both with providers and patients/caregivers and across organizations.	0	0	0	0	0
Factor 14 - IT&HIT resources (HIT systems): electronic information management infrastructure and technologies (e.g. electronic health records) available to clinicians to manage patient care, data, and communications.	0	0	0	0	0
Domain IV - Characteristics of Individuals Factor 15a - Knowledge & Beliefs about the Intervention: healthcare professionals/staff within implementing team's beliefs, expectations, and familiarity with facts, truths, & principles related to the transitional care innovation.	0	0	0	0	0
Factor 15b - Knowledge & Beliefs about the Intervention: patients/older persons' attitudes toward and value placed on the transitional care innovation as well as awareness on its care services & goals.	0	0	0	0	0
Factor 16 - Role: individual's role (healthcare professionals, staff within implementing team) and responsibility for the transitional care innovation; including the degree of multiple or shared roles.	0	0	0	0	0
Factor 17 - Skills & Competencies: degree of relevant subject matter expertise, skills, and competencies within the implementing team, unit, and organization.	0	0	0	0	0
Factor 18a - Other Personal Attributes: healthcare professionals' other personal traits such as motivation levels, values, tolerance of ambiguity, critical attributes, intellectual ability, capacity, and learning style.	0	0	0	0	0
Factor 18b - Other Personal Attributes: patients/ older persons' other personal traits such as health literacy, values, and acknowledgement of own care needs.	0	0	0	0	0
Domain V - (Implementation) Process Factor 19 - Planning: the degree to which a scheme or method of behavior and tasks for implementing an innovation are developed in advance, and the quality of those schemes or methods.	0	0	0	0	0

	Very difficult	Difficult	Neither difficult nor easy	Easy	Very easy
Factor 20 - Transition Roles (Frontline Staff): administrative staff, providers (within and outside the organization), e.g. frontline staff such as transition nurses or advanced practice nurses with designated transition roles who will carry out the innovation or be affected by it.	0	0	0	0	0
Factor 21 - Reflecting & Evaluating: quantitative and qualitative feedback about the progress and quality of implementation accompanied with regular personal and team debriefing about progress and experience.	0	0	0	0	0
Factor 22 - Measurement Capability/Data Availability: availability of timely data. Capacity for monitoring, evaluation, and process improvement. Includes measurement differences; accountability for collection, documentation, and analysis.	0	0	0	0	0
Factor 23 - Engaging Key Stakeholders: individuals from within the organization that are directly impacted by the transitional care innovation, e.g. staff responsible for making referrals to a new program or using a new work process.	0	0	0	0	0
Factor 24 - Engaging Organizations, External Context: developing and capitalizing on relationships with healthcare professionals and frontline staff in the implementing organizations, and promoting external collaborations with outside care providers, and resources linked to the implementation of a transitional care innovation.	0	0	0	0	0
Factor 25 - Engaging Innovation Participants: individuals (patients/older persons, family, informal caregivers) served by the organization that participate in the transitional care innovation.	0	0	0	0	0

Q3) Please provide any further comments you might have on the feasibility (easiness/ difficulty) to address each of the 25 factors in relation to implementing Transitional Care Innovations:

Q4) Please provide any additional factors (barriers or facilitators) that you would consider important/relevant to the implementation of Transitional Care Innovations.

Demographics:

1) Country (current place of residence/work):

- o Netherlands
- o Switzerland
- o United Kingdom
- o Sweden
- o United States of America
- o Canada
- o Australia
- o Singapore
- o Other

2) Education level:

- o Masters level
- o PhD level

3) Current role:

- □ Researcher
- □ Professor
- □ Associate Professor
- □ Assistant Professor
- □ Lecturer
- Practitioner, please specify
- Manager, please specify
- Director, please specify
- Other

4) Please indicate in which field you are an expert (choose all that applies):

- □ Transitional care
- Long-term care
- □ Healthcare innovations
- □ Implementation science

5) Please indicate the total years of experience in your field of expertise:

- o 3 to 5 years
- o 5 to 10 years
- o 10 years and above

Appendix 3B. Modified Delphi Study - Survey Round 2

Dear Expert,

Thank you for participating in the current Modified Delphi study survey and completing round 1. Hereby, we start round 2 of the study in order to reach unattained consensus from round 1 on the barriers & facilitators that influence the implementation of innovations in transitional care.

- Please refer to the results report round 1 that was sent to you by email in order to obtain further information on the results from the first survey.
- We would like to remind you again of key points before starting the survey:
 "Transitional care is defined as a set of actions designed to ensure the coordination and continuity of healthcare as patients transfer between different locations or different levels of care within the same location".

>We refer throughout this survey to Transitional Care (TC) Innovation, as <u>any intervention, model, or program</u>, which has been developed with a goal to improve or prevent care transitions for the <u>older population/persons (</u>> 65 years old) between different long-term care settings.

- The survey is developed based on the five domains: (Interventions characteristics, Outer setting, Inner (organizational) setting, Characteristics of Individuals, Process) of the CFIR (Consolidated Framework for Implementation Research) and selected constructs (factors) from this framework and the CTF (Care Transitions Framework).
- This survey consists of two sections and will explore the following concepts for each of the factors that didn't reach consensus in round 1, in addition to 7 new factors suggested by the respondents from round 1:

 >Section I - to rate the importance of influence of each factor on the implementation of transitional care innovations
 >Section II - to indicate the feasibility (easiness/difficulty) to address each factor in the development of implementation strategies for transitional care innovations

General Instructions: The survey will take approximately 15-20 minutes to complete. Your participation is voluntary, and you can withdraw from the study during the survey round without comment or penalty by closing the survey link. By clicking (yes, I consent) below, you will indicate that you have fully read and understood the complete information document provided to you earlier regarding this study with the participant invitation/recruitment email. Your responses to the survey are automatically saved as you go through the questions, and at any moment you can close the survey link and continue it later by using your same personal link. All data collected and processed will be kept anonymous, confidential, and stored on a password-protected database at Maastricht University. If you have any questions or technical issues please contact: Amal Fakha (main researcher, Department of Health Services Research, Faculty of Health Medicine and Life Sciences, Maastricht University, the Netherlands) by sending an email to<u>a.fakha@maastrichtuniversity.nl</u>. On the behalf of the research team, we thank you greatly for your participation.

- I hereby agree to participate and undertake this survey.
- o Yes, I consent.

Section I - Importance of Influence: determining the level of importance of the influence of each factor on the implementation of a transitional care innovation can be complicated due to the overall context and the interaction among various factors in the process; yet it remains crucial. Therefore, utilize your experience in the current healthcare field in order to respond to this section.

The factors listed below are those that <u>did not reach consensus in round 1 (except for leadership engagement</u> <u>which is redefined in this round)</u>, in addition to 7 new factors suggested by the respondents from round 1.

- Please, for each factor review the group choices & your individual choice from round 1 and re-rate the importance of influence of each on the implementation of Transitional Care Innovations. Also, you may decide to keep your original choice that you had from round.
- > Experts' comments for some factors from round 1 are provided where available & applicable, in order to clarify further their definition and interpretation. For the new factors, please rate accordingly.
- > Group choices are provided as % out of 100% = 29 (total number of respondents in round).
- > Note: for some factors their <u>definitions were revised</u> for this round (based on expert comments the definition is rephrased and enhanced to maximize clarity), in this case a label (<u>revised</u>) is added only next to those factors whose definitions were enhanced.

*Please, note that the group choices, individual choices percentages, and experts' comments were omitted from this appendix, for data protection purposes.

	Not important	Slightly important	Moderately important	Very important	Extremely important
Domain I - Intervention (Transitional Care Innovation) Characteristics Factor 1 - Targeted Groups (revised): patients/older population who are the intended recipients or beneficiaries of the transitional care innovation (e.g. matching the care needs of older persons with high frailty or dementia).	0	0	0	0	0
Factor 4 - Evidence Strength & Quality: stakeholders' perceptions of the quality and validity of evidence (proven effectiveness) supporting the belief that the transitional care intervention will have desired outcomes (e.g. low readmission rates).	0	0	0	0	0
Domain II - Outer Setting Factor 5 - Cosmopolitanism (revised): the degree to which an organization is networked with other external organizations (e.g. pre-existing partnerships; sharing of healthcare practices with external organizations).	0	0	0	0	0
Factor 6a - External Policy: a broad construct that includes external strategies (by government or other central entity) to spread transitional care innovations; including policy, regulations, laws, external mandates, legislative changes, recommendations, and guidelines.	0	0	0	0	0
Domain III - Inner (Organizational) Setting Factor 7 - Networks & Communications: the nature and quality of webs of social networks and of formal/informal communications within an organization (e.g. interdisciplinary teams, coordination & communication among team members).	0	0	0	0	0
Factor 10 - Leadership Engagement (revised): commitment, involvement, and accountability of leaders & managers with the implementation of a transitional care innovation. In addition, the presence of a skilled, motivated, and continuous leadership throughout the implementation (e.g. minimal turnover of dedicated project managers with high interest in the implementation).	0	0	0	0	0
Factor 12 - Access to Knowledge & Information: ease of access to digestible information and knowledge (e.g. mentoring, initial training) about the transitional care innovation and how to incorporate it into work tasks.	0	0	0	0	0
Factor 13 - Continuity (revised): care transitions' information continuity (e.g. exchange of patient medical information and the services & care plans between healthcare providers). In addition, the continuity of steady work relationships between the healthcare providers and patients/caregivers and across all the organizations involved in the transitional care innovation implementation.	0	0	0	0	0

	Not important	Slightly important	Moderately important	Very important	Extremely important
Factor 14 - IT&HIT resources (HIT systems): electronic information management infrastructure and technologies (e.g. electronic health records) available to clinicians to manage patient care, data, and communications.	0	0	0	0	0
Domain IV - Characteristics of Individuals Factor 15b - Knowledge & Beliefs about the Intervention: patients/older persons' attitudes toward and value placed on the transitional care innovation as well as awareness on its care services & goals.	0	0	0	0	0
Factor 18a - Other Personal Attributes: healthcare professionals' other personal traits such as motivation levels, values, tolerance of ambiguity, critical attributes, intellectual ability, capacity, and learning style.	0	0	0	0	0
Domain V - (Implementation) Process Factor 21 - Reflecting & Evaluating: quantitative and qualitative feedback about the progress and quality of implementation accompanied with regular personal and team debriefing about progress and experience.	0	0	0	0	0
Factor 22 - Measurement Capability/ Data Availability: availability of timely data. Capacity for monitoring, evaluation, and process improvement. Includes measurement differences; accountability for collection, documentation, and analysis.	0	0	0	0	0
Factor 24 - Engaging Organizations, External Context (revised): developing and capitalizing on relationships with healthcare professionals and frontline staff in the various organizations involved in the implementation of a transition care innovation, and promoting external collaborations with other outside care providers (e.g. home care agency), and resources (e.g. community resources or social services for older persons) linked to the implementation.	0	0	0	0	0
Factor 25 - Engaging Innovation Participants (revised): individuals (patients/older persons, family, informal caregivers) served by the organization that participate in the transitional care innovation (e.g. ensuring family inclusion in care goals setting).	0	0	0	0	0

	Not important	Slightly important	Moderately important	Very important	Extremely important
*Additional New Factors (suggested by respondents from round 1): New Factor 1 - Power of Decision-makers: defined as the main trigger to allow the successful implementation of a transitional care innovation.	0	0	0	0	0
New Factor 2 - Sense of Urgency: the urgent need and attention given to implementing a specific transitional care innovation with respect to other innovation projects being addressed within an organization.	0	0	0	0	0
New Factor 3 - Adoption of Change in Work Processes: adapting and changing "how things work" across several levels (organizational, group, individual, policy); and in view of the capacity for implementing this change within an organization.	0	0	0	0	0
New Factor 4 - Financing of Transitional Care Innovation Implementations: the existing financing structures that affect the implementation such as fragmented financing & a lack of clear financing structures, or varied reimbursement systems of healthcare services.	0	0	0	0	0
New Factor 5 - Inter-organizational Collaborations: the presence of long- lasting trust relationships between multiple organizations involved in transitional care; in addition to the level of good collaboration and care coordination among different organizations within different and various sectors of the healthcare system, as well as varied care disciplines/services.	0	0	0	0	0
New Factor 6 - Previous Experiences with Implementation of Innovations: a prior history and experience of organizations & healthcare professionals in implementing change and new interventions, programs, and innovations.	0	0	0	0	0
New Factor 7 - Co-design of the Transitional Care Innovation: the degree of involvement of the stakeholders (older people and healthcare providers) in the design of the innovation prior to the implementation stages.	0	0	0	0	0

Q1) Please provide any further comments you might have on the importance of influence of the listed factors in relation to implementing Transitional Care Innovations:
Section II - Feasibility (easiness/difficulty): the ability to address factors in the development of implementation strategies can be intricate and complex due to the overall context, which can be varied for each innovation implementation. Therefore, utilize your experience in the current healthcare field in order to respond to this section.

The factors listed below are those that <u>did not reach consensus in round 1</u>, in addition to 7 new factors suggested by the respondents from round 1.

>Please, for each factor review the group choices & your individual choice from round 1 and re-rate the **easiness/difficulty** to address each in the development of implementation strategies for Transitional Care Innovations. Also, you may decide to keep the original choice that you had from round 1. >Experts' comments for some factors from round 1 are provided where available & applicable, in order to clarify further their definition and interpretation. For the new factors, please rate accordingly.

>Group choices are provided as % out of 100% = 29 (total number of respondents in round 1).

>Note: for some factors their <u>definitions were revised</u> for this round(based on experts comments, the definition is rephrased and enhanced to maximize clarity), in this case a label (<u>revised</u>) is added only next to those factors whose definitions were enhanced.

*Please, note that the group choices, individual choices percentages, and experts' comments were omitted from this appendix, for data protection purposes.

	Very difficult	Difficult	Neither difficult nor easy	Easy	Very easy
Domain I - Intervention (Transitional Care Innovation) Characteristics Factor 1 - Targeted Groups (revised): patients/ older population who are the intended recipients or beneficiaries of the transitional care innovation (e.g. matching the care needs of older persons with high frailty or dementia).	0	0	0	0	0
Factor 3 - Relative Advantage: stakeholders' perception of the advantage (benefits and usefulness) of implementing the transitional care innovation versus an alternative solution.	0	0	0	0	0
Factor 4 - Evidence Strength & Quality: stakeholders' perceptions of the quality and validity of evidence (proven effectiveness) supporting the belief that the transitional care intervention will have desired outcomes (e.g. low readmission rates).	0	0	0	0	0
Domain II - Outer Setting Factor 5 - Cosmopolitanism (revised): the degree to which an organization is networked with other external organizations (e.g. pre-existing partnerships; sharing of healthcare practices with external organizations).	0	0	0	0	0
Domain III - Inner (Organizational) Setting Factor 9 - Relative Priority (revised): individuals' (healthcare professionals, staff within implementing team) shared perception of the importance of the implementation of a transitional care innovation within the organization (e.g. existence of multiple quality improvement projects within the organization at the same time).	0	0	0	0	0
Factor 10 - Leadership Engagement (revised): commitment, involvement, and accountability of leaders & managers with the implementation of a transitional care innovation. In addition, the presence of a skilled, motivated, and continuous leadership throughout the implementation (e.g. minimal turnover of dedicated project managers with high interest in the implementation).	0	0	0	0	0
Factor 12 - Access to Knowledge & Information: ease of access to digestible information and knowledge (e.g. mentoring, initial training) about the transitional care innovation and how to incorporate it into work tasks.	0	0	0	0	0
Factor 13 - Continuity (revised): care transitions' information continuity (e.g. exchange of patient medical information and the services & care plans between healthcare providers). In addition, the continuity of steady work relationships between the healthcare providers and patients/caregivers and across all the organizations involved in the transitional care innovation implementation.	0	0	0	0	0
Domain IV - Characteristics of Individuals Factor 15a - Knowledge & Beliefs about the Intervention: healthcare professionals/staff within implementing team's beliefs, expectations, and familiarity with facts, truths, & principles related to the transitional care innovation.	0	0	0	0	0

Continued.

	Very difficult	Difficult	Neither difficult nor easy	Easy	Very easy
Factor 15b - Knowledge & Beliefs about the Intervention: patients/older persons' attitudes toward and value placed on the transitional care innovation as well as awareness on its care services & goals.	0	0	0	0	0
Factor 16 - Role: individual's role (healthcare professionals, staff within implementing team) and responsibility for the transitional care innovation; including the degree of multiple or shared roles.	0	0	0	0	0
Factor 17 - Skills & Competencies: degree of relevant subject matter expertise, skills, and competencies within the implementing team, unit, and organization.	0	0	0	0	0
Factor 18b - Other Personal Attributes: patients/ older persons' other personal traits such as health literacy, values, and acknowledgement of own care needs.	0	0	0	0	0
Domain V - (Implementation) Process Factor 20 - Transition Roles (Frontline Staff): administrative staff, providers (within and outside the organization), e.g. frontline staff such as transition nurses or advanced practice nurses with designated transition roles who will carry out the innovation or be affected by it.	0	0	0	0	0
Factor 21 - Reflecting & Evaluating: quantitative and qualitative feedback about the progress and quality of implementation accompanied with regular personal and team debriefing about progress and experience.	0	0	0	0	0
Factor 22 - Measurement Capability/Data Availability: availability of timely data. Capacity for monitoring, evaluation, and process improvement. Includes measurement differences; accountability for collection, documentation, and analysis.	0	0	0	0	0
Factor 23 - Engaging Key Stakeholders: individuals from within the organization that are directly impacted by the transitional care innovation, e.g. staff responsible for making referrals to a new program or using a new work process.	0	0	0	0	0
Factor 24 - Engaging Organizations, External Context (revised): developing and capitalizing on relationships with healthcare professionals and frontline staff in the various organizations involved in the implementation of a transition care innovation, and promoting external collaborations with other outside care providers (e.g. home care agency), and resources (e.g. community resources or social services for older persons) linked to the implementation.	0	0	0	0	0
Factor 25 - Engaging Innovation Participants (revised): individuals (patients/older persons, family, informal caregivers) served by the organization that participate in the transitional care innovation (e.g. ensuring family inclusion in care goals setting).	0	0	0	0	0

Continued.

	Very difficult	Difficult	Neither difficult nor easy	Easy	Very easy
*Additional New Factors (suggested by respondents from round 1): New Factor 1 - Power of Decision-makers: defined as the main trigger to allow the successful implementation of a transitional care innovation.	0	0	0	0	0
New Factor 2 - Sense of Urgency: the urgent need and attention given to implementing a specific transitional care innovation with respect to other innovation projects being addressed within an organization.	0	0	0	0	0
New Factor 3 - Adoption of Change in Work Processes: adapting and changing "how things work" across several levels (organizational, group, individual, policy); and in view of the capacity for implementing this change within an organization.	0	0	0	0	0
New Factor 4 - Financing of Transitional Care Innovation Implementations: the existing financing structures that affect the implementation such as fragmented financing & a lack of clear financing structures, or varied reimbursement systems of healthcare services.	0	0	0	0	0
New Factor 5 - Inter-organizational Collaborations: the presence of long-lasting trust relationships between multiple organizations involved in transitional care; in addition to the level of good collaboration and care coordination among different organizations within different and various sectors of the healthcare system, as well as varied care disciplines/services.	0	0	0	0	0
New Factor 6 - Previous Experiences with Implementation of Innovations: a prior history and experience of organizations & healthcare professionals in implementing change and new interventions, programs, and innovations.	0	0	0	0	0
New Factor 7 - Co-design of the Transitional Care Innovation: the degree of involvement of the stakeholders (older people and healthcare providers) in the design of the innovation prior to the implementation stages.	0	0	0	0	0

Q2) Please provide any further comments you might have on the feasibility (easiness/ difficulty) to address each of the listed factors in relation to implementing Transitional Care Innovations:

.....

Appendix 3C.	List of	factors	that	reached	consensus	on	direction	of	influence	on	the
implementatior	of TCIs	s, from r	ound	one							

Factor	Rating: Hindering/Strongly Hindering (Consensus level in %)	CFIR Domain
Round one		
Complexity	96	Intervention characteristics
	Rating: Facilitating/Strongly Facilitating (Consensus level in %)	
Leadership engagement	100	Inner setting
Engaging key stakeholders	97	Process
Relative advantage	93	Intervention characteristics
External incentives	93	Outer setting
Evidence strength and quality	90	Intervention characteristics
Relative priority	90	Inner setting
Available resources	86	Inner setting
Access to knowledge and information	86	Inner setting
Transition roles - frontline staff	86	Process
Skills and competencies	83	Characteristics of individuals
External policy	83	Outer setting
Networks and communications	83	Inner setting
Planning	79	Process
Reflecting and evaluating	79	Process
Knowledge and beliefs of healthcare professionals about the TCIs	76	Characteristics of individuals
Measurement capability/data availability	72	Process
Engaging organizations, external context	72	Process



CHAPTER

Implementing Four Transitional Care Interventions for Older Adults: A Retrospective Collective Case Study

Published as:

Fakha A, Leithaus M, de Boer B, van Achterberg T, Hamers JP, Verbeek H. Implementing four transitional care interventions for older adults: a retrospective collective case study. Gerontologist. 2022. https://doi.org/10.1093/geront/gnac128

ABSTRACT

BACKGROUND AND OBJECTIVES

Four interventions to improve care transitions between hospital and home or community settings for older adults were implemented in Leuven, Belgium over the past four years. These complex interventions consist of multiple components that challenge their implementation in practice. This study examines the influencing factors, strategies used to address challenges in implementing these interventions, and implementation outcomes from perspectives of healthcare professionals involved.

RESEARCH DESIGN AND METHODS

This was a qualitative, collective case study which was part of the TRANS-SENIOR research network. Authors conducted semi-structured interviews with healthcare professionals about their perceptions regarding the implementation. Thematic analysis was used, and the Consolidated Framework for Implementation Research guided the final data interpretation.

RESULTS

Thirteen participants were interviewed. Participants reported major implementation bottlenecks at the organizational level (resources, structure, information continuity), while facilitators were at the individual level (personal attributes, champions). They identified engagement as the primary strategy used, and suggested other important strategies for future sustainability of the interventions (building strategic partnerships, lobbying for policies to support transitional care). They perceived the overall implementation favorably, with high uptake as a key outcome.

DISCUSSION AND IMPLICATIONS

This study highlights the strong role of healthcare providers, being motivated and selfdriven, to foster the implementation of interventions in transitional care in a bottom-up way. It is important to use implementation strategies targeting both the individual-level factors as well as the organizational barriers for transitional care interventions in the future.

KEYWORDS

Integrated care, innovations, barriers, facilitators, strategies.

BACKGROUND AND OBJECTIVES

Across Europe, the population of older adults (65 years and above) with chronic disease and multi-morbidity has risen dramatically in recent years (1). Moreover, older adults have increased health care utilization (2) and are at higher risk of care transitions between multiple care settings (3). Unfortunately, care transitions are vulnerable phases to older adults, who are often confronted with care fragmentation and a lack of coordination among healthcare providers (4). This leads to compromised patient outcomes, such as medication errors or more hospital readmissions (5). To address this challenge, the concept of integrated care was encouraged to enhance transition and coordination across or within the different levels and sites of care sectors (6).

Integrated care approaches (i.e., across different care settings, such as hospitals and primary or community care) are promising solutions to improve the quality and efficiency of care transitions for older adults (7, 8). In this study, we focus on interventions with an integrated care approach that aim to improve the care transitions for older adults with chronic diseases between hospital and home or community settings, which we refer to as transitional care interventions (TCIs). Recent global forces in healthcare delivery to enhance transitional care for older adults have driven the development and implementation of a plethora of innovative TCIs embedding integrated care (9-11). However, although the effectiveness of these interventions is promising (11), there is still an inadequate awareness and understanding of how to successfully implement them in practice (12). Furthermore, studies that comprehensively investigate the implementation (context, strategies, and outcomes) of these interventions are limited (13, 14). To date, literature has highlighted that exploring implementation factors in the context is pivotal for implementing complex interventions in healthcare (15). Research has identified multiple factors (barriers, facilitators) influencing the implementation of integrated care and TCIs for older adults that often behave as two sides of the same coin (e.g., insufficient resources as a barrier/sufficient resources as a facilitator) depending on the context (16, 17). For example, low organizational readiness for change, regulatory challenges, failure to target the right population, and restricted knowledge on the intervention by implementers were key barriers to implement such interventions (16, 18), while appointing champions to promote the interventions or assigning transition roles for staff were strong facilitators (16, 19).

Because of the complexity of implementing interventions such as TCIs, various implementation strategies described as methods used to improve adoption, implementation, and sustainment of interventions in healthcare practice were developed (20). A few examples of such strategies include assessing for implementation readiness and identifying barriers/facilitators, involving executive boards, obtaining formal commitments, involving patients, expanding roles/shifting tasks, or using an implementation advisor (21, 22). These

strategies, especially when tailored to the context, can have potentially positive effects on the implementation. Yet, their use is either rare or not correct when observed in practice, where "we learn as we do" is more likely the trend (23). Hence, there is an ambiguity on how to best embed these interventions in the actual world of transitional care practice, whereby even implementation strategies that can work in one setting might not in another.

Moreover, there is limited knowledge on the particular implementation of TCIs focusing on older adults with chronic diseases moving between hospital and home or community settings. A thorough study of all of the key aspects of implementation is still lagging behind in this field of care. Hence, there is merit to closely investigate this implementation in the real-life context and to obtain an in-depth understanding on what are the practical issues or guarantors of success.

THE CASE: FOUR TRANSITIONAL CARE INTERVENTIONS

In 2018, a government-led pilot project was launched in Leuven, within the Flemish region of Belgium, which aimed to improve integrated care for people with chronic diseases (24). The main objectives were to improve the outcomes of population health, improve patient and provider experiences, and achieve better cost efficiency (25). Within this project, four interventions focused on transitional care. This collective retrospective case study investigates the implementation of these four TCIs: 1. intermediate care center, 2. envelope action/medication reconciliation, 3. caring neighborhood teams, and 4. chronic heart failure care program for enhancing care transitions of older adults with chronic disease between hospital and home or community settings (26). The four TCIs were created in reference to guidance provided by the government on integrated care; however the specific components of each of the interventions were developed from an assessment of the local care needs of the population in Leuven. The needs assessment was a result of discussions and consensus among a multidisciplinary team in the region including general practitioners (GPs), homecare organizations, hospitals, social/community services, and a community pharmacists' network who agreed on the local care needs and designed the interventions accordingly (27).

This study aimed to qualitatively examine the four cases from an implementation science perspective, informed by the viewpoints of project coordinators and healthcare professionals involved in the implementation of the TCIs. The main study objective was to examine three key implementation aspects: a) to explore which factors influenced the implementation of the TCIs, b) to identify if any implementation strategies were used to implement the TCIs, and c) to report on the implementation outcomes of the TCIs and the overall success.

RESEARCH DESIGN AND METHODS

STUDY DESIGN AND CASE SELECTION

This study used a qualitative collective case study research design (28, 29). We selected a case study design with an interpretative and constructivist approach in order to obtain a naturalistic and an in-depth understanding of a complex and context-dependent issue (implementation of transitional care interventions) as perceived by healthcare professionals (29, 30). Cases were defined as the four TCIs. For each case, data were collected using interviews to explore various implementation aspects from the perspectives of project coordinators and healthcare professionals.

INTERVENTION SELECTION AND DESCRIPTION

An initial meeting with the project coordinators of the overarching integrated care pilot project led to the identification and selection of the interventions, which were focused on transitional care for older adults between hospital and home or community settings, implemented in Leuven. Table 1 describes each intervention based on information retrieved from the official project website and documents provided by the project coordinators (26).

PARTICIPANTS

First, we identified the key contacts within the overarching integrated care project, who helped us determine the TCIs cases and directed us to the core project coordinators of these specific interventions. The project coordinators were particularly knowledgeable and played a critical role in developing and implementing the four TCIs (31, 32). These project coordinators were interviewed then asked to suggest additional potential candidates using the snowball sampling. We ensured the inclusion of participants with either an in-depth knowledge of the TCIs' implementation and/or those who were involved in delivering the interventions directly (31). We invited 24 candidates for interviews by email and sent a study information document and consent forms. Saturation was determined as reached when new interviews became redundant and provided little new information (33).

Description	Case A: Intermediate Care Center	Case B: Envelope Action / Medication Reconciliation	Case C: Caring Neighborhood Teams	Case D: Chronic Heart Failure Care Program
Overview	An interim center referred to as "bed house" is developed inside a hospital - initiated during the outbreak of the COVID-19 pandemic. It is a care transition unit between hospital and home settings and vice versa. Implementation started in April 2020 and lasted for approximately 3 months (first wave of COVID-19).	An envelope that contains the patient's medication scheme and prescriptions is provided to patients upon discharge from hospital and addressed to the community pharmacist. Implementation started in 2018:	A network of multidisciplinary primary care providers and social workers within one neighborhood, that work jointly on improving care at the population level. The teams focused on mapping the individuals with multimorbidity within the neighborhood and providing care coordination tailored to their needs. Implementation started in 2018'.	A care program developed specifically for heart failure disease management and care coordination. Implementation started in 2018°.
Function	To prevent hospital bed-blocking by accommodating patients that are either medically stable to be discharged from hospital but not yet able to return to their home due to social and medical reasons, or have care needs which were too complex to be able to stay at home but did not require hospitalization.	To facilitate medical information transfer between hospital care and transfer between hospital care and to perform medication econcilitation (checking for medication discrepancies and medication-related issues) and patient counseling.	To ensure integrated and population- oriented care at the neighborhood level.	To ensure a close follow-up for the chronic heart failure patients discharged from the hospital.
Transitional care aim	Improve care transitions from hospital to home and vice versa.	Improve care transitions from hospital to home or community settings and prevent re-hospitalization.	Prevent care transitions from home or community settings to hospital and reduce readmissions.	Improve care transitions from hospital to home and avoid rehospitalization.
Settings and care organizations involved	Hospital, intermediate care unit in the hospital, private home, homecare, and primary care.	Hospital, community pharmacists' providers' network, homecare, and primary care.	Community service centers, social care, homecare, and primary care.	Hospital, community, private home, homecare, and primary care.

Table 1. Description of the Four Transitional Care Interventions (TCIs)

Description	Case A: Intermediate Care Center	Case A: Intermediate Care Center Case B: Envelope Action / Medication Reconciliation	Case C: Caring Neighborhood Teams	Case D: Chronic Heart Failure Care Program
Other features	\supset			Home care nurse – role as heart
	a digital communication tool			failure patient educator
	developed and used among			Structured transitional protocol to
	care providers within the center			guide post-discharge care
	and outside, which enabled			Development of a discharge
	continuous communication flow to			checklist in the hospital's EHR
	arrange care.			specific for heart failure post-
				discharge follow-up care
				E-learning course for GPs on heart
				failure management
				Automated diagnostic and qualitative
				audits in GPs' EHR to improve
				chronic heart failure case finding

Note: COVID-19 = coronavirus disease 2019; EHR = electronic health record; GP = general practitioner.

*The development and implementation of the TCIs were initiated in 2018 and continued until at least 2021 when this study was conducted. All four interventions were coordinated by a

core team of project coordinators from within the different organizations involved across the interventions in the region and not one primary organization per intervention.

DATA COLLECTION

Interviews

We conducted individual semi-structured interviews using an interview guide (in Dutch language) with questions and prompts specific to either project coordinators or healthcare professionals (31). We developed the questions with the aid of published frameworks and concepts on implementation factors, strategies, and outcomes. Hence, we used the Consolidated Framework for Implementation Research (CFIR) interview tool in order to obtain perspectives on the implementation of the TCIs; for the complete interview guide, see Appendix 4A (21, 34-36). The guide was tested prior to use among the research team, and two masters-level student researchers performed the interviews between February and April 2021, with either of the authors (AF or ML) also present as observers. The interviews (lasting an average of 55 min) were conducted online using a data-protected video conferencing tool then recorded and transcribed verbatim. Transcripts were translated into English by the students who are native Dutch language speakers and checked by author ML as the Dutch-speaking researcher. Then, all transcripts were entered into NVivo (QSR International software, 2020) for coding and analysis.

DATA ANALYSIS

We conducted a combined thematic analysis, starting with an inductive and then a deductive approach, and following a six-step methodology (37). This data-driven analysis with an interpretative and constructivist approach served the objective of building knowledge about and understanding the implementation of the TCIs from the perspectives of individuals involved in the process. Authors (AF, ML) analyzed the data supported by NVivo; see Table 2 for steps of the inductive analysis and Appendix 4B for illustrations of the coding. All transcripts were combined together, and the pooled data was used as one main unit of analysis to allow a collective data analysis and not a comparative one among the cases (28). The second stage of the analysis followed a deductive approach and involved mapping data within the themes only pertaining to the influencing factors to the CFIR's domains/ constructs, using the CFIR's codebook; see Appendix 4C for description of the CFIR constructs (38). This provided a further classification and interpretation of the findings on the implementation factors.

ETHICAL CONSIDERATIONS

The study was approved by the Ethics Committee Research UZ/KU Leuven (approval number MP017284), and an informed consent form was obtained from each participant.

St	eps	Description
	Familiarizing with the data	Authors (AF and ML) read through the full transcripts in order to familiarize themselves with the data and obtain an overall preliminary understanding of the content, alongside taking important notes.
	Generating initial codes	The lead researcher (AF) started the inductive coding of all transcripts by first generating initial codes from the data, then collating the relevant extract data under each code. Simultaneously, ML independently co-coded all same transcripts. After rounds of coding, AF and ML reviewed and compared the codes along with the coded data extracts, and minor disagreements were discussed and resolved.
	Searching for themes	Following four rounds of coding and adjustments, AF developed an initial set of potential summary themes.
	Reviewing themes	AF and ML jointly reviewed the themes in relation to both the codes and the entire data set in an iterative way until both agreed on the final themes and their meaningfulness. Then, they developed a thematic map to provide an overview of the analysis.
5.	Defining and naming themes	The research team developed, discussed, and agreed on a clear description, detailed summary analysis, and naming of each theme.
	Producing the report	The research team produced a final report summarizing the key analysis results with selected quotes from the data, which they aligned it with the existing literature on implementation science concepts.

Table 2. Six steps of inductive thematic analysis

RESULTS

PARTICIPANTS' CHARACTERISTICS

Thirteen participants (five project coordinators and eight healthcare professionals) were interviewed as the following: four (Case A), two (Case B), five (Case C), and three (Case D). One participant was involved as a project coordinator in both Cases A and B, and hence was interviewed twice. The participants were almost equally distributed between men and women; and their professions included pharmacists, GPs, nurses, physiotherapists, and cardiologists. All participants were located and working across the various care organizations relevant to each TCl implementation (Case). Table 3 provides a breakdown of the participants' professional backgrounds and demographics.

OVERVIEW

Our thematic analysis yielded eight themes reflecting the three implementation aspects studied (implementation factors, strategies, and outcomes) for the four interventions combined. The relevant themes for each aspect are described below, see Figure 1 for listing of themes. In addition, a total of 28 codes were identified for the entire sample and across all themes; see Appendix 4B (illustrations of the coding) for the individual codes and count per each theme.

Table 3. Description of the participants			
Cases	Participants ^a Profession (N, gender)	Organization type	Time of involvement in the TCI
Case A Intermediate Care Center	Project coordinators: Pharmacists (3W) HCP: Homecare nurse (1M)	Community pharmacists' providers' network Homecare organization	Approximately 3 months
Case B Envelope Action / Medication Reconciliation	Project coordinators: Pharmacist (1W) HCP: Pharmacist (1W)	Community pharmacists' providers' network	Average of 3 years ^b
Case C Caring Neighborhood Teams	Project coordinators: Physiotherapist (1W) HCP: Physiotherapist (1M), GP (2M), Policy advisor welfare & care (1W)	Hospital Primary care group practice, GP group practice, Government administration for Leuven city	Average of 3 years ^b
Case D Chronic Heart Failure Care Program	Project coordinators: GP (1M) HCP: Cardiologist (1M), Nurse (1M)	GP group practice Hospital	Average of 3 years ^b
Noto: W = woman: M = man: HCB = healthcare p	Noto: W – woman: M – man: LICB – hoalthoaro professional: CB – donoral practitionar: TCI– transitional care intorviou	ional caro intervention	

Note: W = woman; M = man; HCP = healthcare professional; GP = general practitioner; TCI= transitional care intervention

^a All participants are White, of European origin (race).

^b Time of involvement in each intervention could be variable but is presumable to be an average of 3 years, between year 2018 when the development and implementation of the interventions started and until at least 2021 when this study was conducted.





ASPECT I – IMPLEMENTATION FACTORS

Four themes describing the factors influencing the implementation of the TCIs emerged. By mapping the data within these themes to the CFIR domains/constructs, we obtained a clearer vision on the influencing factors, which were found across all the CFIR's domains. Key barriers were linked to the inner setting (organizational level) while main facilitators belonged to the characteristics of the individuals and the process of engaging. Table 4 presents the corresponding influencing factors as per the CFIR for each theme, along with the supporting quotes.

Theme One: Significant Barriers at the Organizational Level

According to participants, the implementation of the interventions was mainly hindered by a lack of organizational resources. They reported that the shortage of staff (e.g., nurses), heavy workloads, and insufficient time for care providers to perform their usual work duties plus new tasks exerted an extra pressure to implement the interventions (Table 4, quotes 3, 4). Moreover, participants indicated that low available funds for the implementation led them to operate with existing organizational budgets and resources (Table 4, quotes 5, 6). They reported that the budget provided by the government to care organizations and project coordinators was below the requirements to support the implementation of the interventions. This led to a lack of funds to hire more staff or pay overtime hours for existing staff in order to support the implementation. Correspondingly, one participant implied that not every organization could fulfill the structural demands associated with implementing a specific intervention. This was seen in Case D, in which large organizations had better capacity to implement versus smaller ones (Table 4, quote 1). Furthermore, the absence of CHAPTER 4

an integrated health information technology (HIT) platform within and between different care organizations compromised the communication among teams and the exchange of patients' medical information during care transitions (Table 4, quotes 2, 7, 8). Participants also identified another barrier to implementation in the outer setting of the implementing organizations. The presence of multiple and misaligned governmental healthcare policies (federal or regional) and a fragmented financing structure for integrated care services impeded a smooth implementation of the interventions (Table 4, quotes 9, 10).

Theme Two: Power of Committed Individuals – "The Key Triggers & Facilitators"

Across the cases, the presence of a triad of highly motivated, committed, and self-driven care providers who initiated the implementation with a bottom-up approach was perceived as a big facilitator (Table 4, quote 11). Participants emphasized that the implementers' strong willpower combined with their great enthusiasm to improve care for older adults with complex care needs was necessary to the implementation of the interventions (Table 4, quotes 14, 15). In addition, the existing work relationships among GPs, nurses, social workers, and other providers, along with supportive community resources, enabled a smooth implementation (Table 4, quotes 12, 13). Correspondingly, participants reported that engaging the right individuals (key stakeholders, champions, and innovation participants) was critical to the implementation (Table 4, quotes 16-19). In addition, participants highlighted that the champions of the TCIs played a key role in achieving buy-in for implementation through being present at implementation sites and leveraging their internal connections. These champions created awareness about the interventions, motivated care providers, and convinced them to adopt it. Similarly, involving the key stakeholders and players, such as pharmacists' representatives in Case B or large GP practices in Case C, facilitated the implementation, according to participants' viewpoints (Table 4, quote 20).

Theme Three: Imperfect Fit between Interventions' Components & Older Adults' Profile

The interventions' design and elements at times mismatched the care needs and characteristics of the target population of older adults, which in turn impeded the implementation (Table 4, quote 21). From the participants' point of view, the patient identification criteria of some interventions (Cases C and D) or the ability to accommodate their complex medical and psychosocial needs was difficult (Table 4, quote 22, 23). Nevertheless, it was indicated that sometimes the older adults' insufficient knowledge on how an intervention works (Case B), their low information technology (IT) competences, or insufficient awareness of the interventions' components posed challenges to the implementation (Table 4, quote 24, 25).

Theme Four: Disruption of Implementation by COVID-19

Participants reported how the COVID-19 pandemic disrupted the implementation of the interventions in various ways. Mainly, there were difficulties in communication among care

providers which was problematic to managing and implementing the interventions (Table 4, quote 26, 27). Furthermore, during COVID-19 the numbers of older persons enrolled in the interventions were much lower such as in Case D, so that home education of the chronic heart failure program patients was discontinued (Table 4, quote 28). As seen in Case C, by contrast, the implementation of this intervention was accelerated during the pandemic in light of an urgent need for creating a sense of community, delivering medications to homes, providing support, and making formal agreements with hospitals on discharge policies (Table 4, quote 29).

Table 4.	Quotes	illustrating	factors	influencing	the	implementation	of	the	four	transitional	care
interventi	ons (TCIs	5).									

Theme	CFIR (domain/factor)	Quotes
Significant barriers at	Inner setting:	
the organizational level	Structural characteristics	 "So, in practice, you only see it being rolled out in the centers that are indeed capable of it and those are really the larger centers who are big enough, the cardiologists who are willing to do that, and who can convince the hospital that it is important." (HCP, Case D)
	Networks and communications	 "Yes, in terms of stumbling blocks. I think it is really just communication with GPs that is the biggest issue, but they are looking for a digital solution." (HCP, Case D)
	Readiness for implementation -	 "The obstacle in the implementation is that I personally have to be able to do it within my time." (HCP, Case D)
	available resources	4) "Things that discourage are the administrative burden, things that have to be done extra, lack of time, all the care providers simply have an impossible amount of work and they are always short of time." (Project Coordinator, Case C)
		5) "That is the biggest challenge and in terms of barrier, that means lack of funding for staff, for interventions, well for everything." (Project Coordinator, Case D)
		6) "when you say about the difficulties of financing, yes the funny thing was that there was actually rarely a budget for the nurse coordinator in the intermediate care center." (HCP, Case A)
	HIT systems resources	7) "So, communication by electronic means is the most difficultbut the big problem remains finding a platform where first, second, and third can communicate." (HCP, Case C)
	Information continuity	8) "What we have noticed and continue to notice that data sharing just doesn't work, because you have different platforms." (HCP, Case A)
	Outer setting:	
	External policy and incentives	9) "I think what is blocking us most of all is the fact that the supra-local policy is not so well coordinated; the federal health policy and the Flemish welfare policy." (HCP, Case C)
		10) "The way primary care is currently financed is not always very conducive to integrated care, because it actually keeps the partitions in place, and partitioning is difficult if the financing model doesn't change". (Project Coordinator, Case A)

Table 4. Continued

Theme	CFIR (domain/factor)	Quotes
Power of committed	Intervention characteris	tics:
individuals – "the key triggers and facilitators"	Intervention source	11) "A very important one has been the neighborhood teams; these are teams of care providers in the first line. GPs, pharmacists, nurses, physiotherapists, psychologists, etc. who work together at neighborhood level on chronic patients and they implement the care programs that is actually how it came about from the bottom up, because the initiators of those neighborhood teams were the GPs." (Project Coordinator, Case A)
	Outer setting:	
	Cosmopolitanism	12)"We are very lucky to have a chronic care project in this region, because, yeah, healthcare providers they kind of know each other and that is like very beneficial or very facilitating for this cooperation." (Project Coordinator, Case A)
	Community resources	13) "I think the fact that we have community centers and local service centers in Leuven is a good thing, yes, anchor points for a caring neighborhood." (HCP, Case C)
	Characteristics of individ	duals (healthcare providers, implementers):
	Knowledge and beliefs about the intervention	14) "In addition, what certainly plays a role is that, as I said at the beginning, there is a motivator for more quality care, so a facilitator or motivator, achieving support and shared responsibility are all things that motivate". (Project Coordinator, Case C)
	Other personal attributes	15) "There is a lot of commitment, there is a lot of enthusiasm, there is a lot of goodwill and openness to try things not only to that heart failure project, but also to the broader project. So that is positive, but actually implementing it then yes, that also depends on how much personal affinity people or carers have with it. That varies greatly, but the basis is that there is a lot of goodwill it was our own motivation and commitment." (Project Coordinator, Case D)
	Process – Engaging:	
	Champions	16) "What was good about facilitating us was the people [TCIs team] themselves. They really took the initiative to set up meetings with the community team because otherwise, I think, it would never have happened so quickly." (HCP, Case C)
		17) "And [person name from TCIs team] was kind of more a backup that was very good because she is like the face of [TCIs], so this really helps for this kind of work; and I think this really works for the motivation of the pharmacists the healthcare providers, in general. " (Project Coordinator Case A)
	Key stakeholders	18) "but [person name from TCIs team] yes, she also works at hospital for her other job. So that was also a gateway to making it easier to contact other people within the hospital."; "Yes, the pharmacists within [TCIs team] and [person name] are the representatives of the pharmacists. So, in the meantime you have good contact with most pharmacists, so that also helps to motivate them"; "So I think in this way they were the most important triggers for this action." (Project Coordinator, Case B)
	Innovation	19) "The fact that citizens are also involved means that they
	participants	also see that yes, we can play a role in this". (HCP, Case C)
	Engaging organizations, external context	20) "That there are a number of strong players involved, two large GPs practices that are both committed a number of institutions are also involved."(HCP, Case C)

Theme	CFIR (domain/factor)	Quotes	
Imperfect fit between interventions' components and older adults' profile	Intervention characteristics:		
	Design quality and packaging	21) "So those bottlenecks were initially the design of the envelope which was not clear enough for the older population, and something that clearly needed to be addressed was raising awareness among patients and nurses in any case, and also among pharmacists." (Project Coordinator, Case B)	
	Targeted groups	22) "the problems are mainly related to identifying the right patient, and a second problem is once the patients are identified to get them to the right person, and that is something that is not quite running smoothly yet." (HCP, Case C)	
		23) "Being confronted once again with the complexity of the patient group that is indeed at risk, and the complexity is in the medical but also in the social, psychological, element that is really crucial, and that complicates a number of things, such as early care planning." (HCP, Case D)	
	Characteristics of individuals (older adults):		
	Knowledge and beliefs about the intervention	24) "We have also had a patient before who said I have had a lot of explanations, but I didn't understand a thing, so could you please do it again?" (HCP, Case B)	
	Other personal attributes	25) "Asking an 85-year-old to log into an app on his own. That is still difficultthere will always be a generation gap with every modernization." (HCP, Case D)	
Disruption of implementation by COVID-19	Outer setting - miscellaneous:	26) "But we have never actually been able to sit together in real life with all the people from the neighborhood team, which made communication a bit more difficult in the beginning." (HCP, Case C)	
		27) "So yes, that was difficult because of COVID, that not everyone's role was equally clearwho can I talk to and who is here." (Project Coordinator, Case A)	
		28) "That was a real disaster, wasn't it? You saw that within the care program not only on cardiac consultation simply. I think we had a time when 30% of the patients did not show up without calling. If necessary, I would say give the people a tablet so that I can give them their education via the computer, but the people all refused. That really was a period of time; I think it was more than two months that I couldn't include anyone". (HCP, Case D)	
		29) "I think very strongly. On the one hand, there are many bottom-up initiatives. I think much more than in other times. Neighbors helping each other out A lot has been set up. There are also, I think, a lot of agreements with hospitals their discharge policy. I know that the pharmacists have also taken very nice actions with home delivery of medication; so many things have been accelerated. So that has been very nice." (HCP, Case C)	

Table 4. Continued

Notes: CFIR = Consolidated Framework for Implementation Research; HCP = healthcare professional; GP = general practitioner; TCI = transitional care intervention; COVID-19 = coronavirus disease 2019.

*The presented quotes are representative of and can be generalizable across the cases.

ASPECT II – IMPLEMENTATION STRATEGIES

Theme Five: Engagement as a Significant Implementation Strategy Used

Bringing all key actors together, creating knowledge exchange collaborations, capitalizing on existing healthcare providers' unions, and identifying early adopters were seen as the main engaging activities that supported the implementation of the four TCIs. In Case A, project coordinators indicated that healthcare providers in the hospital and intermediate care center, home care nurses, and pharmacists were actively engaged and brought together to establish working agreements for the intervention.

"A moment with healthcare providers from the hospital and the center and another one from home setting and nurses from the center. It was really a moment that they were engaged of making these agreements, which is very important." (Project Coordinator, Case A)

Furthermore, project coordinators explained how they ensured the buy-in and active participation of healthcare providers by communicating the processes required for the interventions and demonstrating the benefit.

"I started to explain this action and always with the nurse or the head nurse of the department and then try to convince them or make them see the benefit of it." (Project Coordinator, Case B)

"We actually just set up these processes and then communicated them to the doctors. They were very happy that there was a process and noticed immediately that it went well. So it didn't really take much effort to get people on board. The team spirit was there from the start...especially with the doctors that went very smoothly." (Project Coordinator, Case A)

Utilizing the existing healthcare providers' unions was another way to bring together all interested parties and key stakeholders (e.g., pharmacists' associations, home care organizations) leading to higher engagement in developing protocols for implementing interventions such as in Case D. Moreover, the implementation of interventions, for example in Cases A and B, was supported by collaborating and engaging with the university hospital in Leuven, which helped project coordinators exchange expertise and knowledge. Similarly for Case C, it was reported how the implementation of caring neighborhood teams was driven by ensuring a bottom-up cooperation and involvement. Therefore, primary healthcare providers with an already innovative idea were identified and invited to take lead in implementing the intervention.

"The early adopters, who are the people already working on things and

who are the quickest to get involved or who want to take the lead. And so, we brought them together to say, this is what we want to do, we are going to start up neighborhood teams ... do you want to cooperate and are you prepared to set up a neighborhood team in your area together with us? So, that is how we approached it." (Project Coordinator, Case C)

Also, the appointment of a reference person with a facilitator role for the intervention was perceived a manner to drive the implementation and help resolve arising issues.

"...that is really very important in implementation that there is a person ... the person of that intervention..." (Project Coordinator, Case B)

Theme Six: Implementation Guided by Overarching Project Plans and Protocols

Participants explained how a formal implementation blue print for the interventions was lacking and instead a general project plan existed. This plan was developed for all interventions combined and in collaboration with healthcare providers and organizations involved; however, it was modified since its initiation and was not followed exactly as it should be.

"Now, of course, that plan is evolving and in the meantime it is already four years old. So, we no longer implement exactly what was in the original plan at the time, but we do implement the broad outlines." (Project Coordinator, Case A)

Protocols, guidance documents, and training plans were developed to support the implementation. However, participants noted that healthcare providers implementing the interventions did not always adhere to the exact project plans or their predefined responsibilities but rather implemented them in a more intuitive manner.

"So there was like a protocol to make sure the operation or implementation would be good." (Project Coordinator, Case A)

"...and then a protocol was drawn up. This is how the care should be for heart failure patients, and these are everyone's responsibilities." (Project Coordinator, Case D)

Theme Seven: Imperative Elements and Suggestive Strategies for Future Sustainability of the Interventions

Participants recommended a number of strategies for the future sustainment of the implementation of the four TCIs. First, securing sufficient and continuous funding for the

interventions was expressed as crucial to maintain the resources (e.g., staff, HIT) needed for implementation.

"There is always a financing side to sustainability." (HCP, Case C)

"...so more funding, more resources, more staff for the primary care areas ... and better means of communication ... better e-health possibilities integrated in the medical files..." (Project Coordinator, Case D)

Second, involving the government and vouching for supportive policies for providers of transitional care (e.g., reimbursement structures) was seen as another key strategy.

"I think the general lesson is that around transition of care, the government should be a real partner of care providers to make this possible. I think that is a precondition. I feel that is still not enough. I also think that the region should be given the freedom to experiment and that sufficient financial resources should be made available to make this possible." (Project Coordinator, Case A)

Third, building strategic partnerships, making formal agreements, and instituting the interventions within large healthcare organizations was indicated as highly needed.

"I think a broader partnership is needed. I say if you want a home care worker at the table locally, then that also has to be coordinated supra-locally and that is why in the future we are going to have a real partner consultation with the strategic partners, where we can make agreements with the management level of home care services ... of umbrella organizations of residential care centers about how their staff can be involved." (HCP, Case C)

Fourth, ensuring the presence of motivated implementers (e.g., champions of the interventions) that lobby continuously for the interventions to keep it going on, as well as building a team capacity with the right skills, was indicated as instrumental for sustainability.

"I think first of all you have to have a permanent team that coordinates everything and that can fill in and handle everything perfectly. And to ensure continuity, who know what they are doing." (HCP, Case D)

The last strategy suggested was to consistently monitor the implementation of the interventions and to obtain convincing data on patient outcomes in order to demonstrate the interventions' benefits and help sustain it.

ASPECT III - IMPLEMENTATION OUTCOMES

Theme Eight: A Satisfactory and Quick-start Implementation

Participants regarded the implementation as favorable, and indicated that the implementation started rather quickly and smoothly with noticeable enthusiasm and collaborative work, see Table 5.

Adoption of the interventions by the healthcare providers was high initially, and many were easily convinced, attracted, and open to adopting the new practices, although it slowed as time passed. The adoption was high among the pharmacists and also providers with a younger age who were more willing to adopt new innovations. Participants indicated that developing an intervention from within; and by the healthcare providers of each community, created a sense of group feeling and promoted its adoption.

Appropriateness of the intervention's components to the care needs of the target population of older adults was sometimes not achieved, according to the participants. For example, in Case C, the neighborhood caring team's intervention provided overarching services to various target groups in the community, which might not fit the specific needs of each group. Participants noted that healthcare providers tend to presume the needs of the older adults and decide on their behalf. However, in some instances healthcare providers were keener to involve and ask the older adults for their needs and then helped them to acquire it.

The interventions were perceived as of high "*acceptability*" and added value to the older adults. Participants indicated that older adults were satisfied and felt supported and acknowledged with the care services provided by the interventions.

"*Fidelity*" to the interventions' core components has changed across the implementation, whereby some were performed in the same manner and as originally planned, but some interventions' components were no longer delivered or were adapted according to the local context (i.e., community needs such as in Case C).

Implementation Outcomes	Quotes
Adoption	 "Yes, the pharmacists are very motivated; the adoption rate is also very high. I think that is because they are actually appreciated for something that they have always done and now they get the right information." (Project Coordinator, Case B) "What shocked me in a positive way is how little energy I had to put into convincing others to participate. I didn't have to convince anyone. It was like, of course we're going to participate certainly in the beginning I didn't have to make any effort to draw people into this story." (HCP, Case C)
Appropriateness	 "Because sometimes they themselves are not well they need care Yes, I sometimes have the feeling that perhaps without realizing it, you are deciding too hard for them, which is best for them." (HCP, Case C) "The important thing is that you also ask your patients what do you want and what are your goals? And we actively questioned that, because we had an objective scale, but we also wanted to look at 'Do you want to go back home, yes, okay, what can we help you with? How is your home situation? But also, how can we help you physically?' So, you need that active participation from your patient anyway." (HCP, Case A)
Acceptability	"Very satisfying in that way that they have to worry a lot lessI think that is positive for them, that they feel more acknowledged and that in itself provides a more positive experience." (HCP, Case B)
Fidelity	"We have planted the seed and put forward the idea, but we have said this is the way that you could do it, but as a neighborhood team you may want to decide to do it differently, as long as you make sure that you do population management" (Project Coordinator, Case C)

 Table 5. Quotes illustrating the implementation outcomes of the four transitional care interventions (TCls).

Note: HCP = healthcare professional.

DISCUSSION

Findings revealed that the prominent implementation barriers of the four TCIs were linked to the organizational setting and included insufficient resources and funding, small structure with low capacity, suboptimal internal work networks and communication, and discontinuous information exchange between care providers. On the other hand, the project coordinators and healthcare providers' great motivation and commitment, as well as strong beliefs and favorable attributes to initiate and drive the implementation, were facilitators. Also, the presence of champions for the TCIs fostered the process further. In this study, strategies used to implement the interventions were limited yet largely focused on engaging the right people, such as early adopters, key actors, and existing partners. Participants suggested other key strategies are needed (e.g., monitoring, lobbying for transitional care policies) to continue the implementation of the four TCIs in the future. Overall, participants perceived the implementation outcomes as favorable, as indicated by quick and high adoption, as well as general acceptability, yet participants also reported variable appropriateness of the interventions' components to the needs of the older adults.

The current results are in line with our previous work on factors influencing the implementation of TCIs in general, as reported in a scoping review (16). However, unlike the scoping review, in this case study we found a clear distinction implying that key barriers

belonged to the organizational setting while facilitators were linked to the characteristics of individuals and the implementation process (engaging). Similarly, Lutz and colleagues (39) showed that healthcare providers' willingness, commitment, and ownership were enablers for the implementation of TCIs. Our results were further mirrored in a systematic review on implementing integrated care interventions, which identified factors at organizational and healthcare system levels (e.g. limited staffing capacity, poor communication, restrictions in funding reimbursement systems) as implementation barriers (40). In contrast with this, however, it was seen in other studies on the implementation of TCIs that individual-level factors behaved more as barriers or had a mixed influence (16, 41). Another important barrier in this study was the lack of coordinated and well-structured national health policies to support the implementation of the four TCIs. Likewise, this resonates with recent evidence from Belgium indicating that its current federal government structure, healthcare financing system, and lack of digital system/data sharing among providers hinders the implementation of care integration in general (42). One factor we missed in this study was leadership which was not pointed out explicitly as a crucial factor to the implementation, as frequently recognized in the literature (43). Perhaps this could be because the project coordinators and healthcare professionals saw themselves as the leaders and facilitators of the four TCIs, hence driving through the implementation accordingly.

Our analysis deduced that engagement was the major implementation strategy used, although it was performed intuitively, and without any previous decision. It is most likely that the strategies used came habitually to project coordinators and healthcare providers and were outside their awareness of growing evidence indicating the essential role of using implementation strategies to put new interventions in practice. Nonetheless, these strategies coincide with known ones (e.g., facilitation, conducting education/training meetings, obtaining work commitments) being used in the implementation of other TCIs (44). At the same time, in this study there was no local needs assessment, identification of barriers and facilitators beforehand, or development of monitoring systems, which are commonly recommended implementation strategies (21).

According to our study, the individuals' realm of personality, attributes, beliefs, and cognition was a distinguishable facilitator to the implementation. The underlying key lever here was their continuing motivation coupled with a strong intention to bring about the change (behave differently to implement the four TCIs) in order to achieve expected outcomes. Our results confirm and expand the existing evidence regarding the role of human agency in changing behavior, as explained by various social cognitive/behavioral theories (45, 46). Notably, it was established that motivation is essential to both instigate and direct behavior, especially new behavior (47). As per Michie et al.'s behavior system, motivation sits at the core and in between the individual's capability (physical, psychological) and opportunity (all factors outside the individual, i.e. context) and can directly induce behavior (45). Therefore, it is of

no surprise that implementing new interventions, such as TCIs, should involve considering this behavior system and choosing strategies to leverage the individual's motivation and capability (48). Hence, focusing on the internal factors of individuals has a high potential to achieve a target behavior and thus implement new interventions.

Even though individual factors appear promising for enabling a successful implementation, individuals are often part of a whole organization. Organizational factors (mainly unavailability of resources) are frequently reported, or better "blamed" as we saw in our study, for hindering the implementation of new interventions in transitional care. This rhetoric of organizational barriers necessitates further exploration. The concept of "organizational adaptation" is relevant, whereby organizations can rearrange their existing capabilities (e.g., operational capacity, infrastructure, financial resources) to implement a new intervention (49). As an example, a hospital wanting to incorporate a transition care nurse (considered a TCI) can do minimal adaptation by changing the job duties/description of a present frontline nurse without a new hire. This relates to looking inside organizations for slack resources - a cushion of extra staff, time, and space - that goes usually under-assessed but can actually be used for implementing a new intervention (50). Organizations can be more dynamic in utilizing their capabilities to implement change by continuously reflecting/tweaking their inefficient work routines or taking low-cost initiatives (e.g., form new alliances between hospitals and homecare services in one region to enhance care transitions) (51). Nonetheless, these strategies are ultimately linked to the presence of individuals within the organization who also possess dynamic and influential capacities sufficient to foster change.

STRENGTHS AND LIMITATIONS

The present study has some limitations. Selection bias can play a role, however the sampling methods used allowed us to obtain insights from the core individuals involved who were the closest to the implementation process. The number of participants and their occupations as interviewed per case was not evenly distributed, yet we chose to combine the data in one unit of analysis. Also, if more information on the participants' age and length of time in each profession were available, it could have added to the interpretation of the findings. We note that the representation of diversity in a study sample is important. Therefore, if participants of other ethnicities were included, there might have been more diverse viewpoints presented, other elucidations to data patterns, and possibly better generalization of the results. The retrospective data collected were based on self-reporting and reflection which could be subject to personal recall biases. Nevertheless, our study's strength lies in providing a broad and in-depth understanding of how the implementation of TCls occurs in real-life, using implementation science approach.

IMPLICATIONS FOR PRACTICE AND RESEARCH

Given both the lack of insight about the influencing factors and the absence of a deliberate selection of implementation strategies prior to implementing the four TCIs, we hereby propose recommendations to implement TCIs using implementation science concepts.

- Understand the context early on Prior to any implementation effort, a thorough assessment of the contextual factors is vital and gives a heads-up to implementers. Checking what can hinder/enable the implementation in a specific context can help capture the complexity of the settings involved, especially in transitional care. This allows implementers to understand the capabilities and opportunities existing in their current context, and whether a new TCI has a chance to be implemented.
- Use implementation strategies Choosing strategies from the various available taxonomies can guide the implementers on how to best implement a TCI (21). Specific strategies must be carefully selected according to their effectiveness as well as the ability to address the relevant influencing factors.
- 3. Empower the people and forge partnerships Implementing TCIs is a team activity and requires leveraging the personal factors of the individuals involved. Knowing what motivates, activates, and inspires the individuals and offering it to them can support the implementation of a new TCI. Also, creating partnerships is nevertheless critical in transitional care; involving the key actors necessary for implementing a TCI can only propel the process.
- 4. Research First, future studies should examine more bottom-up initiatives of implementing TCIs performed with an intuitive implementation approach. This would allow comparison of such studies with planned pilots and trials. Second, developing implementation strategies tailored to TCIs and testing its effectiveness in practice is needed.

REFERENCES

- 1. Palladino R, Pennino F, Finbarr M, Millett C, Triassi M. Multimorbidity and health outcomes in older adults in ten European health systems, 2006-15. Health Aff (Project Hope). 2019;38(4):613-23. 10.1377/hlthaff.2018.05273
- Barnett K, Mercer SW, Norbury M, Watt G, Wyke S, Guthrie B. Epidemiology of multimorbidity and implications for health care, research, and medical education: a cross-sectional study. Lancet (London, England). 2012;380(9836):37-43. 10.1016/s0140-6736(12)60240-2
- Baxter R, Shannon R, Murray J, O'Hara JK, Sheard L, Cracknell A, et al. Delivering exceptionally safe transitions of care to older people: a qualitative study of multidisciplinary staff perspectives. BMC Health Serv Res. 2020;20(1):780. 10.1186/s12913-020-05641-4
- 4. Coleman EA. Falling through the cracks: challenges and opportunities for improving transitional care for persons with continuous complex care needs. J Am Geriatr Soc. 2003;51(4):549-55. 10.1046/j.1532-5415.2003.51185.x
- Scott AM, Li J, Oyewole-Eletu S, Nguyen HQ, Gass B, Hirschman KB, et al. Understanding facilitators and barriers to care transitions: insights from project ACHIEVE site visits. Jt Comm J Qual Patient Saf. 2017;43(9):433-47. 10.1016/j.jcjq.2017.02.012
- 6. WHO. Integrated care models: an overview. WHO 2016.
- Brown CL, Menec V. Integrated care approaches used for transitions from hospital to community care: a scoping review. Canadian Journal on Aging/La Revue canadienne du vieillissement. 2018;37(2):145-70. 10.1017/ s0714980818000065
- 8. Goodwin N. Understanding integrated care. Int J Integr Care. 2016;16(4):6-. 10.5334/ijic.2530
- Deschodt M, Laurent G, Cornelissen L, Yip O, Zúñiga F, Denhaerynck K, et al. Core components and impact of nurse-led integrated care models for home-dwelling older people: A systematic review and meta-analysis. Int J Nurs Stud. 2020;105:103552. 10.1016/j.ijnurstu.2020.103552
- 10. Antunes V, Moreira JP. Approaches to developing integrated care in Europe: a systematic literature review. Journal of Management & Marketing in Healthcare. 2011;4(2):129-35. 10.1179/175330311X13016677137743
- 11. Lee JY, Yang YS, Cho E. Transitional care from hospital to home for frail older adults: A systematic review and meta-analysis. Geriatr Nurs. 2022;43:64-76. https://doi.org/10.1016/j.gerinurse.2021.11.003
- 12. Naylor MD, Bowles KH, McCauley KM, Maccoy MC, Maislin G, Pauly MV, et al. High-value transitional care: translation of research into practice. J Eval Clin Pract. 2013;19(5):727-33. 10.1111/j.1365-2753.2011.01659.x
- 13. de Bruin SR, Stoop A, Billings J, Leichsenring K, Ruppe G, Tram N, et al. The SUSTAIN project: a European study on improving integrated care for older people living at home. Int J Integr Care. 2018;18(1):6-. 10.5334/ijic.3090
- Stadnick NA, Sadler E, Sandall J, Turienzo CF, Bennett IM, Borkan J, et al. Comparative case studies in integrated care implementation from across the globe: a quest for action. BMC Health Serv Res. 2019;19(1):899. 10.1186/ s12913-019-4661-5
- May CR, Johnson M, Finch T. Implementation, context and complexity. Implement Sci. 2016;11(1):141. 10.1186/ s13012-016-0506-3
- Fakha A, Groenvynck L, de Boer B, van Achterberg T, Hamers J, Verbeek H. A myriad of factors influencing the implementation of transitional care innovations: a scoping review. Implement Sci. 2021;16(1):21. 10.1186/s13012-021-01087-2
- 17. Ling T, Brereton L, Conklin A, Newbould J, Roland M. Barriers and facilitators to integrating care: experiences from the English Integrated Care Pilots. Int J Integr Care. 2012;12:e129-e. 10.5334/ijic.982
- Maruthappu M, Hasan A, Zeltner T. Enablers and barriers in implementing integrated care. Health Syst Reform. 2015;1(4):250-6. 10.1080/23288604.2015.1077301
- Threapleton DE, Chung RY, Wong SYS, Wong E, Chau P, Woo J, et al. Integrated care for older populations and its implementation facilitators and barriers: A rapid scoping review. Int J Qual Health Care. 2017;29(3):327-34. 10.1093/intqhc/mzx041
- Proctor EK, Powell BJ, McMillen JC. Implementation strategies: recommendations for specifying and reporting. Implement Sci. 2013;8(1):139. 10.1186/1748-5908-8-139
- Powell BJ, Waltz TJ, Chinman MJ, Damschroder LJ, Smith JL, Matthieu MM, et al. A refined compilation of implementation strategies: results from the Expert Recommendations for Implementing Change (ERIC) project. Implement Sci. 2015;10(1):21. 10.1186/s13012-015-0209-1
- 22. Cochrane. Effective Practice and Organisation of Care (EPOC). EPOC Taxonomy. 20152015 [Available from: epoc.cochrane.org/epoc-taxonomy].
- Powell BJ, Fernandez ME, Williams NJ, Aarons GA, Beidas RS, Lewis CC, et al. Enhancing the impact of implementation strategies in healthcare: a research agenda. Front Public Health. 2019;7. 10.3389/ fpubh.2019.00003
- 24. Integreo. Les Soins Integres 2018 [Available from: https://www.integreo.be/fr/soins-integres/contexte].
- Goderis G, Colman E, Irusta LA, Van Hecke A, Pétré B, Devroey D, et al. Evaluating Large-Scale Integrated Care Projects: The Development of a Protocol for a Mixed Methods Realist Evaluation Study in Belgium. Int J Integr Care. 2020;20(3):12. 10.5334/ijic.5435

- 26. ZorgzaamLeuven. Actieplan 2018 [Available from:https://www.zorgzaamleuven.be/actieplan]
- 27. Winter MD. Reshaping health care governance using pilot projects as public policy implementation instruments. International Review of Public Policy. 2020;2:3. 10.4000/irpp.1422
- 28. Yin RK. Case study research design and methods. 4th ed. Los Angeles: Sage 2009.
- Crowe S, Cresswell K, Robertson A, Huby G, Avery A, Sheikh A. The case study approach. BMC Med Res Methodol. 2011;11(1):100. 10.1186/1471-2288-11-100
- 30. Stake RE. The art of case study research: Sage 1995.
- Hamilton AB, Finley EP. Qualitative methods in implementation research: An introduction. Psychiatr Res. 2019;280:112516. 10.1016/j.psychres.2019.112516
- Palinkas LA, Horwitz SM, Green CA, Wisdom JP, Duan N, Hoagwood K. Purposeful sampling for qualitative data collection and analysis in mixed method implementation research. Adm Policy Ment Health. 2015;42(5):533-44. 10.1007/s10488-013-0528-y
- Guest G, Namey E, McKenna K. How many focus groups are enough? Building an evidence base for nonprobability sample sizes. Field Methods. 2016;29(1):3-22. 10.1177/1525822X16639015
- Damschroder LJ, Aron DC, Keith RE, Kirsh SR, Alexander JA, Lowery JC. Fostering implementation of health services research findings into practice: a consolidated framework for advancing implementation science. Implement Sci. 2009;4(1):50. 10.1186/1748-5908-4-50
- Proctor E, Silmere H, Raghavan R, Hovmand P, Aarons G, Bunger A, et al. Outcomes for implementation research: conceptual distinctions, measurement challenges, and research agenda. Adm Policy Ment Health. 2011;38(2):65-76. 10.1007/s10488-010-0319-7
- 36. CFIR. Interview Guide Tools 2022 [Available from: https://cfirguide.org/tools/].
- Braun V, Clarke V. Using thematic analysis in psychology. Qual. Res. Psychol. 2006;3(2):77-101. 10.1191/1478088706qp063oa
- 38. CFIR. Constructs Qualitative codebook guidelines 2022 [Available from: https://cfirguide.org/constructs/].
- Lutz BJ, Reimold AE, Coleman SW, Guzik AK, Russell LP, Radman MD, et al. Implementation of a Transitional Care Model for Stroke: Perspectives From Frontline Clinicians, Administrators, and COMPASS-TC Implementation Staff. Gerontologist. 2020;60(6):1071-84. 10.1093/geront/gnaa029
- 40. Sadler E, Potterton V, Anderson R, Khadjesari Z, Sheehan K, Butt F, et al. Service user, carer and provider perspectives on integrated care for older people with frailty, and factors perceived to facilitate and hinder implementation: A systematic review and narrative synthesis. PLOS ONE. 2019;14(5):e0216488. 10.1371/journal. pone.0216488
- Geerligs L, Rankin NM, Shepherd HL, Butow P. Hospital-based interventions: a systematic review of staffreported barriers and facilitators to implementation processes. Implement Sci. 2018;13(1):36. 10.1186/s13012-018-0726-9
- Danhieux K, Martens M, Colman E, Wouters E, Remmen R, van Olmen J, et al. What makes integration of chronic care so difficult? A macro-level analysis of barriers and facilitators in Belgium. Int J Integr Care. 2021;21(4):8-. 10.5334/ijic.5671
- Fakha A, de Boer B, van Achterberg T, Hamers J, Verbeek H. Fostering the implementation of transitional care innovations for older persons: prioritizing the influencing key factors using a modified Delphi technique. BMC Geriatr. 2022;22(1):131. 10.1186/s12877-021-02672-2
- 44. Toles M, Frerichs A, Leeman J. Implementing transitional care in skilled nursing facilities: evaluation of a learning collaborative. Geriatr Nurs. 2021;42(4):863-8. 10.1016/j.gerinurse.2021.04.010
- 45. Michie S, van Stralen MM, West R. The behaviour change wheel: a new method for characterising and designing behaviour change interventions. Implement Sci. 2011;6(1):42. 10.1186/1748-5908-6-42
- 46. Bandura A. Social cognitive theory: an agentic perspective. Annu Rev Psychol. 2001;52(1):1-26. 10.1146/annurev. psych.52.1.1
- 47. Schunk DH, DiBenedetto MK. Motivation and social cognitive theory. Contemp Educ Psychol. 2020;60:101832. https://doi.org/10.1016/j.cedpsych.2019.101832
- Michie S, Richardson M, Johnston M, Abraham C, Francis J, Hardeman W, et al. The behavior change technique taxonomy (v1) of 93 hierarchically clustered techniques: building an international consensus for the reporting of behavior change interventions. Ann Behav Med. 2013;46(1):81-95. 10.1007/s12160-013-9486-6
- Lengnick-Hall R, Willging CE, Hurlburt MS, Aarons GA. Incorporators, early investors, and learners: a longitudinal study of organizational adaptation during EBP implementation and sustainment. Implement Sci. 2020;15(1):74. 10.1186/s13012-020-01031-w
- Mallidou AA, Cummings GG, Ginsburg LR, Chuang Y-T, Kang S, Norton PG, et al. Staff, space, and time as dimensions of organizational slack: a psychometric assessment. Health Care Manage Rev. 2011;36(3):252-64. 10.1097/HMR.0b013e318208ccf8
- 51. Karali E, Angeli F, Sidhu JS, Volberda H. Understanding healthcare innovation through a dynamic capabilities lens. Healthcare Entrepreneurship. UK: Routledge; 2018.

APPENDICES

Appendix 4A. Interview guide used in the semi-structured interviews

Appendix 4B. Illustrations of Coding

Appendix 4C. Consolidated Framework for Implementation Research (CFIR) Constructs Description

Appendix 4A. Interview guide used in the semi-structured interviews

(Note – please refer to the underlined questions which are the ones pertaining to implementation and which were used to collect data for this study. The other questions were used for a second study).

Questions – Project coordinator

- 1. Can you shortly describe the action and your role in those actions?
- 2. Which stakeholders and healthcare professionals were involved in those actions?
- 3. What did you do to encourage, motivate and engage healthcare professionals to commit to using these actions? Did you assign any champions, leaders, persons with a facilitator role, or any other assigned roles to help spread those actions to the healthcare providers?
- 4. Now we will focus on the aspect transition of care for elderly. What does transitional care mean for you?
- 5. Can you shortly describe the impact of the action on the transition of care for elderly people?
- a. Communication
- b. Shared-decision making/ patient involvement/ informal caregiver involvement
- c. Person-centered care
- d. Medication reconciliation
- e. Continuity of care at home (organization of follow-up care)
- 6. Can you shortly describe the impact of the action on the healthcare professionals?
- 7. Can you shortly describe the impact of the action on the healthcare system?
- 8. To what extent are those actions fully integrated? What stage are they at?
 - a. <u>Status of implementation</u>
 - b. Does the current state of implementation of those actions meet your expectations?
 - c. <u>Is the action still carried out as planned?</u>
 - i. How has the action been affected in the context of COVID-19
 - ii. Sustainability of the project
- 9. <u>Did you develop a project plan to implement and roll out those actions to the various</u> <u>healthcare providers? If yes, can you briefly describe the plan?</u>
 - a. <u>Did you perform any activities or use specific strategies to implement those</u> <u>actions? (e.g.: education and training, champions, mandate change)</u>
- 10. <u>Did you already receive some feedback from primary and secondary healthcare</u> providers? If yes, what did you learn about it?
- 11. Adoption (= intention to try to use the intervention)
 - a. <u>How would you assess (or what would you say) on the adoption level of these</u> actions by healthcare professionals? This means the intention to use the intervention by the healthcare professionals.
 - b. What are the barriers/obstacles/challenges in the implementation of those actions?

And were there any facilitators to enable a better and successful implementation of those actions?

- 12. What is needed to continue these actions?
- 13. How can this action be expanded?
- 14. Which lessons could be learned from this action regarding the future?

Completing the interview

Would you like to add something to this interview? Thank you for your participation.

Questions – Healthcare professional

- Can you shortly describe the action and your role in those action?
 a. How do you experience this role?
- 2. Which other persons are involved?
 - a. How did you experience the collaboration and the communication? Is everyone aware of his/her responsibilities?
- 3. Now we will focus on the aspect transition of care for elderly. What does transitional care mean for you?
- 4. Can you shortly describe the impact of the action on the transition of care for elderly people?
 - a. Communication
 - b. Shared-decision making/ patient involvement/ informal caregiver involvement
 - c. Person-centered care
 - d. Medication reconciliation
 - e. Continuity of care at home (organization of follow-up care)
- 5. Can you shortly describe the impact of the action on the healthcare professionals?
- 6. Can you shortly describe the impact of the action on the healthcare system?
- Acceptability (perceived views that an intervention is agreeable, satisfactory, credible, and comfortable): to what extent do you think those actions are satisfactory and advantageous to the older persons with chronic disease and requiring care transitions between hospital & home?
- 8. <u>Appropriateness (perceived compatibility of the intervention with needs & practices of a setting or population): to what extent do you think these actions address/meet the care needs of older persons with chronic disease and requiring care transitions?</u>
- 9. <u>Is the action still performed as planned?</u>
 - a. <u>Status of implementation</u>
 - b. How has the action been affected in the context of COVID-19?
- 10. <u>Sustainability (extent to which an intervention is routinized or maintained within an</u> organization): How would you ensure that these actions become a routine part (used regularly) of the daily work of the healthcare professionals within your team/in your

organization?

- 11. Experiences and expectations:
 - a. How did you experience those actions?
 - b. Is the action meeting your expectation? Can you explain the reasons as to why it is or isn't meeting your expectations?
 - c. According to you, do you think the implementation of the action is successful? Can you explain the reasons as to why the implementation is or isn't successful?
- 12. What are the barriers/obstacles/challenges in the implementation of those actions? And were there any facilitators to enable a better and successful implementation of those actions?
- 13. Which lessons could be learned from this action regarding the future?

Completing the interview

Would you like to add something to this interview? Thank you for your participation.

Appendix 4B. Illustrations of Coding

*n: number of codes per theme






Appendix 4C. Consolidated Framework for Implementation Research (CFIR) Constructs Description

CEIP	CFIR constructs / CTF* selected constructs Short Description				
	main: INTERVENTION CHARACTERIST	•			
A.	Intervention Source	Perception of key stakeholders about whether the intervention is			
^	intervention source	externally or internally developed.			
в	Evidence Strength & Quality	Stakeholders' perceptions of the quality and validity of evidence supporting the belief that the intervention will have desired outcomes.			
с	Relative Advantage	Stakeholders' perception of the advantage of implementing the intervention versus an alternative solution.			
D	Adaptability	The degree to which an intervention can be adapted, tailored, refined, or reinvented to meet local needs.			
E	Trialability	The ability to test the intervention on a small scale in the organization, and to be able to reverse course (undo implementation) if warranted.			
F	Complexity	Perceived difficulty of implementation, reflected by duration, scope, radicalness, disruptiveness, centrality, and intricacy and number of steps required to implement.			
G	Design Quality & Packaging	Perceived excellence in how the intervention is bundled, presented, and assembled.			
н	Cost	Costs of the intervention and costs associated with implementing the intervention including investment, supply, and opportunity costs.			
CTF	Targeted Groups	Staff and others (vendors, patients) who are the intended recipients or beneficiaries of the intervention.			
II. Do	main: OUTER SETTING				
Α	Patient Needs & Resources	The extent to which patient needs, as well as barriers and facilitators to meet those needs, are accurately known and prioritized by the organization.			
в	Cosmopolitanism	The degree to which an organization is networked with other external organizations.			
с	Peer Pressure	Mimetic or competitive pressure to implement an intervention; typically because most or other key peer or competing organizations have already implemented or are in a bid for a competitive edge.			
D	External Policy & Incentives	A broad construct that includes external strategies to spread interventions, including policy and regulations (governmental or other central entity), external mandates, recommendations and guidelines, pay-for-performance, collaboratives, and public or benchmark reporting.			
CTF	Community Resources	Availability and access of service providers, aging resources, and multiple levels of community services and supports not directly involved in the intervention.			
III. D	III. Domain: INNER SETTING				
Α	Structural Characteristics	The social architecture, age, maturity, and size of an organization.			
в	Networks & Communications	The nature and quality of webs of social networks and the nature and quality of formal and informal communications within an organization.			
с	Culture	Norms, values, and basic assumptions of a given organization.			
D	Implementation Climate:	The absorptive capacity for change, shared receptivity of involved individuals to an intervention, and the extent to which use of that intervention will be rewarded, supported, and expected within their organization.			
D.1	Tension for Change	The degree to which stakeholders perceive the current situation as intolerable or needing change.			

CFIR	constructs / CTF* selected constructs	Short Description
D.2	Compatibility	The degree of tangible fit between meaning and values attached to the intervention by involved individuals, how those align with individuals' own norms, values, and perceived risks and needs, and how the intervention fits with existing workflows and systems.
D.3	Relative Priority	Individuals' shared perception of the importance of the implementation within the organization.
D.4	Organizational Incentives & Rewards	Extrinsic incentives such as goal-sharing awards, performance reviews, promotions, and raises in salary, and less tangible incentives such as increased stature or respect.
D.5	Goals and Feedback	The degree to which goals are clearly communicated, acted upon, and fed back to staff, and alignment of that feedback with goals.
D.6	Learning Climate	A climate in which: a) leaders express their own fallibility and need for team members' assistance and input; b) team members feel that they are essential, valued, and knowledgeable partners in the change process; c) individuals feel psychologically safe to try new methods; and d) there is sufficient time and space for reflective thinking and evaluation.
E	Readiness for Implementation:	Tangible and immediate indicators of organizational commitment to its decision to implement an intervention.
E.1	Leadership Engagement	Commitment, involvement, and accountability of leaders and managers with the implementation.
E.2	Available Resources	The level of resources dedicated for implementation and on-going operations, including money, training, education, physical space, and time.
E.3	Access to Knowledge & Information	Ease of access to digestible information and knowledge about the intervention and how to incorporate it into work tasks.
CTF	IT and HIT Resources:	Technological infrastructure in place to support electronic information management, including IT that crosses organizations.
CTF	HIT Systems	Electronic information management infrastructure and technologies available to clinicians to manage patient care, data, and communications.
CTF	Continuity	Information continuity (exchange of information) and relationship continuity, both with providers and patients/caregivers and across organizations.
IV. D	omain: CHARACTERISTICS OF INDIVID	DUALS
A	Knowledge & Beliefs about the Intervention	Individuals' attitudes toward and value placed on the intervention as well as familiarity with facts, truths, and principles related to the intervention.
в	Self-efficacy	Individual belief in their own capabilities to execute courses of action to achieve implementation goals.
С	Individual Stage of Change	Characterization of the phase an individual is in, as he or she progresses toward skilled, enthusiastic, and sustained use of the intervention.
D	Individual Identification with Organization	A broad construct related to how individuals perceive the organization, and their relationship and degree of commitment with that organization.
E	Other Personal Attributes	A broad construct to include other personal traits such as tolerance of ambiguity, intellectual ability, motivation, values, competence, capacity, and learning style.
V. Do	omain: PROCESS	
Α	Planning:	The degree to which a scheme or method of behavior and tasks for implementing an intervention are developed in advance, and the quality of those schemes or methods.
В	Engaging:	Attracting and involving appropriate individuals in the implementation and use of the intervention through a combined strategy of social marketing, education, role modeling, training, and other similar activities.

Appendix 4C. Continued

Appendix 4C. Continued

CFIR	constructs / CTF* selected constructs	Short Description
B.1	Opinion Leaders	Individuals in an organization who have formal or informal influence on the attitudes and beliefs of their colleagues with respect to implementing the intervention.
B.2	Formally Appointed Internal Implementation Leaders	Individuals from within the organization who have been formally appointed with responsibility for implementing an intervention as coordinator, project manager, team leader, or other similar role.
B.3	Champions	"Individuals who dedicate themselves to supporting, marketing, and 'driving through' an [implementation]", overcoming indifference or resistance that the intervention may provoke in an organization.
В.4	External Change Agents	Individuals who are affiliated with an outside entity who formally influence or facilitate intervention decisions in a desirable direction.
B.5	Key Stakeholders**	Individuals from within the organization that are directly impacted by the innovation, e.g., staff responsible for making referrals to a new program or using a new work process.
B.6	Innovation Participants**	Individuals served by the organization that participate in the innovation, e.g., patients in a prevention program in a hospital.
CTF	Engaging Organizations, External Context	Developing and capitalizing on relationships with providers, leaders, and frontline staff in the implementing organizations, and to external providers, resources, funders.
с	Executing:	Carrying out or accomplishing the implementation according to plan.
D	Reflecting & Evaluating:	Quantitative and qualitative feedback about the progress and quality of implementation accompanied with regular personal and team debriefing about progress and experience.

*Few constructs from the Care Transitions Framework (CTF) were added and used here within CFIR. CTF is based on the CFIR with additional constructs for care transitions (check below reference 2). **Two additional constructs (engaging: key stakeholders, innovation participants) under engaging in the process domain were added as per CFIR research group (<u>https://cfirguide.org/</u>).

REFERENCES:

- Damschroder LJ, Aron DC, Keith RE, Kirsh SR, Alexander JA, Lowery JC. Fostering implementation of health services research findings into practice: a consolidated framework for advancing implementation science. Implement Sci. 2009;4:50.
- Dy SM, Ashok M, Wines RC, Rojas Smith L. A framework to guide implementation research for care transitions interventions. J Healthc Qual. 2015;37(1):41-54.



CHAPTER

Stakeholders' Experiences and Perception on Transitional Care Initiatives within an Integrated Care Project in Belgium: A Qualitative Interview Study

Published as:

Leithaus M, **Fakha A**, Flamaing J, Verbeek H, Deschodt M, van Pottelbergh G, Goderis G. Stakeholders' experiences and perception on transitional care initiatives within an integrated care project in Belgium: a qualitative interview study. BMC Geriatr. 2023; 23(1):41. https://doi.org/10.1186/s12877-023-03746-z

ABSTRACT

BACKGROUND

In 2015, a plan for integrated care was launched by the Belgium government that resulted in the implementation of 12 integrated care pilot project across Belgium. The pilot project Zorgzaam Leuven consists of a multidisciplinary local consortium aiming to bring lasting change towards integrated care for the region of Leuven. This study aims to explore experiences and perceptions of stakeholders involved in four transitional care actions that are part of Zorgzaam Leuven.

METHODS

This qualitative case study is part of the European TRANS-SENIOR project. Four actions with a focus on improving transitional care were selected and stakeholders involved in those actions were identified using the snow-ball method. Fourteen semi-structured interviews were conducted and inductive thematic analysis was performed.

RESULTS

Professionals appreciated to be involved in the decision making early onwards either by proposing own initiatives or by providing their input in shaping actions. Improved team spirit and community feeling with other health care professionals (HCPs) was reported to reduce communication barriers and was perceived to benefit both patients and professionals. The actions provided supportive tools and various learning opportunities that participants acknowledged. Technical shortcomings (e.g. lack of integrated patient records) and financial and political support were identified as key challenges impeding the sustainable implementation of the transitional care actions.

CONCLUSION

The pilot project Zorgzaam Leuven created conditions that triggered work motivation for HCPs. It supported the development of multidisciplinary care partnerships at the local level that allowed early involvement and increased collaboration, which is crucial to successfully improve transitional care for vulnerable patients.

KEYWORDS

Integrated care, interdisciplinary communication, continuity of patient care, health care policy, qualitative research.

BACKGROUND

The ageing of the population puts a burden on current health care systems. Older patients with multiple chronic conditions often have complex health care needs over a long period of time. The need for complex health care services frequently leads to health care transitions between different locations or at different levels of care within the same location (1). Poorly performed healthcare transitions can cause harm, such as poor clinical outcomes, missed diagnosis, incorrect treatment, dissatisfaction among patients, inappropriate use of healthcare services, rehospitalization and mortality (2).

Integrated care initiatives are recommended to ensure continuity of health care and thus particularly benefit older chronically ill patients who often experience transitions. Integrated care is defined as "the management and delivery of health services so that clients receive a continuum of preventive and curative services, according to their needs over time and across different levels of the health system" (3). Research shows that the integration of health care can result into improved access to services, reduced hospitalizations and readmissions, enhanced adherence to treatment, increased patient satisfaction, improved health literacy and self-care, greater job satisfaction of health care workers, and overall improved health outcomes (4).

Integrated care initiatives that support continuity in care across boundaries require interdisciplinary collaboration, however, establishing consistent multidisciplinary work structures is a complex task (5, 6). It is known that implementing integrated care programs is difficult as it is often co-determined by unique dynamics and characteristics that can hinder or facilitate implementation within and across health care settings (5, 7). Studies suggest that attitudes of health-care professionals towards change is crucial. Involving professionals during the development of integrated care programs encourages closer team work early onwards and allows professionals to understand their role as part of the whole (8).

The project 'Integrated Care for Better Health' (Integreo) that was launched by the Belgian government in 2016, follows this perspective of including stakeholders in the design of integrated care initiatives (9). In Integreo, 12 pilot projects for integrated care were designed in co-creation with interested professionals from different local regions (10). As a result, multidisciplinary local consortia were created to develop a plan with common visions and objectives within a certain geographical area in Belgium. This approach allowed professionals to share their hands on experience while creating a common plan for integrated care. While it is desirable to involve professionals in the development process, it is an extensive process that often leads to challenges, such as additional workload, conflicting opinions and regular changes in scope. These challenges often create an uncomfortable climate of uncertainty, which might create additional burden for health care professionals (10).

This paper focuses on Zorgzaam Leuven (ZZL), one of the 12 Integreo projects (11), by presenting four actions of ZZL that aim to achieve integrated care across settings, thereby improving transitional care and the quality of care interactions. The aim of the paper is twofold. First, we describe the four selected actions that focus on improving transitional care for older chronically ill patients. Second, we aim to get a deeper understanding on the experiences and perceptions of involved stakeholders (health care professionals (HCP) and project coordinators) on transitional care at local level and in the everyday practice.

METHODS

DESIGN

The study used a qualitative naturalistic case study design, which allows to gain understanding of a complex issue in its real life context (12). The pilot project ZZL was defined as the case, while the four transitional care actions were selected to study the case. To understand individual perceptions, semi-structured interviews were used as data collection method.

STUDY SETTING

In 2015 the joint plan for chronic care 'Integreo' was agreed on by all health ministers in Belgium. This resulted in the start of implementing integrated care pilot projects with the aim to develop and test integrated care initiatives in 12 different regions in Belgium (13). Each pilot project covers a different geographical region between 75,000 and 360,000 inhabitants. The Belgium government defined 14 components of integrated care to be implemented by the projects. The local project team were allowed to define care goals and the target population in order to implement integrated care where most needed and suited for their region (14). Thus, new care initiatives can be developed and tested within the 12 projects. The ultimate goal of Integreo is to improve the quadruple aim objectives for healthcare: improving population health and patient experience, reducing costs, and enhancing well-being of health care providers (14, 15).

ZZL is conceptualized as a project to bring lasting change towards integrated care for the region of Leuven, Belgium. The pilot phase of the project took place from 2017 to 2022. ZZL consists of a multidisciplinary local consortium with a core team of ten part-time project coordinators and more than 60 local organizations from the region, including home care organizations, pharmacies, GP practices, regional hospitals, non-profit organizations and research organizations (16). It covers the region of Leuven and consists of the following sub municipalities: Leuven center, Heverlee, Kessel-Lo, Wijgmaal and Wilsele, including 100,516 inhabitants in October 2021 (17). Leuven is known as a university city and is therefore characterized by a relatively young population (18). The percentage of people aged 65+ in

Leuven is 16.7% in 2021 (17) while the regional average in Flanders was 20.2% in 2019 (19). Two hospitals are located in the project region: the regional hospital Heilig Hart Leuven (287 beds) and the University Hospitals Leuven (1995 beds). Within this region ZZL acts as a vehicle to bring different actors together and to promote and initiate change in co-creation.

The multidisciplinary local consortium of ZZL started by conceptualizing a large project plan for the region that was based on three pillars: 1. population-oriented thinking, 2. caring neighborhood concept, and 3. smart-actions in co-creation (11). The project plan provides a framework for the consortia by outlining the target population and the main six themes for action. The target population is divided into 1. the vulnerable multimorbid ill population, 2. people with one chronic disease and 3. people at risk for chronic disease and the healthy population. The six central themes for developing actions are 1. care programs, 2. care coordination, 3. medication, 4. caring neighborhoods, 5. assisted living and 6. health promotion (20). The project plan therefore guides the local consortia in developing actions for integrated care however providing enough flexibility allowing stakeholders to experiment in finding the right local solutions.

SELECTION OF TRANSITIONAL CARE ACTIONS AND INTERVIEWEES

We selected actions with presumed impact on transitional care. Therefore, the researchers reviewed available information online (e.g. project plan) and developed a list of actions from ZZL that focused on improving transitional care pathways. This list was discussed with a project coordinator from ZZL and was subsequently narrowed down to four actions, which formed our final selection: 1. Care program heart failure to improve care for chronic heart failure patients by implementing four-guideline recommended interventions; 2. Intermediate care center to reduce the burden on hospital during the first wave of COVID-19; 3. Neighborhood teams to create close networks of local health care providers; and 4. Medication reconciliation envelope to provide a link between the hospital and the community pharmacy (see Table 1).

For each action we first conducted an interview with the designated project coordinator from ZZL. Project coordinators were the first point of contact as they were involved in the development and implementation of each action and thus particularly knowledgeable. We also asked project coordinators to suggest relevant stakeholders from the field. We ensured to include stakeholders that were directly involved in the action. We continued to use the snowball method for additional selection of potential participants. Stakeholders were invited by e-mail including information about the study and an informed consent form. In total 24 invitations were send out of which 13 participants accepted the invitation.

DATA COLLECTION

Semi-structured interviews were carried out by two student researchers. The interviews were

CHAPTER 5

conducted in the timeframe from February until April 2021 via video conferencing software in order to oblige with the COVID-19 regulations. Two interview guides were developed: one for interviewing the project coordinators and one for interviewing other stakeholders. Questions were developed by reviewing relevant literature and in discussion with the research team. After each interview, the interview guide was revised and adjustments to the questions were made if deemed necessary. The main topics of the interview guide were: description of the action, role of the involved stakeholders, stakeholders perception on transitional care, sustainability of four actions, and lessons learned for the future of transitional care within ZZL. The complete interview guide can be found in Appendix 5A. Also, we conducted a document analysis to collect additional data on the action description and searched for relevant documents using websites, actions plans, published papers and government reports. Included documents had to discuss at least one of the four selected actions.

Thirteen interviews were conducted in Dutch and one in English. The interviews lasted 55 minutes on average. The interviews were recorded and transcribed ad verbatim. English translations of transcripts were conducted by two student researchers and checked by a third researcher (ML). All transcripts were then imported to NVivo for data analysis.

DATA ANALYSIS

First, thematic analysis with an inductive approach following the six steps as described by Braun & Clarke 2006 was used to organize, describe and analyze the qualitative data in detail (21). Second, collected data from the document analysis was added to the interview data in order to conduct the action description. This allowed for comparison of various sources in order to reach corroboration. Interview data of each action was analyzed separate for the action description. In order to analyze the experiences and perceptions of participants on transitional care, we analyzed interview data of all four actions combined. NVivo software was used to support the analysis process.

For the thematic analysis, first the familiarization of data was conducted by two researchers (ML, AF) reading and re-reading through the full transcripts to get a first understanding about the data as well as ideas for coding. Second, initial codes were generated from the data by one researcher (ML) and relevant data were collected for each code by coding all transcript interviews. A second researcher (AF) co-coded simultaneously all interviews independently. The researchers were meeting each other at three different time moments to discuss meaning of codes and resolve disagreements before, during and after coding. As a third step, themes were searched by listing all codes and starting the process to sort and combine codes leading to first ideas of themes and sub-themes. For the fourth step, themes were jointly reviewed by two researchers (ML, AF) to further refine themes and their meaning. A second revision with a third researcher (GG) was conducted and resulted in a

thematic 'map'. Defining, naming and describing themes separately and collectively was conducted in step five in discussion with three researchers (ML, AF, GG) and was discussed with the whole research team. Lastly, the final report was produced with supporting quotes from the data.

The Ethics Committee UZ Leuven/KU Leuven approved the study (registration number: MP017284) and the interviewees provided written informed consent.

RESULTS

INTERVIEW PARTICIPANT CHARACTERISTICS

Fourteen interviews were conducted with thirteen stakeholders as one stakeholder was interviewed twice on two actions. Among the five project coordinators that were interviewed, there were three pharmacists, one physiotherapist and one general practitioner. The eight other stakeholders involved in the transitional care actions were two general practitioners; one home care nurse, one cardiac nurse, one cardiologist, one physiotherapist and one policy advisor for welfare & care (see Appendix 5B for an overview of interview candidates across the four actions).

TRANSITIONAL CARE ACTIONS

The thematic analysis and document analysis to describe four transitional care actions resulted in an item list that is presented for each of the four actions in Table 1. It includes the objective, transitional care focus, patient target group, main HCPs included, key components of the action, synergies, implementation status, context information. Appendix 5C lists all items and their meaning.

Item List	Action 1: Intermediate Care Center	Action 2: Neighborhood teams	Action 3: Medication envelope action	Action 4: Care program chronic heart failure
Objective	To reduce the burden on hospitals during the first wave of COVID-19 and therefore to provide an immediate solution during the crisis.	To create close networks of local health care professionals at neighborhood level in order to provide integrated care for patients and to detect vulnerable groups early on. The action aims to strengthen and structure primary care and improve cooperation between primary care.	To support the discharge process from the hospital by providing information on medication to the community pharmacist therefore allowing pharmacists to provide medication reconciliation to discharged patients.	To improve care for heart failure patients by implementing four guideline-recommended disease management interventions.
Transitional care focus	To provide support during two care transitions: (The transition from hospital to intermediate care center and, 2) transition from intermediate care center to the community.	To prevent unnecessary care transitions from the community setting to the hospital by connecting the neighborhood and providing early integrated care for vulnerable patients including coordination of care.	To facilitate the information transfer on medications during the care transition from hospital to the community pharmacist.	To avoid unnecessary care transitions to the hospital and improve necessary transitions from the hospital to the community.
Patient target group	Vulnerable patients who were medically able to leave the hospital and who (often for social reasons) couldn't go home yet - both patients with a COVID-19 infection or without an infection.	Each neighborhood team decided on All patients discharged from their own target population, however participating hospital depart the focus was especially on the vulnerable or multimorbid population.	All patients discharged from participating hospital departments	Heart failure patients or patients with a risk for heart failure living in the community or being admitted at the cardiology ward.
Main HCPs involved	Coordinating pharmacist, community pharmacist, home care nurse, head nurse, social worker, GP, specialist, psychologist.	Local health care professionals from the same neighborhood including: GP, pharmacist, physiotherapist, home care nurse, psychologist, tabaccologist, dietician, social worker	Nurse at discharge and community pharmacist.	Heart failure educator, GP, heart failure nurse at the hospital ward, cardiologist, home pharmacist.

Table 1. Description of actions

Item List	Action 1: Intermediate Care Center	Action 2: Neighborhood teams	Action 3: Medication envelope action	Action 4: Care program chronic heart failure
Key components	 Coordination of medical, pharmaceutical and social care to arrange a seamless transition between the hospital and the intermediate care center and the intermediate care center and the home setting The pharmacist and home nurse were contacted and informed and if necessary a follow-up appointment was scheduled for the patient. 	Each neighborhood team developed specific aims and approaches for their patients that resulted in diverse projects such as: group sessions to provide information on different topics (e.g. loneliness, healthy habits, positive health) coaching sessions within the neighborhood such as walking moments or smoking cessation campaign for COPD patients programs.	 The patient receives an envelope from the nurse at discharge containing necessary documents to perform a medication reconciliation: medication scheme from the hospital, medication prescriptions at discharge and a registration form. The envelope is addressed to the is addressed to the community pharmacist. To keep track of conducted medication reconciliations, community pharmacist are requested to scan the code of the envelope and to fill in a registration form 	 The following four interventions were implemented: To improve the first-line diagnostics by reimbursing the natriuretic peptides test (NT- proBNP) for GP's in Leuven which allows diagnosing heart failure To implement automated diagnostic and qualitative audits for heart failure in primary care settings that help to safeguard high quality care for patients To provide a heart failure education session that focus on self-care management for high- risk patients. GP's or HCBs at the hospital can contact trained nurses to provide an education session to their patients
				 To improve the discharge moment for heart failure patients at the hospital by implementing a structured discharge protocol including: a checklist for high risk pincluding: a checklist for high risk predients, telephone contact with the patient GP to plan follow-up appointment and heart failure education session 14 days after discharge

Item List	Action 1: Intermediate Care Center	Action 2: Neighborhood teams	Action 3: Medication envelope action	Action 4: Care program chronic heart failure
Synergies	 Siilo application: a secure medical messenger for HCPs Medication envelope (Action 3). 	 Sillo application. NexusHealthPro software: NexusHealthPro software: NexusHealthPro software access to several health care professional groups including physiotherapists, nurses, pharmacists, dentists and midwives. The software allows to consult hospital files of patients. 		Medication envelope (Action 3).
Implementation status	Implementation status The action was implemented during the first wave of COVID to provide an immediate solution during the crisis and was stopped afterwards.	8 neighborhood teams are running in the region of ZZL and each team roughly covers a population of 5000- 8000 inhabitants.	The action is implemented within various departments of three hospitals in the region/close to the region of ZZL.	The action is implemented at various hospitals (cardiology department) and at the community (homecare and primary care).
Context information	Coordinating pharmacist organized the medication follow-up from admission until discharge. The role of the coordinating pharmacist was tested first in Belgium within the intermediate care center.	The neighborhood teams were structured based on 'natural' networks in the community. A division of 24 small neighborhoods of 4000 inhabitants exist in Leuven and has been used to start the networking of neighborhoods within ZZL (22).	The action fills a current gap to digitally share information on medications at discharge.	The four interventions have been tested previously in the Belgium setting (23).

Table 1. Continued.

EXPERIENCES AND PERCEPTIONS

Thematic analysis to explore the stakeholders experiences on transitional care resulted into the five themes: 1. Involvement of HCPs in decision making, 2, Improved community feeling – reduced barriers of communication, 3. Supporting tools and learning opportunities for HCPs, 4. Transitional care for patients in practice, 5. Key challenges: coordination, resources, financial & political support. Each theme is described separately in the next sections.

THEME 1: INVOLVEMENT OF HCP IN DECISION-MAKING

HCPs appreciated being involved early on in shaping actions and welcomed the bottomup approach from ZZL allowing them to propose own ideas for initiatives. Stakeholders felt motivated to be involved in all four actions that can benefit the care for their patients and bring themselves closer to local health care professionals from primary care and secondary care.

"The approach that we are not going to come to you with a finished programme, but you are going to have a say in that programme. They really like that, they feel acknowledged..." (Project Coordinator)

"And a lot of specific questions, a lot of input from primary care providers. That was greatly appreciated, because it was actually the first time that the primary care providers had the feeling that they could directly take part in this process." (Project Coordinator)

Actions were considered as a potential leverage for change and positive experiences were shared with colleagues, who then convinced other HCPs to join. Moreover, HCPs appreciated that their input was taken into account while shaping actions and that for example a new role for coordinating pharmacists could be tested at the intermediate care center (Action 1). These positive changes triggered discussions on how health care professionals see their role and increased their confidence to be more assertive about being involved and voicing their opinion.

"The neighborhood teams also have a very strong pull effect and that is because they are set up by the care providers themselves." (Project Coordinator)

"They're stronger, more assertive, and they now also asked to join the vaccination centers. That also makes communication more easy. They have an opinion [...] "(Project Coordinator)

THEME 2: IMPROVED COMMUNITY FEELING - REDUCED BARRIERS FOR COMMUNICATION

Across all actions, stakeholders reported that the threshold for communication had reduced

and they noticed a significant improvement in the overall team spirit. HCPs experienced more direct and straightforward communication with each other.

By doing so yes we have got to know each other, haven't we? So now we know each other and yes we have a much lower threshold to send a Siilo message" (HCP 1)

Also, HCPs perceived an increased openness for providing interdisciplinary care. Stakeholders reported that the actions change their way of thinking about interdisciplinary care and that patients with chronic care needs should not be treated alone.

"There is much more openness to seeking and giving interdisciplinary advice "(HCP 1)

"And I think that this project has made me think more about the fact that chronic care, the people who need chronic care, should not be tackled alone. Um, that I should take steps towards other people and say, we're going to tackle this more together." (HCP 2)

In addition, the actions allowed HCP to get to know each other and to build up trust over time within and between primary care and secondary care. This was often highlighted in Action 1 where care decisions were shared with all involved HCPs using the Siilo application. A feeling of professional joy and being proud of what was achieved together was reported by one stakeholder.

"That professional joy that yes, everyone was really like that. I made my contribution here to a greater whole, even more than usual, you could really see that. Every little step that was taken by a particular care provider was shared in the group and everyone was like oh wow yes ok, we have to build on this. That was fantastic actually." (HCP 3)

THEME 3: SUPPORTING TOOLS AND LEARNING OPPORTUNITIES FOR HCPS

HCPs felt the necessity to have modern secure tools. The communication application Siilo was appreciated for small care teams allowing to solve misunderstandings, to make adjustments to the care plan or to receive information on medication in a quick and uncomplicated way.

"I am convinced that Siilo can work on a small scale in defined patient groups" (HCP 3)

Also, the paper envelope for conducting medication reconciliation was acknowledged by HCP as a tool for information transfer between hospital and the community pharmacy. Although many participants experienced these tools as useful, it was often stated that the current tools are seen as a temporary solution and a digital integrated patient record with messaging function is needed in the long-term.

"The problem lies in the fact that we still have to work on paper. Both for the home care nurses and for the pharmacists as well. So in Belgium there is no safe, for the time being, no platform where we can work together in the same module to follow up on the treatment [...]"(Project Coordinator)

HCP's valued the learning opportunities that were provided within the actions. Stakeholders reported that the actions increased their awareness for the importance to conduct high-quality follow-up care (e.g. medication reconciliation guideline in action 3).

"I actually think that this is an added value because the envelope campaign made us aware that we can get some extra information from the patient that we sometimes don't think of ourselves" (HCP 4)

THEME 4: TRANSITIONAL CARE FOR PATIENTS IN PRACTICE

Across all actions, stakeholders perceived that the actions offered support to guide their patients through complex treatments. Stakeholders experienced that the transitional care actions led to increased consultations with patients allowing HCPs for a better understanding of patient needs which ultimately resulted in delivering a more focused care approach.

"But I think there is a lot more consultation and we can approach the care in a much more focused way and guide the people better" (HCP 5)

"Because you have a lot more, details about the person themselves. So I can really look more at what applies to this person. Not the generalized rattling off of questions to tell them. But with more information, you can indeed focus much more on the patient in front of you, in his individual context." (HCP 4)

In addition, stakeholders shared feedback from primary care professionals highlighting improved knowledge and awareness of patients in better understanding early signs and symptoms of their disease. This was observed for patients who received a heart failure session during follow-up (Action 4), leading to improved awareness and increased independence.

"We have already heard from the home care nurses that the patients really do realize what they have to look out for. They are also becoming partly more independent, because they know that they have to start moving." (HCP 5)

Moreover, stakeholders highlighted that the actions within the neighborhood allowed patients to connect with their community by joining activities organized at the local level (e.g. Action 2 – walking/exercising moments, group sessions).

"I think that at the moment, the impact of [name of neighborhood team] is certainly not that big on the things that you mention now. Erm, but it will rather have an impact on, erm, community work that brings people together more, such as the walks that are organized, the evenings that are organized around a particular theme" (HCP 6)

Although the actions offered additional support for their patients, stakeholders overall felt that actions provided limited follow-up after transition and provided limited possibilities for patient involvement. Action 3 and 4 provided a one-time moment for follow-up which was perceived as too short by participants to ensure continuity of care.

"But after that, it is up to the GP and the informal carer to follow up properly and there is no actual follow-up. There is no long-term follow-up within our project [...]" (Project Coordinator)

Also, many stakeholders perceived patient involvement as difficult in practice, partly due to technological challenges and partly due to their own reluctance in being involved in their own care. Difficulties were mentioned in particular for older vulnerable patients as they were asked to login to a patient platform which was perhaps not the right solution for this population group and did not provide flexible solutions. Additionally, stakeholders observed that patients did not agree to be included in the actions likely due to limited reach and awareness of ZZL or the COVID-19 situation that led to many cancellations.

"Um, but if we look at the chronic care population that I see, it's mainly the 70-80 plus, yes. The whole internet thing, online login, all that stuff, is perhaps not so applicable to them. In 20 years' time, it might be completely different, but at the moment it's perhaps not perfectly adapted to the needs of the patient." (HCP 2)

"It's not always possible to really involve the patient according to the books, simply because sometimes there is a reluctance on the part of the patients and [...]. I don't mean to say that you can't or you can't do something about that, because that takes a lot of time, a lot of guidance" (HCP 7)

THEME 5: KEY CHALLENGES – COMMUNICATION, RESOURCES, FINANCIAL & POLITICAL SUPPORT

A lack of concrete communication and coordination was reported, especially in the beginning of the pilot project as involved HCPs needed to take over new roles and responsibilities that often required to learn new skills. Therefore, bringing disciplines together and deciding on a common method for coordination was perceived as difficult.

"because I had the impression that everyone was a bit on their own, all the disciplines were a bit on their own, but not enough working together" (Project Coordinator)

Additionally, stakeholders reported that a clear protocol for each action was missing, as a general project plan already existed that was used for all actions combined. This general project plan was set up in the development phase of the pilot project, however it was reported that the plan was modified in the meantime and not followed exactly as outlined. Stakeholders reported that the missing guidance created difficulties for implementation and decisions on ownership.

"But because it didn't have a very clear protocol of 'yes, we're going to do it like this and like that'. In practice, it was a bit difficult to decide who should take the initiative for which patients" (HCP 2)

Moreover, the COVID-19 measurements impeded smooth communication and increased the effort needed to have regular meetings. For example, COVID-19 made it hard to onboard newly introduced roles such as coordinating pharmacists as all meetings were virtual and hence other HCPs were unaware of the existence of such roles.

"So yes, that was what was difficult because of COVID, that not everyone's role was equally clear [...]" (Project Coordinator)

In addition to communication challenges, several structural challenges like increased workload of administrational tasks and shortness of staff was often mentioned. Also, due to the bottom-up approach from ZZL, HCPs active participation required extra energy and time, particularly in the early phase of the actions.

"But then you also have the problem of resources. Neighborhood teams,

that's not really backed up by structural resources. The future of Zorgzaam Leuven itself is not yet very clear." (Policy Maker)

"I think for some colleagues it's an extra burden at the moment because it's still under construction, so it takes energy to think about how we're going to change that and what does that mean for my own practice" (HCP 1)

Stakeholders also mentioned the lack of financial and political support. Financial resources were limited and there was limited regulatory leeway for projects to experiment with new care approaches. Moreover, stakeholders expressed that the lack of long-term perspective felt demotivating as no clear future and roadmap of ZZL was communicated making future planning of the various actions difficult.

"Especially I think from the government and from, yes from government agencies more support I think or more also more shorter possibilities to tune things and roll them out. In such a way that, yes, we can also give a perspective to the people we are working with. I have the feeling that we are not able to offer enough perspective [...]" (Project Coordinator)

DISCUSSION

In this qualitative case study we described four transitional care actions that are part of the pilot project ZZL and explored experiences and perceptions of involved stakeholders on transitional care in practice. Across the four actions, stakeholders valued active involvement and increased collaboration. Stakeholders brought forward that the co-designing approach of ZZL increased active participation as received input was taken into consideration when shaping actions. Additionally, actions encouraged interdisciplinary collaboration, which in turn reduced barriers for communication and created a community feeling that benefitted professionals and patients. However, stakeholders emphasized key challenges, such as technological infrastructure, clear coordination, financial support and political support that impeded sustainable implemented to provide (temporary) solutions to address these key challenges.

Our results highlight that HCPs intrinsic motivation and feeling connected with other HCPs facilitated the implementation of integrated care. In the literature this has been emphasized by the self-determination theory. The self-determination theory states that the satisfaction of the three basic psychological needs (autonomy, competence and relatedness) facilitates and sustains high quality motivation (24). The need for autonomy refers to experience

choice in the own role. Competence concerns the need to feel successful within the own role and to have the ability to develop necessary skills. The need of relatedness refers to the feeling of belonging and acceptance within a group (25). Evidence has shown that applying the self-determination theory in the work context is linked to a wide range of positive outcomes for employees, such as developing autonomous work motivation (25). Moreover, autonomous motivation is the most sustainable type of motivation, predicting high-quality performance and positive outcomes related to well-being (24, 26). Our findings suggest that the pilot project ZZL created conditions that supported the three needs for self-determination. This was perceived as positive by HCPs and triggered their autonomous work motivation. In practice these needs were supported for involved HCPs namely by encouraging involvement in decision making, providing tools for communication or stimulating regular communication to build up trust.

The early involvement of stakeholders in the co-creation process encouraged stakeholders to share experience and increased collaboration which allowed to build up trust. Stakeholders had to get to know each other and learn new roles, such as designing and implementing actions. Therefore, the pilot phase of the project was used as a testing phase to experiment and to learn from mistakes. Actions that are undertaken in the pilot phase are first planned at small scale which is supposed to make the development and implementation of new actions less frightening and allows for correction of mistakes along the way (10). The study of Fakha et al. (2022), a second study that analyzed the interview data on ZZL and focused on implementation research, supports the importance of involvement (27). The study highlights engagement as the main implementation strategy used to back the implementation of the four transitional care actions. Also, motivated key individuals were identified as a crucial facilitating implementation factor. Using engaging activities allowed individuals to connect and build partnerships that strengthened motivation and commitment of HCPs, which was positively perceived by stakeholders as highlighted in our study. This is in line with further research stating that the approach to stimulate local dynamics of flexibility and experimentation is crucial, however that the successfulness depends on committed leaders and stakeholders (7).

While ZZL created several previously discussed motivating conditions for HCPs to bring change towards integrated care, various key challenges were identified that impeded further development, therefore leading to demotivation of HCPs. A key challenge that was identified in this study is the lack of integrated patient files in Belgium (26, 28). Various digital systems were used by stakeholders and a lack of interoperability of these systems was stated, leading actions to develop temporary solutions to facilitate communication between HCPs. This finding is in line with other analyses on the Belgium health care context identifying the lack of shared electronic files as a major weakness hindering the development of integrated care in Belgium (13, 29). Moreover, this resonates with findings of

a recent systematic review indicating that none of the 15 reviewed transitional care models (TCMs) used electronic health records that allowed sharing of information between health care settings, stressing the need to create stronger digital links between settings (30). Also the importance of using technologies within TCMs is underpinned by the SELFIE framework (Sustainable integrated chronic care models for multi-morbidity: delivery, financing, and performance), including shared information systems as a main component to facilitate care coordination of TCMs (31).

Furthermore, research shows that the lack of electronic health records tracking the patients decision-making process creates a barrier for patient engagement (29). This is in line with our study findings as across all four actions patient engagement was described as limited. Stakeholders reported being reluctant towards patient engagement due to fear of increased workload and also described signs of reluctance of their patients. The findings are in line with a recent review highlighting the three main barriers for patient engagement as patient unwillingness, HCPs unwillingness and inadequate infrastructure (32). The review concludes that attention should be paid to these barriers by creating a promoting environment ensuring sufficient resources and infrastructure and additionally establishing educational programs for patients and professionals (32).

STUDY LIMITATIONS

This has been the first study providing an initial understanding of what concrete actions have been locally implemented within one of the 12 Integreo projects in Belgium. These government initiated pilot projects are a huge undertaking, but have not been described in detail. To describe the four selected transitional care actions we used the two methods of document analysis and thematic analysis to reach corroboration and therefore increase trustworthiness.

The study findings also need to be interpreted in the light of some methodological considerations. In terms of the selection of stakeholders to be interviewed, only 13 of the potential 24 stakeholders agreed to participate in the study. They might have been more positive about the actions then those who were not interviewed. Also, the number of stakeholders interviewed per action varied with one action where we could only interview two stakeholders. We assume that this impact was limited as we did not analyze perceptions for each individual action separately, but made an overall analyses at the level of the four actions. Further, the total number of 14 interviews across 4 actions might lack representation, yet we collected meaningful data by interviewing project coordinators who are particular knowledgeable and who helped us to select key stakeholders that provided valuable insights. Lastly, we did not explore the views or experiences of patients involved in the transitional care actions and could therefore not critically discuss differences in experiences and perceptions between patients and HCPs. We strongly recommend to

focus on the patient perspective in future research activities related to the evaluation of the Integreo projects. Moreover, we recommend a thorough evaluation of the effectiveness of ZZL and the other Integreo projects taking into account the quadruple aim objectives, which was defined as the ultimate goal of Integreo. Additionally, we suggest to continue to conduct process evaluations to better understand how the outcomes were achieved in order to build on lessons learned and adapt actions. We suggest to regularly evaluate the development of pilot projects as they are dynamic, change over time and are contextspecific.

CONCLUSION

To conclude, our findings indicate that the integrated care project ZZL created conditions to promote autonomous work motivation for HCPs. The project encouraged the development of multidisciplinary care networks at the local level, which allowed professionals to connect and create partnerships. Also, the project offered (temporary) solutions to address pressing problems for continuity of care. Yet, key challenges were identified that impeded long-term planning for integrated care within the project region. Additionally, our analysis highlights how integrated care and transitional care are interconnected, by describing how the four actions linked to the integrated care project addressed various transitional care components to achieve continuity of care for patients.

REFERENCES

- World Health Organization. Transitions of care. Geneva: World Health Organization; 2016. https://apps.who.int/ iris/handle/10665/252272.
- Laugaland K, Aase K, Barach P. Addressing risk factors for transitional care of the elderly Literature review. 2011.
- World Health Organization. Integrated health services what and why? 2008. https://www.who.int/healthsystems/ service_delivery_techbrief1.pdf.
- World Health Organization. Technical series on primary health care Integrating health services. 2018. https:// www.who.int/docs/default-source/primary-health-care-conference/linkages.pdf.
- Hogan C, Barry M, Burke M, Joyce P. Healthcare professionals' experiences of the implementation of integrated care pathways. Int J Health Care Qual Assur. 2011;24:334-47.
- Jolanki O, Liina-Kaisa T, Sinervo T. Professionals' views on integrated care: Managing Community Care. Int J Integr Care. 2017;25(4):247-55.
- Grudniewicz A, Tenbensel T, Evans JM, Steele Gray C, Baker GR, Wodchis WP. 'Complexity-compatible' policy for integrated care? Lessons from the implementation of Ontario's Health Links. Soc Sci Med. 2018;198:95-102.
- 8. Bryan S, Holmes S, Postlethwaite D, Carty N. The role of integrated care pathways in improving the patient experience. Prof Nurse. 2002;18(2):77-9.
- Integreo. GEÏNTEGREERDE ZORG VOOR EEN BETERE GEZONDHEID. https://www.integreo.be/nl. Accessed 13 April 2022.
- 10. Winter MD. Reshaping health care governance using pilot projects as public policy implementation instruments. International Review of Public Policy. 2020;2(3):317-41.
- 11. Integreo. Zorgzaam leuven (Regio Groot-Leuven). https://www.integreo.be/nl/uw-omgeving/zorgzaam-leuvenregio-groot-leuven. Accessed 11 May 2022.
- 12. Crowe S, Cresswell K, Robertson A, Huby G, Avery A, Sheikh A. The case study approach. BMC Med Res Methodol. 2011;11(1):100.
- Danhieux K, Martens M, Colman E, Wouters E, Remmen R, Van Olmen J, et al. What Makes Integration of Chronic Care so Difficult? A Macro-Level Analysis of Barriers and Facilitators in Belgium. Int J Integr Care. 2021;21(4).
- Goderis G, Colman E, Irusta LA, Van Hecke A, Pétré B, Devroey D, et al. Evaluating Large-Scale Integrated Care Projects: The Development of a Protocol for a Mixed Methods Realist Evaluation Study in Belgium. Int J Integr Care. 2020;20(3).
- Bodenheimer T, Sinsky C. From triple to quadruple aim: care of the patient requires care of the provider. Ann Fam Med. 2014;12(6):573-6.
- 16. ZorgzaamLeuven. Wie zijn wij?. https://www.zorgzaamleuven.be/wie-zijn-wij. Accessed 19 June 2022.
- 17. Leuven S. Stad Leuven in cijfers Bevolkingscijfer per maand 2021. https://leuven.incijfers.be/dashboard/ dashboard/demografie. Accessed 8 November 2021.
- Ranking S. 2021 Academic Ranking of World Universities. 2021. https://www.shanghairanking.com/rankings/ arwu/2021. Accessed 8 November 2021.
- 19. Statistic Flanders. Population by age and gender. https://www.vlaanderen.be/en/statistics-flanders/population. Accessed 8 November 2021.
- 20. ZorgzaamLeuven. Onze acties. https://www.zorgzaamleuven.be/onze-acties-1. Accessed 19 June 2022.
- 21. Braun V, Clarke V. Using thematic analysis in psychology. Qual. Res. Psychol. 2006;3:77-101.
- Heyerman J, Pint G, Peters A, Van de Putte M, Van Pottelbergh G. Zorgzaam Leuven: binnen de nationale projecten rond chronische zorg een succesvolle oefening in buurtgerichte zorgreorganisatie Tijdschrift voor Geneeskunde en Gezondheidszorg. 2021;3.
- Raat W, Smeets M, Van Pottelbergh G, Van De Putte M, Janssens S, Vaes B. Implementing standards of care for heart failure patients in general practice – the IMPACT-B study protocol. Acta Cardiologica. 2021;76(5):486-93.
- 24. Deci EL, Ryan RM. Facilitating optimal motivation and psychological well-being across life's domains. Canadian Psychology/Psychologie canadienne. 2008;49(1):14-23.
- Forner VW, Jones M, Berry Y, Eidenfalk J. Motivating workers: how leaders apply self-determination theory in organizations. Organ Manag J. 2020;18(2):76-94.
- Dorssen-Boog P, Vuuren T, Jong JP, Veld M. Facilitating health care workers' self-determination: The impact of a self-leadership intervention on work engagement, health, and performance. J Occup Organ Psychol. 2021;94(2):259-81.
- 27. Fakha A, Leithaus M, de Boer B, van Achterberg T, Hamers JP, Verbeek H. Implementing four transitional care interventions for older adults: a retrospective collective case study. Gerontologist. 2022.
- Busetto L, Luijkx KG, Elissen AMJ, Vrijhoef HJM. Context, mechanisms and outcomes of integrated care for diabetes mellitus type 2: a systematic review. BMC Health Serv Res. 2016;16(1):18.
- Van Durme T, Macq J, Anthierens S, Symons L, Schmitz O, Paulus D, et al. Stakeholders' perception on the organization of chronic care: a SWOT analysis to draft avenues for health care reforms. BMC Health Serv Res.

2014;14(1):179.

- Leithaus M, Beaulen A, De Vries E, Goderis G, Flamaing J, Verbeek H, et al. Integrated care components in transitional care models from hospital to home for frail older adults: a systematic review. Int J Integr Care. 2022;22(0):28.
- 31. Leijten FRM, Struckmann V, Van Ginneken E, Czypionka T, Kraus M, Reiss M, et al. The SELFIE framework for integrated care for multi-morbidity: development and description. Health Policy. 2018;122(1):12-22.
- 32. Chegini Z, Arab-Zozani M, Shariful Islam SM, Tobiano G, Abbasgholizadeh Rahimi S. Barriers and facilitators to patient engagement in patient safety from patients and healthcare professionals' perspectives: a systematic review and meta-synthesis. Nursing Forum. 2021;56:1-12.

APPENDICES

Appendix 5A. Semi-structured interview guideAppendix 5B. Characteristics of intervieweesAppendix 5C. Action description item list

Appendix 5A. Semi-structured interview guide

(Note – the underlined questions are the ones to consider for this study as they refer to descriptions and exploring experiences perceptions on transitional care. The other questions were used for a second study investigating the implementation of the four actions [27])

Questions – Project coordinator

- 1. <u>Can you shortly describe the action and your role in those actions?</u>
- 2. Which stakeholders and healthcare professionals were involved in those actions?
- 3. What did you do to encourage, motivate and engage healthcare professionals to commit to using these actions? Did you assign any champions, leaders, persons with a facilitator role, or any other assigned roles to help spread those actions to the healthcare providers?
- 4. <u>Now we will focus on the aspect transition of care for elderly. What does transitional care mean for you?</u>
- 5. <u>Can you shortly describe the impact of the action on the transition of care for elderly people?</u>
 - a. <u>Communication</u>
 - b. <u>Shared-decision making/ patient involvement/ informal caregiver involvement</u>
 - c. <u>Person-centered care</u>
 - d. <u>Medication reconciliation</u>
 - e. <u>Continuity of care at home (organization of follow-up care)</u>
- 6. Can you shortly describe the impact of the action on the healthcare professionals?
- 7. <u>Can you shortly describe the impact of the action on the healthcare system?</u>
- 8. <u>To what extent are those actions fully integrated? What stage are they at?</u>
 - a. <u>Status of implementation</u>
 - b. Does the current state of implementation of those actions meet your expectations?
 - c. <u>Is the action still carried out as planned?</u>
 - i. How has the action been affected in the context of COVID-19
 - ii. Sustainability of the project
- 9. Did you develop a project plan to implement and roll out those actions to the various healthcare providers? If yes, can you briefly describe the plan?
 - a. Did you perform any activities or use specific strategies to implement those actions? (e.g.: education and training, champions, mandate change)
- 10. <u>Did you already receive some feedback from primary and secondary healthcare</u> providers? If yes, what did you learn about it?
- 11. Adoption (= intention to try to use the intervention)
 - a. How would you assess (or what would you say) on the adoption level of these actions by healthcare professionals? This means the intention to use the intervention by the healthcare professionals.

- b. What are the barriers/obstacles/challenges in the implementation of those actions? And were there any facilitators to enable a better and successful implementation of those actions?
- 12. What is needed to continue these actions?
- 13. <u>How can this action be expanded?</u>
- 14. Which lessons could be learned from this action regarding the future?

Completing the interview

Would you like to add something to this interview? Thank you for your participation.

Questions – Healthcare professional

- <u>Can you shortly describe the action and your role in those action?</u>
 <u>How do you experience this role?</u>
- 2. <u>Which other persons are involved?</u>
 - a. <u>How did you experience the collaboration and the communication? Is everyone</u> <u>aware of his/her responsibilities?</u>
- 3. <u>Now we will focus on the aspect transition of care for elderly. What does transitional care mean for you?</u>
- 4. <u>Can you shortly describe the impact of the action on the transition of care for elderly</u> people?
 - a. <u>Communication</u>
 - b. <u>Shared-decision making/ patient involvement/ informal caregiver involvement</u>
 - c. <u>Person-centered care</u>
 - d. <u>Medication reconciliation</u>
 - e. Continuity of care at home (organization of follow-up care)
- 5. Can you shortly describe the impact of the action on the healthcare professionals?
- 6. <u>Can you shortly describe the impact of the action on the healthcare system?</u>
- 7. Acceptability (perceived views that an intervention is agreeable, satisfactory, credible, and comfortable): to what extent do you think those actions are satisfactory and advantageous to the older persons with chronic disease and requiring care transitions between hospital & home?
- 8. Appropriateness (perceived compatibility of the intervention with needs & practices of a setting or population): to what extent do you think these actions address/meet the care needs of older persons with chronic disease and requiring care transitions?
- 9. <u>Is the action still performed as planned?</u>
 - a. <u>Status of implementation</u>
 - b. How has the action been affected in the context of COVID-19?
- 10. <u>Sustainability (extent to which an intervention is routinized or maintained within an</u> organization): How would you ensure that these actions become a routine part (used

regularly) of the daily work of the healthcare professionals within your team/in your organization?

- 11. Experiences and expectations:
 - a. <u>How did you experience those actions?</u>
 - b. <u>Is the action meeting your expectation?</u> Can you explain the reasons as to why it is or isn't meeting your expectations?
 - According to you, do you think the implementation of the action is successful?
 Can you explain the reasons as to why the implementation is or isn't successful?
- 12. What are the barriers/obstacles/challenges in the implementation of those actions? And were there any facilitators to enable a better and successful implementation of those actions?
- 13. Which lessons could be learned from this action regarding the future?

Completing the interview

Would you like to add something to this interview? Thank you for your participation.

Appendix 5B. Characteristics of interviewees

Number of interviewees	Coordinator or stakeholder	Profession of interviewees	Action number
1	Coordinator	Pharmacist	1
2	Coordinator	Pharmacist	1
3	Coordinator	Pharmacist	1
4	Stakeholder	Home nurse	1
5	Coordinator	Physiotherapist	2
6	Stakeholder	GP	2
7	Stakeholder	GP	2
8	Stakeholder	Physiotherapist	2
9	Stakeholder	Policy advisor	2
2	Coordinator	Pharmacist	3
10	Stakeholder	Pharmacist	3
11	Coordinator	GP	4
12	Stakeholder	Cardiologist	4
13	Stakeholder	Nurse	4

Table. Characteristics of interviewees per action

Appendix 5C. Action description item list

Iten	1 description	
1.	Objective	Action purpose
2.	Transitional care focus	How actions improve transitional care for patients and across what care settings
3.	Patient target group	Patient groups that are targeted with the actions
4.	Main HCPs involved	List of all HCPs involved in delivering the actions
5.	Key components	Essential pillars that define the actions
6.	Synergies	Possible links with other actions or with existing care
7.	Implementation status	Implementation status at the moment of data collection: actions ongoing or stopped and locations of implementation
8.	Context information	Unique local aspects that need to be known to understand the actions

Table. Action description item list



CHAPTER

Systematic Development of a Set of Implementation Strategies for Transitional Care Innovations in Long-Term Care

Fakha A, de Boer B, Hamers J, Verbeek H, van Achterberg T. Systematic development of a set of implementation strategies for transitional care innovations in long-term care. (submitted, under review).

ABSTRACT

BACKGROUND

Numerous Transitional Care Innovations (TCIs) are being developed and implemented to optimize care continuity for older persons when transferring between multiple care settings, help meet their care needs, and ultimately improve their quality of life. Although the implementation of TCIs is influenced by contextual factors, the use of effective implementation strategies is largely lacking. Thus, to improve the implementation of TCIs, we systematically developed a set of strategies selected to address the influencing factors.

METHODS

As part of the TRANS-SENIOR research network, a stepwise approach following Implementation Mapping was applied to select implementation strategies. Building on the findings of previous studies, existing TCIs and factors influencing their implementation were identified. A combination of four taxonomies and overviews of change methods as well as relevant evidence on their effectiveness were used to select the implementation strategies targeting each of the relevant factors. Subsequently, individual consultations with scientific experts were performed for further validation of the process of mapping strategies to implementation factors and for capturing alternative ideas on relevant implementation strategies.

RESULTS

20 TCIs were identified and 12 influencing factors (mapped to the Consolidated Framework for Implementation Research) were designated as priority factors to be addressed with implementation strategies. A total of 40 strategies were selected. The majority of these target factors at the organizational level, e.g., by using structural redesign, public commitment, changing staffing models, conducting local consensus discussions, and organizational diagnosis and feedback. Strategies at the level of individuals included active learning, belief selection, and guided practice. Each strategy was operationalized into practical applications.

CONCLUSIONS

This project developed theory and evidence-based implementation strategies to address the influencing factors and enhance the implementation of TCIs in daily practice settings. Such work is critical to advance the use of implementation science methods to implement innovations in long-term care successfully.

KEYWORDS

Implementation mapping, implementation strategy, innovation, intervention, transitional care, factors, older persons, long-term care.
Contributions to the literature

- This original work provides a comprehensive set of theory and evidence-based implementation strategies selected specifically to improve the implementation of transitional care innovations.
- This set of strategies serves as an invaluable and practical guide for future implementers of transitional care innovations in long-term care practice.
- The application of Implementation Mapping method to select strategies for implementing transitional care innovations was not conducted beforehand.
- The used methodology can be replicated to improve the implementation of other innovations in long-term care.

BACKGROUND

Since the early 2000s, the concept of transitional care, defined as "a broad range of timelimited services designed to ensure health care continuity, avoid preventable poor outcomes among at-risk populations, and promote the safe and timely transfer of patients from one level of care to another or from one type of setting to another", has gained momentum (1, 2). Moreover, improving care transitions for highly vulnerable and chronically ill older persons during their movement between different care settings is emerging as the face of delivering exemplary modern-day long-term care services.

In speed with this, the development of a wide range of transitional care innovations (TCIs) flourished as a potential solution to minimize the care fragmentation and adverse events associated with poor care transitions (3-5). To date, the literature indicates that at least 55 different TCIs were implemented covering multiple care pathways (e.g., hospital to home, home to nursing home, hospital to transfer unit to home) and targeting older persons with various chronic conditions (e.g., stroke, dementia, heart failure) (3, 5-10). Some of these TCIs showed possible beneficial effects such as reducing hospital readmissions, preventing redundant emergency department visits, avoiding unnecessary admission to a nursing facility, enhancing treatment adherence, or improving the quality of life for older persons (9, 11-13).

While the aims of many TCIs are diverse, there are similarities among their components as well as ambiguity on how they are implemented in real practice. Accordingly, a rising universal awareness exists among healthcare professionals, scientists, and policy-makers, that despite the evidence of the benefit of healthcare innovations such as TCIs, their implementation is hardly straightforward (14). Specifically, the implementation of TCIs is often complex and influenced by an array of implementation factors (barriers, facilitators) (3). A lack of organizational resources, low feasibility of implementing the TCI within an organization, and variable staff commitment are among the common barriers. Whereas the presence of staff with transition roles, supportive organizational leadership, and strong engagement of key stakeholders are all facilitating factors to implement a TCI (3, 15). Hence, there is a need to use effective implementation strategies, defined as "methods or techniques used to improve the adoption, implementation, sustainment, and scale-up of evidence-based health interventions into usual care" (16, 17) to address the influencing factors. This will help foster the implementation of TCIs into practice to ensure that older persons receive the expected benefits (14, 18). To promote the use of implementation strategies, several taxonomies and compilations of strategies were developed to help implement interventions in healthcare in a successful way (19-23).

Few studies that implemented TCIs reported on factors (barriers, facilitators) that influenced the implementation (7, 24, 25). Moreover, they hardly indicated if implementation strategies

were used, and if so, what they were. On the other hand, although the selection of implementation strategies to use in implementing TCIs is starting to pick up as indicated in recent studies, it remains vague how implementers of innovations select strategies to improve the implementation (10, 26). Consequently, TCIs tend to be more likely implemented in a manner of either 'this had worked in the past', 'this is known to work', 'this seems promising', or 'this is how we have always done it'. Furthermore, implementation strategies that were used and effective in certain studies are usually selected and copied for use in subsequent studies, despite differences in the intervention itself, the recipients, and the context where they are implemented. Hence, this approach will probably lead to limited success in implementing TCIs (16). Therefore, two problems arise. First, implementation strategies are context-specific, and what works in one context might not work in another. Second, implementation strategies should be linked to the implementation factors in the relevant context as well as selected based on both mechanisms of change that explain how factors can be addressed and on available evidence for their effectiveness (27).

While the literature provides several taxonomies or overviews of theory-based or expertrecommended implementation strategies, some of which provide linkages to implementation factors as well as some guidance on selection, feasibility, and importance of strategies, these stem mostly from fields of implementing interventions in general healthcare (20, 22, 23, 28-30). To date, there is an absence of a set strategies developed for implementing TCIs specifically. This paper aims to describe a novel and systematic development of a set of strategies to improve the implementation of TCIs in long-term care and increase their chances of success. This work is based on integrating findings on barriers to and facilitators of TCIs' implementation from previous studies by the research team and others, to then propose linkages to and the application of implementation strategies to address or leverage these factors. The overall goal of this project is to guide future implementers of TCIs (i.e., scientists, health care professionals, and leaders of long-term care organizations) and help minimize the gap between insights on optimal transitional care from existing TCIs and the limited use of these insights in practice.

METHODS

This project followed a stepwise approach informed by the Implementation Mapping methodology to develop, choose, and design a carefully selected set of implementation strategies specifically for implementing TCIs (31). We formed a working group to perform the different steps consisting of the core research team (AF, BdB, TvA; holding expertise in both transitional care and implementation science) and one additional expert in the Implementation Mapping technique. The Maastricht University Faculty of Health, Medicine, and Life Sciences Ethics Committee approved this work (approval no. FHML-REC/2022/003). This paper followed the Standards for Quality Improvement Reporting Excellence (SQUIRE) guidelines for reporting new knowledge about how to improve healthcare (32).

RATIONALE FOR USING IMPLEMENTATION MAPPING

Based on combining aspects from both implementation science and Intervention Mapping, Implementation Mapping provides five consecutive tasks to develop, select, and design implementation strategies (31). Implementation Mapping incorporates theory and evidence and provides a systematic way to address the key implementation factors by linking them to the relevant change methods to guide successful implementation. For the scope of this project, we applied iteratively only the first three tasks of Implementation Mapping: *1. conduct a needs assessment; 2. identify implementation outcomes, performance objectives, determinants, and change objectives; 3. select theoretical change methods and design implementation strategies,* to develop implementation strategies for TCls, see Table 1.

As a *preliminary step* and prior to performing tasks 1-3 in Implementation Mapping, we opted to first describe what TCIs and their core components are, in order to clarify what the innovation to be implemented is for future users of the implementation strategies. Hence, we utilized the findings from a scoping review by the research team that identified different TCIs and their specific key elements (e.g., case management, follow-up visits after a transition, appointing a transitional care nurse, etc.) (3). For each TCI, we mapped its elements to eight core transitional care components: patient engagement, caregiver engagement, patient education, caregiver education, complexity management, care continuity, wellbeing, and accountability; as defined by Naylor et al. 2017 to achieve a holistic care process (33). Consequently, the elements of the TCIs belonging under each core component were combined. Three of the research team (AF, BdB, TvA) performed this mapping individually and then convened to discuss and compare results until an agreement was reached.

Step 1. Conduct a Needs Assessment

For this step, we integrated the findings of three previous studies (scoping review, Delphi study, collective case study) on TCIs and their implementation (3, 15, 34). This task helped to determine the priority implementation factors (barriers, facilitators) that influence the implementation of different types of TCIs in practice. The Consolidated Framework for Implementation Research (CFIR) was applied to match the implementation factors to the relevant domains and constructs (35).

Step 2. Identify Implementation Outcomes, Performance Objectives, Determinants, and Change Objectives

First, each of the implementation factors determined in step 1 was linked to its equivalent theoretical constructs (i.e. determinants) by considering the definition of each factor and understanding the essence or central meaning of the factor. We used the taxonomies of

behavioral change to identify the equivalent theoretical constructs (23, 28, 36). The core research team members performed this individually and then convened to discuss and compare results until an agreement was reached. Second, aided by the preliminary step on the description of the TCIs, we identified the implementation outcomes (referring to service and/or patient outcomes expected upon implementing a TCI), the actors (who will perform the actions needed to implement the core components of the TCIs), and the performance objectives (what do the actors have to do to promote the implementation of a TCI). Then matrices of change objectives were created, indicating what has to change in the determinants to bring about the performance objectives. AF developed the matrices and initially discussed these with researchers BdB and TvA, and following adjustments, then with the expert on Implementation Mapping, who advised on further alterations and enhancements.

Step	IS	Objectives	Methods
Prel	iminary step	Describe TCIs and identify their core components	 Literature review by the research team (3) Analysis and mapping to transitional care core components proposed by Naylor et al. (33)
1.	Conduct a needs assessment	Determine the priority implementation factors that influence the implementation of different types of TCIs	• Compilation of findings from three previous studies by the research team (literature review, Delphi study, collective case study) (3, 15, 34)
2.	Identify implementation outcomes, performance objectives, determinants, and change objectives	 Identify underlying changeable determinants (theoretical constructs) for the implementation factors from step 1 Determine the TCIs' implementation outcomes Specify performance objectives Develop matrices of change objectives based on performance objectives and determinants of implementing TCIs 	 Data from the preliminary step and needs assessment step Input from the expert on Implementation Mapping
3.	Select theoretical change methods and design implementation strategies	 Identify and select theoretical change methods (strategies) Retrieve empirical evidence on the effectiveness of strategies Validate and refine the final selection of strategies Develop practical applications for selected strategies 	 Literature on theoretical change methods (published behavioral change taxonomies and compilations of implementation strategies) Literature scan for evidence on the effectiveness of strategies Input from experts in the fields of implementation science, transitional care interventions, and long-term care (four consultation sessions)

Table 1. Overview of steps, objectives, and methods performed

Notes: TCIs = Transitional care innovations

Step 3. Select Theoretical Change Methods and Design Implementation Strategies

Step 3 (a) – In this step, theoretical change methods (strategies used to address the determinants relevant for each factor) at the individual or environmental level (including policy, social, and organizational) were selected from four taxonomies or overviews of change methods (22, 23, 28, 30, 37, 38). These taxonomies or overviews indicate which strategies could be used to target the relevant determinants. Hence, a number of potential strategies were selected for each determinant. This step of mapping change methods to determinants was iterative whereby the core researchers performed it individually, after which they convened for four sessions to discuss and compare results until an agreement was reached.

Step 3 (b) – In this step, empirical evidence on the effectiveness of each of the strategies was assessed from published literature where possible. The search was guided mainly by considering systematic reviews and/or randomized-controlled trials or effectivenessimplementation hybrid design studies (39) conducted to demonstrate the effectiveness of strategies to implement innovations in either long-term care, transitional care, or general healthcare settings, as a first choice. In case only studies with other designs (qualitative, mixed-methods) were available to provide evidence on effectiveness, they were considered as a second choice. Moreover, when no evidence for a strategy was found in the literature, we referred to the relevant theory of change as a foundation for potential effectiveness. The main researcher (AF) performed the literature scan for evidence and then summarized the findings on the effectiveness of each strategy. Throughout this process, (AF) performed four individual consultation sessions with experts who provided feedback and advice on the strategies proposed (see description below) and the rationale for these. Furthermore, the three core team researchers discussed the available evidence on each strategy, which led to formulating a narrative conclusion on their effectiveness.

Expert consultation sessions – Individual consultation sessions were performed with four scientific experts who have extensive knowledge and experience in the fields of implementation science, transitional care interventions, and long-term care. The experts were purposefully selected to cover all three areas of expertise and to make sure that the dominant area of expertise varied. Sessions were conducted online using a data-protected videoconferencing platform and each lasted an average of 1.5 hours, and was performed in the same manner and using the same content. The sessions aimed to discuss the various proposed implementation strategies, obtain feedback on their perceived importance, practicality and applicability, and ask for further recommendations on sources of evidence on the effectiveness of, and suggestions for other strategies. This helped to iteratively refine the list of strategies.

Step 3 (c) – The core team held three iterative discussion sessions to determine the final selection of the core strategies based on: i) empirical evidence on effectiveness, ii) support by the relevant theory of change, iii) pragmatic rationale (feasibility, importance, practicality, and applicability of each strategy to the context of transitional care), and iv) feedback from experts' consultations. Consequently, practical applications (i.e. ways to apply and operationalize the strategy) were suggested for each selected implementation strategy, considering the context of transitional care. For each strategy, the target (i.e. who the strategy is directed at), and actor (i.e. who will deliver the strategy) were proposed (17).

RESULTS

We present the build-up of results for each step performed, leading to the final selection of implementation strategies for TCIs.

PRELIMINARY STEP - IDENTIFICATION AND DESCRIPTION OF THE TCIS' CORE COMPONENTS

Twenty different TCIs were identified from previous research work published elsewhere (3). A total of 16 focused on improving care transitions while another four aimed to prevent transitions between care settings such as private homes, hospitals, intermediary care places, and nursing/residential care facilities. All 20 TCIs combined were found to encompass six out of the eight proposed core components of transitional care, as defined by Naylor and colleagues (33). Table 2 describes the key elements of all 20 TCIs mapped to and combined under each of the six core components for transitional care (patient engagement, caregiver engagement, patient education, caregiver education, complexity management, and care continuity). Based on the existing TCIs, this table describes what a typical TCI is usually composed of and serves as a basic guide to key elements found in various innovations in transitional care. Care continuity presents as an extensive core component that prevails in the majority of TCIs. Hence, it is a backbone component specific to these innovations, given their nature to organize the continuum of care for older persons across different settings.

Core Component	Elements
1. Patient Engagement	 Establishment of trusting relationship with patient Development of rapport with patient, and understand the patient's goals & preferences Active engagement of patient, family caregivers, and collaboration with primary care providers Active involvement of patients and informal caregivers for example in a triage decision-making for care transitions Integration of psychotherapeutic methods in care coordination and case management to increase patient engagement (e.g., motivational interviewing & behavioral activation)
2. Caregiver Engagement	 Establishment of trusting relationship with caregiver Active engagement of patient, family caregivers, and collaboration with primary care providers Active involvement of patients and informal caregivers for example in a triage decision for care transitions
3. Patient Education	 Discharge planning using "teach-back" methods Active engagement of patients and their family or informal caregivers by focusing on education and support Improvement of patient's capacity in: medication self-management, using a patient-centered health record, knowledge of "red flags", and making primary care provider/specialist appointments Provision of patient and caregiver education tools Coordination of education and community services to develop self-management skills Provision of a 30-day post-acute care bundle of transitional care services
4. Caregiver Education	 Building of the caregiver's ability to identify early symptoms and apply strategies to prevent poor outcomes for patient Active engagement of patients and their family or informal caregivers by focusing on education and support Provision of patient and caregiver education tools
5. Complexity Management	 Development of individualized care plans, patient-caregiver goals (with patient, caregiver, and healthcare providers) Implementation of risk reduction strategies to minimize effects for example of cognitive impairment Daily hospital visits to patient-caregiver dyad as well as pre-discharge In-hospital patient case assessment and development of care plan Advanced care planning (assessment of needs at patient home and building a tailored care plan) Early identification and response to health risks of patient Comprehensive patient assessment within 3 days upon discharge by a home care nurse Development of a care plan based on input from patient and caregiver as well as a biopsychosocial needs assessment (post-discharge) Provision for example of a triage instrument for in-hospital assessment of patient needs for admission to a geriatric-rehabilitation unit before movement to a home setting Provision of acute-level care services at home as a substitute for hospital admission Identification, and communication of the resident status changes using communication tools (stop and watch warning tool), decision support (care pathways), and advance care planning Provision of a patient-centered holistic approach

 Table 2. Summary description of the TCIs' elements as mapped to the six core components of transitional care

Core Component	Elements
6. Care Continuity	Majority of elements in this core component include a 'Transition Role' with various tasks,
	described below
	 Presence of staff with a transition role such as Advanced Practice Nurse to perform: home/nursing facility visits 24 hours post-discharge
	 ✓ Home/fullsing facility visits 24 hours post-discharge ✓ telephone follow-up and support
	 ✓ coordination with a multidisciplinary local team of healthcare experts
	 Presence of staff with a transition role such as <i>Transitional Care Manager</i> (social worker, or any other healthcare professional, except nurses) to perform: ✓ discharge planning by management of environmental and community barriers ✓ coordination of transitional care (e.g., implementing home care, organizing home visits by professionals, delivering equipment to patient home, contacting long-term care placement agency, and reorienting patient when needed to rehabilitation services) ✓ exchange of patient information between providers
	 Presence of staff with a transition role such as <i>Health Coach</i> (nurse or social worker) to perform:
	✓ home visits (within 24–48 hours) post-discharge
	 ✓ follow-up phone calls and appointments with primary care providers ✓ connecting older adults to community services and resources
	 Presence of staff with a transition role such as <i>Hospital Care Transition Nurse</i> to perform: patient care handoff between hospital care transition nurse and community rapid response nurse
	✓ home care and follow-up period up to 30 days
	\checkmark referral to hospital-based chronic disease management clinics
	 Presence of staff with a transition role such as <i>Transitional Care Nurse</i> to perform: ✓ care coordination among providers and ensuring a multidisciplinary approach with open communication
	 ✓ regular home visits and ongoing telephone support (7 days/week over 2 months post-dis charge)
	 ✓ continuity of medical care with hospital/primary care and accompanying patients on follow-up visits
	 Presence of staff with transition role such as Care Coordinator to perform: ✓ home visits, telephone monitoring
	✓ care coordination with a network of medical and social care providers in/out of hospital (including: symptoms management, functional management, psycho-support, medication management, promotion of self-care, referral to other services, appointment management, management of social issues, assessment of home environment)
	• Presence of staff with transition role such as <i>Transition Coach</i> (nurse or social worker), to perform home visits post-discharge and follow-up telephone calls
	 Presence of staff with transition role such as <i>Care Coordinator</i> (social worker) to perform: care coordination and follow-up in person or by telephone throughout 30 days post- discharge (e.g. provide brief counseling, arrange services and follow-ups, collaborate with other healthcare and social service providers)
	 Presence of staff with transition role such as Care Pathway Coordinator in order to perform:
	✓ coordination and continuity of care across settings

Core Component	Elements
	 Exchange of patient discharge information between the hospital, local healthcare allocations (municipality-level), and home care services, in order to: vevaluate and decide on care assistance v prepare home care service for transition v provide general practitioner consults to patient 14 days post-discharge v perform extended assessment during the first 4 weeks by district nurse/nursing assistant
	 Delivery of transition care in temporary transition care places (e.g., low-intensity therapies and services, case management, & finalization of long-term care arrangements) Provision of community geriatric services (e.g., geriatrician and community nurse, 24 hours telephone support and advisory service to nursing facility staff) Collaboration of hospital and community-based organizations for aging services (network collaboration) Availability of a Community Psychiatric Nurse, to perform follow-up visits and provide support/advice to family caregivers and facility nurses after the placement of an older person in a nursing home facility Provision of a 30-day post-acute period of transitional care bundle care coordination Combination of disease management in primary care settings and home care by coordinating care during an episode of acute illness across settings, and facilitated by a transitional care nurse

Step 1. Conduct a Needs Assessment

Twelve factors were selected as key implementation factors for TCIs, based on the combined results of previous studies conducted by the research team (3, 15, 34). These factors were reported and concluded as the most important to address in implementing TCIs. Table 3 describes these factors spanning the five domains of the CFIR. The majority of factors were at the organizational level (inner setting), such as the leadership commitment, involvement, and role in initiating the implementation of TCIs. Moreover, the availability of organizational resources along with the provision of access to knowledge and information on the TCIs were key factors to support the implementation. Continuous information exchange among various care providers involved in a care transition is another factor highlighted, as well as the sense of urgency to implement a TCI within an organization and the perception of it as a relative priority by individuals. Besides, one factor was linked to the outer setting and related to systems to finance the TCIs' implementation. At the process level, engagement of key stakeholders and main participants in the implementation, as well as creating transition roles of staff, and evaluating the implementation process were regularly indicated to affect implementing TCIs. Furthermore, designing a TCI to match the care needs of the targeted groups (older persons) and considering the knowledge, beliefs, and personal attributes of healthcare professionals involved in implementing TCIs were also described as factors pertaining to each of the characteristics of the innovation and the individuals.

Step 2. Identify Implementation Outcomes, Performance Objectives, Determinants, and Change Objectives

Table 3 lists the multiple determinants (i.e. relevant theoretical constructs) identified for

each of the 12 factors, (e.g., attitudes, beliefs, and motivation for the factor engagement). In addition, based on the preliminary step, we determined the relevant TCIs' implementation outcomes, the corresponding necessary performance objectives to achieve these outcomes, and the actors: i) leaders and organizations; and ii) healthcare professionals. Determinants were allocated to the pertaining performance objectives. Hence, two matrices of change were developed for each type of actor, and we formulated the change objectives linked to the performance objectives and determinants (see Appendix 6A for matrices of change). As an example, one performance objective was to engage the patient and caregiver, which is linked to the determinants (attitudes and beliefs of healthcare professionals), and hence the corresponding change objectives were to believe that the TCI is beneficial to enhance transitional care for older persons and to express a positive attitude towards the TCI as an innovation.

Implementation Factor ^a (description)	CFIR domain	Determinant (i.e. equivalent theoretical construct)
Leadership (Commitment, involvement, and accountability of leaders and managers with the implementation of a TCI. In addition, the presence of a skilled, motivated, and continuous leadership throughout the implementation, e.g., minimal turnover of dedicated project managers with high interest in the implementation.)	Inner setting ^b	 Commitment Organizational commitment Leadership
 Engagement (Attracting and involving appropriate individuals as below in the implementation and use of the intervention through a combined strategy of social marketing, education, role modeling, training, and other similar activities): Key stakeholders (individuals from within the organization that are directly impacted by the TCI, e.g., staff responsible for making referrals to a new program or using a new work process), Champions (individuals who dedicate themselves to supporting, marketing, and 'driving through' an implementation, overcoming indifference or resistance that the intervention may provoke in an organization that participate in the TCI, e.g., ensuring family, and informal caregivers served by the organization that participate in the TCI, e.g., ensuring family inclusion in care goals setting). Organizations/external context (developing and capitalizing on relationships with healthcare professionals and frontline staff in the implementing organizations, and promoting external collaborations with outside care providers (e.g. home care agency), and resources (e.g. community resources or social services for older persons) linked to the implementation of a TCI) 	Process	 Attitudes Beliefs Motivation
Information continuity (Care transitions' information continuity such as exchange of patient medical information, services, and care plans between healthcare providers. In addition, the continuity of steady work relationships between the healthcare providers and patients/caregivers and across all the organizations involved in the TCI implementation.)	Inner setting	 Environmental conditions (structural, organizational) Social networks

Table 3. Implementation factors with	h the corresponding	equivalent determinants
--------------------------------------	---------------------	-------------------------

Implementation Factor ^a (description)	CFIR domain	Determinant (i.e. equivalent theoretical construct)
Financing of TCIs' implementation (The existing financing structures that affect the TCI implementation such as fragmented financing and a lack of clear financing structures, or varied reimbursement systems of healthcare services.)	Outer setting	• Policy
Available resources & HIT systems (The level of resources dedicated for the implementation and on-going operations of a TCI; including staffing levels, money, funding, training, education, physical space, equipment, and time. HIT – the electronic information management infrastructure and technologies, e.g., electronic health records available to clinicians to manage patient care, data, and communications.)	Inner setting	 Environmental conditions (structural, organizational)
Access to knowledge & information (Ease of access to digestible information and knowledge, e.g., mentoring, initial training about the TCI and how to incorporate it into work tasks.)	Inner setting	 Environmental conditions (structural, organizational)
Sense of urgency (The urgent need and attention given to implementing a specific TCI with respect to other innovation projects being addressed within an organization.)	Inner setting	Risk perceptionOrganizational commitment
Relative priority (Individuals' e.g., healthcare professionals, staff within implementing team, shared perception of the importance of the implementation of a TCI within the organization)	Inner setting	AttitudesBeliefsOrganizational commitment
Reflecting & evaluating (Quantitative and qualitative feedback about the progress and quality of implementation accompanied with regular personal and team debriefing about progress and experience.)	Process	 Feedback and reinforcement Monitoring
Targeted groups (Patients/older persons who are the intended recipients or beneficiaries of the TCI, e.g., matching the care needs of older persons with high frailty or dementia.)	Intervention characteristics	Innovation's compatibility
Transition roles (Administrative staff, providers within and outside the organization, e.g., frontline staff such as transition nurses or advanced practice nurses with designated transition roles who will carry out the TCI or be affected by it.)	Process	 Environmental conditions (structural, organizational) Professional role
Knowledge, beliefs, and personal attributes of healthcare professionals (Individuals' attitudes toward and value placed on the intervention as well as familiarity with facts, truths, and principles related to the TCI, and other beliefs, expectations, and personal traits such as motivation levels, values, tolerance of ambiguity, critical attributes, intellectual ability, capacity, and learning style.)	Characteristics of individuals	 Knowledge Attitudes Beliefs Motivation Skills

Notes: ^aFactors are listed in descending ranking order of priority, with leadership as most important; check reference for Delphi study (15) for further details; ^bInner setting is also referred to as the organizational context.

Step 3. Select Theoretical Change Methods and Design Implementation Strategies

Step 3 (a, b) - This step was completed by selecting strategies (i.e. theoretical change methods) expected to address the determinants and change objectives identified in step 2. An extensive list of strategies, such as modeling and active learning from the Social Cognitive Theory, consciousness raising from the Trans-Theoretical Model, persuasive

communication from the Communication-Persuasion Matrix, and belief selection from the Theory of Planned Behavior, was identified (23, 28). Theories of Organizational Development and Organizational Readiness for Change provided a number of strategies to address determinants such as structural influences and organizational commitment (23, 28, 40). Other selected strategies included building a coalition, conducting local consensus discussion, role expansion/task shifting, and revising professional roles (30, 37). All identified methods focused on either the organizational level (e.g., home care organization, hospital, etc.) or at the individual level (e.g., general practitioners, nursing home staff, etc.), see Table 4. For example, persuasive communication can be used to create convincing arguments on the importance and effectiveness of a TCI in improving care transitions and hence address the commitment (determinant) of leadership to implement a new TCI in an organization. Similarly, to improve the knowledge and attitudes of healthcare professionals, using guided practice as a method to help train their ability to deliver the components of a TCI can address the skills (determinant) of the professionals involved in implementing the TCI.

The list of identified strategies was iteratively refined following feedback from the experts' consultation sessions, who reviewed, validated, and proposed new strategies or amendments. Specifically, the experts confirmed the selection of strategies to address the leadership skills and capabilities, as they were considered a priority for implementing TCIs. Moreover, they emphasized the importance of the strategy 'participation', and the inclusion of certain aspects within it such as stakeholder mapping and building an interdisciplinary coalition across care settings. Similarly, strategies to enhance communication and information exchange, and networking across care settings and healthcare professionals were proposed as essential for implementing TCIs. The strategy of 'advocacy and lobbying' at the policy level was indicated by experts as difficult and time-consuming, yet important to keep in order to have continuous catalysts to lobby for implementing innovations in transitional care. Tailoring the TCIs was also considered a necessary strategy by the experts and was proposed to be operationalized as conducting local care and needs assessment to make the innovation context/target population specific.

A number of strategies were supported by either only theory (e.g., consciousness raising) or only evidence from the literature (e.g., conduct local consensus discussions), while others were supported by both (e.g., modeling, participation). Evidence on the effectiveness of some identified strategies to achieve change was retrieved from the literature and where possible strategies could be denoted as having either a positive effect or association to implement an innovation in a care setting. Moreover, some strategies were indicated as having various degrees of effects such as small, modest, and moderate. Many studies explored the effects of a combination of strategies (multifaceted) on implementing change (41-44), and few were specific to long-term care settings and transitional care (45-47). Table 4 describes the summary list of the final implementation strategies that were selected and details the relevant theory and/or evidence on effectiveness for each.

Step 3 (c) - Eventually, a total of 40 strategies were selected, four of which (persuasive communication, belief selection, structural redesign, and organizational diagnosis and feedback) address more than one factor, see Table 4. The majority of strategies (n=21) were at the organizational level and almost half were supported by evidence as having a positive effect on implementing change such as TCIs in practice (45-49). For example, facilitation and creating a supportive organizational environment proved as effective to improve the adoption and implementation of new guidelines in clinical settings (49). Likewise, using communication systems including health information technology (HIT) to improve information continuity among providers within and across care organizations improved adherence to new guidelines in a nursing home (46). Sense-making was another strategy that can effectively address organizational leadership and foster leaders' commitment to implementing an innovation such as TCIs (48). Some strategies (e.g., building a coalition, enhancing network linkages, changes in staffing models, and developing resource sharing agreements) exhibited a positive - but not necessarily causal - association with implementing change, and they were selected due to their high relevance to the context of transitional care, whereby multiple care settings and organizations are usually involved in implementing TCIs (41, 43, 44). Other selected strategies (e.g., audit and feedback, educational materials, educational meetings/training) at the organizational level are commonly used and presented with effect sizes on implementing change, hence they were considered essential in the implementation of TCIs (50-52).

Strategies at the individual level (n=13) such as belief selection and scenario-based risk information, were all supported by theory. Participation, modeling, and guided practice were the only three strategies that were supported by evidence, in addition to theory, as having a positive effect or association (42-44, 53-56). For example, engaging all the key stakeholders in an implementation effort early on and continuously could directly improve the adoption and implementation of an innovation.

Otherwise, one strategy at the policy level (i.e. advocacy and lobbying) was selected, and another at the innovation level (i.e. tailoring). Evidence on tailoring as a strategy indicated a positive effect, hence matching the TCIs' components to the care needs of older persons is considered essential for successful implementation (57, 58).

Afterward, suggestions on how to operationalize the selected strategies were created as practical applications and the corresponding target was proposed, see Table 5 for a full description. For example, a public announcement of the introduction of a new TCI made by an organizational leader and included in a newsletter could increase the organizational commitment to implementing the TCI.

ess
iven
fect
n ef
ce o
iden
r ev
o/pu
ory a
theo
/ant
rele
and
ion,
cript
des
gies,
ateç
n st
itatio
men
nple
i pa
elect
of se
List
e 4.
able

Factor	Stra	Strategy (change method)	Description	Relevant theory and/or evidence from literature
Leadership		Persuasive communication	Guiding individuals and environmental agents toward the adoption of an idea, attitude, or action by using arguments or other means.	 Communication-Persuasion Matrix, Elaboration Likelihood Model, Diffusion of Innovations Theory (23, 28) The use of persuasive, novel, relatable, and important arguments and communication with leaders can be effective to achieve change.
	Э	Public commitment	Stimulating pledging, promising or engaging oneself to perform the healthful behavior, & announcing that decision to others.	 Theories of Automatic, Impulsive and Habitual Behavior (23, 28) Indications for positive effect: organizational leaders committed to implementing an intervention to reduce hospitalization from a nursing home directly increased the staff commitment to implementation (47).
	м	Sense-making	Leaders reinterpret and relabel processes in organization, create meaning through dialogue, and model and redirect change.	Organizational Development Theory (23, 28) > Indications for positive effect: a committed organizational leadership who created an encouraging environment within healthcare organizations (e.g., supportive policies and practices, allocation of funds and resources) to implement an innovation increased directly the staff commitment to implementation (48).
	4.	Development of leadership skills	Development of strong leadership skills in clinical leaders to align and sustain evidence-based practice implementation efforts.	Development of strong leadership skills in clinical > Indications for positive effect: demonstrated as useful to build leaders to align and sustain evidence-based practice implementation efforts. practice implementation efforts.

intinued.	
ő	
4	
<u>e</u>	
ą	
Ъ	

196

Factor	Strategy	egy (change method)	Description	Relevant theory and/or evidence from literature
Engagement	ப்	Participation	Assuring high level engagement of the participants' group in problem solving, decision making, and change activities; with highest level being control by the participants' group.	 Diffusion of Innovations Theory, Theories of Power, Organizational Development Theories, Models of Community Organization (23, 28) Indications for positive effect: engaging stakeholders (e.g., managers, local providers) through meetings, briefings, and consultations increased the reach and adoption of a mental health intervention in primary care clinics (53). Patient activation improved health-related outcomes suggesting substantial positive effects (54).
	ن	Modeling	Providing an appropriate model being reinforced for the desired action.	 Social Cognitive Theory (60) Providing models (similar persons) that can do the same tasks and demonstrate steps to attain a complex objective can be effective to motivate and increases self-efficacy. Indications for positive association: champions were related to increased use of innovations in healthcare organizations (55) / identifying <i>early adopters</i> and learning from their experiences was associated with the uptake of a new evidence-based treatment in a clinical care setting (43, 44). Indications for positive effect: oplinion leaders can probably improve healthcare professionals' compliance with evidence-based practice (effect sizes are modest and varied) (56).
	7.	Build a coalition	Recruit and cultivate relationships with partners in the implementation effort.	Recruit and cultivate relationships with partners in in indications for positive association: building local teams to address the implementation effort. based treatment in a clinical care setting (43, 44).
	α	Conduct local consensus discussions	Include local providers and other stakeholders in discussions that address whether the chosen problem is important and whether the clinical innovation to address it is appropriate.	Indications for positive effect: local consensus discussions proved effective to improve adherence to a new guideline in a nursing home (used in combination with other strategies) (45).
	ര്	Use mass media	Use media to reach large numbers of people to spread the word about the clinical innovation.	Indications for positive effect: evidence showed that these channels of communication may have an important role in influencing the use of healthcare interventions and services (61).

Factor	Stre	Strategy (change method) Description	Description	Relevant theory and/or evidence from litera
Information continuity	<u>.</u>	10. Structural redesign	Change organizational elements such as formal statements of organizational philosophy, communication flow, reward systems, job descriptions, and lines of authority.	Organizational Development Theory (23, 28) Planned development, improvement, or rei and processes can promote organizational deal with resistance to change.
	11.	Change in communication between providers, including distant ones	Systems to improve the communication between > Indications for positive effect: the use of the althcare providers, including establishment systems improved adherence to a new gui of any type of telecommunication link for (used in combination with other strategies) implementation (e.g., use of information and communication technology).	Indications for positive effect: the use of i systems improved adherence to a new gui (used in combination with other strategies)
	5	Enhancing network	12. Enhancing network Training network members to provide support	Theories of Social Networks and Social Supp. Theories of Social Networks and Social Supp.

Factor	Strat	Strategy (change method) Description	Description	Relevant theory and/or evidence from literature
Information continuity	<u>.</u>	Structural redesign	Change organizational elements such as formal statements of organizational philosophy, communication flow, reward systems, job descriptions, and lines of authority.	 Organizational Development Theory (23, 28) Planned development, improvement, or reinforcement of structures and processes can promote organizational effectiveness and help deal with resistance to change.
	,	Change in communication between providers, including distant ones	Systems to improve the communication between healthcare providers, including establishment of any type of telecommunication link for implementation (e.g., use of information and communication technology).	Indications for positive effect: the use of HIT communication systems improved adherence to a new guideline in a nursing home (used in combination with other strategies) (46).
	5.	Enhancing network linkages	Training network members to provide support and members of the target group to mobilize and maintain their networks.	Theories of Social Networks and Social Support (23, 28) P Indications for positive association: promoting working relationships and networks inside/outside the organization was associated with the uptake of a new evidence-based treatment in a clinical care setting (43, 44).
Financing of TCIs' implementation	Έ	Advocacy & Lobbying	Arguing and mobilizing resources on behalf of a particular change, giving aid to a cause, active support for a cause or position.	 Stage Theory of Organizational Change: Models of Community Organization: Agenda-Building Theory: Multiple Streams Theory (23, 28) Is a primary method to achieve change at the environmental/ community level and highlights the role of community activism to support important issues.

198

Factor	Strat	Strategy (change method)	Description	Relevant theory and/or evidence from literature
Available resources & HIT systems	4.	Changes in Staffing models	Interventions to achieve an appropriate level and mix of staff, recruitment and retention of staff, and transitioning of healthcare workers from one environment to another, for example interventions to increase the proportion of healthcare workers in underserved areas.	Indications for positive association: forming new interdisciplinary teams was associated with the uptake of a new evidence-based treatment in a clinical care setting (43, 44).
	15.	Use of HIT systems	Health record and health management systems to store and manage patient health information, for example electronic patient records, or systems for recalling patients for follow-up or prevention.	Indications for positive effect: the use of HIT communication systems improved acherence to a new guideline in a nursing home (used in combination with other strategies) (46).
	9.	Develop resource sharing agreements	Develop partnerships with organizations that have resources needed to implement the innovation.	Organizational theories – Resource Dependency Theory (62) Todications for positive association: partnening with other resourceful organizations can help implement changes and was associated with the uptake of a new evidence-based treatment in a clinical care setting (43, 44).

ed.
ntinu
Ö.
ole 4
Tab

Factor	Stra	Strategy (change method)	Description	Relevant theory and/or evidence from literature
Access to knowledge & information	Ę.	Facilitation	Creating an environment that makes the action easier or reduces barriers to action.	Social Cognitive Theory (60) Indications for positive effect: improved the adoption of evidence- based guidelines in various clinical areas that focused on prevention, system-level improvements, and outcomes associated with chronic disease management within clinical practice settings (49).
	ĕ	Develop and distribute educational materials	Develop guidebooks, manuals, toolkits, and other supporting materials in ways that make it easier for stakeholders to learn about the innovation and for clinicians to learn how to deliver the clinical innovation, and distribute it in person, by mall, and/or electronically.	Indications for positive effect: can probably improve healthcare professionals' practice but slightly, and electronic versions have little or no difference compared to printed versions (effect sizes are small and variable) (51).
	<u>0</u>	Conduct educational meetings and ongoing trainings	Hold meetings and trainings (e.g., courses, workshops, conferences) in an ongoing way targeted toward different stakeholder groups (e.g., providers, administrators, other organizational stakeholders, and community, patient/consumer, and family stakeholders) to teach them about the clinical innovation.	▶ Indications for positive effect: can probably improve compliance with desired practice, and interactive formats are more effective than didactic ones (effect sizes are modest) (52).
	20.	Create a learning collaborative	Facilitate the formation of groups of providers or provider organizations and foster a collaborative learning environment to improve implementation of the clinical innovation.	➤ Indications for positive association: improvement in the adoption and implementation quality of expanding new clinical services in primary care settings were moderately associated with the engagement of clinicians in learning collaboratives (4).
Sense of urgency	2 1.	Consciousness raising	Providing information, feedback, or confrontation about the causes, consequences, and alternatives for a problem or a problem behavior.	Health Belief Model, Precaution-Adoption Process Model, Trans- Theoretical Model (23, 28) Theoretical Model (23, 28)
	22.	Scenario-based risk information	Providing information that may aid the construction of an image of the ways in which a future loss or accident might occur.	can be effective to change perceptions and implement change.
	23.	Organizational diagnosis and feedback	Assessing of organizational structures and employees' beliefs and attitudes, desired outcomes and readiness to take action, using surveys and other methods.	Organizational Development Theory, Organizational Readiness for Change (23, 28, 40) Assessing an organization's readiness and accordingly fostering its capacity, capabilities, commitment, and efficacy to change are key drivers to developing readiness to implement change within an organization.

ntinued.	
8	
9	
4	
<u>e</u>	
P	
Га	

Factor	Strat	Strategy (change method)	Description	Relevant theory and/or evidence from literature
Relative priority	24.	Belief selection	Using messages designed to strengthen positive beliefs, weaken negative beliefs, and introduce new beliefs.	Theory of Planned Behavior, Reasoned Action Approach (23, 28) leantifying the beliefs and awareness level of the receiver regarding the problem and its negative outcomes, then providing a relevant positive evaluation for a recommended behavior can be effective to achieve change.
	25.	Persuasive communication	Guiding individuals and environmental agents toward the adoption of an idea, attitude, or action by using arguments or other means.	Communication-Persuasion Matrix, Elaboration Likelihood Model, Diffusion of Innovations Theory (23, 28) The use of persuasive, novel, relatable, and important arguments and communication with leaders can be effective to achieve change.
	26.	Organizational diagnosis and feedback	Assessing of organizational structures and employees' beliefs and attitudes, desired outcomes and readiness to take action, using surveys and other methods.	 Organizational Development Theory, Organizational Readiness for Change (23, 28, 40) Assessing an organization's readiness and accordingly fostering its capacity, capabilities, commitment, and efficacy to change are key drivers to developing readiness to implement change within an organization.
Reflecting & evaluating	27.	Audit and feedback	A summary of health workers' performance over a specified period of time, given to them in a written, electronic or verbal format. The summary may include recommendations for clinical action.	 Theories of Learning, Goal-Setting Theory, Social Cognitive Theory (23, 28) Indications for positive effect: can probably lead to important improvements in professional practice (effect sizes are small), yet is most effective when the performance is low, provided regularly by supervisors or colleagues, and includes clear targets/action plans (50).
	28.	Monitoring the performance of the delivery of healthcare	Monitoring of health services by individuals or healthcare organizations, for example by comparing with an external standard.	Indications for positive effect: improved adherence to a new guideline in a nursing home (used in combination with other strategies) (45, 46).
Targeted groups	29.	Tailoring	Matching the intervention or components to previously measured characteristics of the participant.	Trans-Theoretical Model, Precaution Adoption Process Model, Protection Motivation Theony, Communication-Persuasion Matrix (23, 28) 28) > Indications for positive effect: suggested as important and essential to design evidence-based treatments in mental health (57) can probably lead to the successful implementation of innovations in long-term residential care (58)

Factor	Stra	Strategy (change method)	Description	Relevant theory and/or evidence from literature
Transition roles	30.	Structural redesign	Change organizational elements such as formal statements of organizational philosophy, communication flow, reward systems, job descriptions, and lines of authority.	Organizational Development Theory (23, 28) > Planned development, improvement, or reinforcement of structures and processes can promote organizational effectiveness and help deal with resistance to change.
	31. &	Role expansion or task shifting	Expanding tasks undertaken by a cadre of health workers or shifting tasks from one cadre to another, to include tasks not previously part of their scope of practice.	Indications for positive effect: showed improvements in the process of care outcomes (49) / was associated with the uptake of a new evidence-based treatment in a clinical care setting (43, 44).
	32.	Revise professional roles	Shift and revise roles among professionals who provide care, and redesign job characteristics.	
Knowledge, beliefs, and personal attributes	33.	Active learning	Encouraging learning from goal-driven and activity-based experience.	
or nearncare professionals	34.	Chunking	Using stimulus patterns that may be made up of parts but that one perceives as a whole.	Elaboration Likelihood Model, Social Cognitive Theory. Theories of Information Processing, Theories of Stigma and Discrimination,
	35.	Advance organizers	Presenting an overview of the material that enables a learner to activate relevant schemas so that new material can be associated.	 Communication-reneated action Approach (23, 28) Planned Behavior, Reasoned Action Approach (23, 28) All these methods can help improve the individual's capabilities and
	36.	Shifting perspective	Encouraging taking the perspective of the other.	aptitude to implement change.
	37.	Arguments	Using a set of one or more meaningful premises and a conclusion.	
	38.	Direct experience	Encouraging a process whereby knowledge is created through the interpretation of experience.	
	39.	Belief selection	Using messages designed to strengthen positive beliefs, weaken negative beliefs, and introduce new beliefs.	
	40.	Guided practice	Prompting individuals to rehearse and repeat the behavior various times, discuss the experience, and provide feedback.	 Social Cognitive Theory: Theories of Self-Regulation (23, 28) Indications for positive effect: coaching can lead to the implementation of more quality improvement interventions for cardiovascular health (42).

Factor	Stra (chá	Strategy (change method)	Practical applications (suggested)	Target (directed at)
Leadership	÷	Persuasive communication	 Persuasive communication Organize meetings with organizational leaders to introduce the new TCI and deliver convincing arguments and important messages (e.g., evidence on the effects of TCIs, need for person-centeredness, etc.) Perform open discussions with organizational leaders to discuss objections, change their attitude/perspective and guide them toward adopting the TCI 	Leaders of organizations (including management staff)
		Public commitment	 Obtain formal written commitments from organizational leaders to implement the TCI Make a public declaration/announcement expressing organizational leaders' determination to implement the TCI (e.g., put proof on websites, or in a newsletter) 	Leaders of organizations (including management staff)
	'n	Sense-making	 Assess and analyze current care integration and care continuity practices, then reinterpret to signal shortcomings and the need to provide transitional care in a new or different way. Recreate Anorcesses and redefine organizational ethos/culture to support 	Organizational board of directors, executives, Quality improvement department

Leaders of organizations (including management staff)

Provide coaching on the job by an external expert to promote the leaders' readiness and skills

implementation LOCI training) (59).

•

Train on the change effort (e.g., group-based interactive and didactic training sessions, following the leadership and organizational change for

Identify formal and informal leaders and assess the need for leadership

skills-building

Development of leadership skills

4.

•

innovations and new ideas for transitional care

+	
đ) ת
L'	-
+	3
τ	2
,	3
U	ĵ
2	5
ti	5
C	
C	
2	2
-	;
ĉ	5
÷	5)5
č	3
Ē)
τ	5
Ť.	
ď)
2	ת ת
	2
U	2
ŭ)
C	- ת
D D D	- ת
rated	ימנימי
strated	- ת
n strated	בינית
1tr	בינית
1tr	בינית
1tr	
mentation str	בינית
1tr	
mentation str	
t of selected implementation str	
mentation str	
l ist of selected implementation str	
l ist of selected implementation str	
l ist of selected implementation str	
l ist of selected implementation str	
. List of selected implementation str	

Factor	Stra (cha	Strategy (change method)	Practical applications (suggested)	Target (directed at)
Engagement	ப்	Participation	 Identify relevant key persons (stakeholders) Map stakeholders by their influence, power, authority, and importance in relation to implementing the TCI Activate stakeholder groups (e.g. older persons associations, long-term care lobbysis) Identify issues in suboptimal transitional care and involve stakeholders to take control and be accountable to resolve the problems jointly 	Stakeholders
	ö	Modeling	 Identify role-models within the local context (e.g., champions, early adopters, opinion leaders) Support the role-models to lead the TCI and to help influence and motivate fellow colleagues to adopt the TCI and help drive the implementation Invite the role models to share their experiences with the TCI in practice (e.g., testimonials, informal meetings to share stories) 	Healthcare professionals
	7.	Build a coalition	Organize meetings and information sessions across different care settings (e.g., hospital, homecare, social care) to increase mutual understanding about the TCI implementation workflow, i.e., align on common goals, values, and needs	Organizations, Healthcare professionals
	œ	Conduct local consensus discussions	 Conduct discussions with staff of different care settings (e.g., hospital, homecare, social care) to agree on core problems, needs, and goals, and to assess the benefit of implementing the TCI to address issues in delivering transitional care 	Organizations, Stakeholders
	Ö	Use mass media	 Spread information about the TCI and its benefits using media channels (e.g., a quick quide on the website, refresher sessions, newsletters, and social media) 	Stakeholders

lable 5. Continued.				
Factor	Strat (cha	Strategy (change method)	Practical applications (suggested)	Target (directed at)
Information continuity 10.	10.	Structural redesign	Develop a formal communication workflow, and a guide for healthcare professionals to integrate care collaboratively and to share information between different care settings (e.g., hospital, homecare, social care) involved Develop a standardized discharge letter to indicate the treatment plan upon transition from one care setting to another (e.g., hospital to home care)	Organizations, Healthcare professionals
		Change in communication between providers, including distant ones	Develop HIT systems interlinked between different organizations, or find ways to automatize information exchange between systems when possible Optimize communication links (e.g., smartphone communication apps, conferencing technology, web chat, email)	Organizations
	<u>5</u>	Enhancing network linkages	Create inter-organizational networks (common platforms, regular conferences & meetings, collaborative learning sessions) to discuss issues and updates in transitional care delivery Conduct teams meetings for staff implementing the TCI to share experiences, problem-solving, and support one another	Organizations, Healthcare professionals
Financing of TCIs' implementation	ά	Advocacy & Lobbying	Develop a proposal for new funding formulas supportive of healthcare providers to implement innovations in transitional care Draft a new policy brief and business case supportive of improving transitional care Identify key funders in the system and request discussions with them Identify the key TCIs' leaders and prepare the advocacy and lobbying with them	National healthcare system, Ministry of health, Regional public health governance, Health insurers, Funding agencies
Available resources & HIT systems	4.	Changes in Staffing models	Identify staff needed to improve transitional care and implement TCIs Develop a proposal for new staffing arrangements (number, roles) Develop a proposal for retaining & attracting transitional care experts (e.g., care transition nurse)	Organizations, Policymakers, Funders
	15.	Use of HIT systems	Assess existing/non-existing HIT systems Develop a proposal to use and potentially integrate HIT systems to share medical information in care transitions and to support TCI implementation	Organizations
	16.	Develop resource sharing agreements	Establish formal agreements (e.g., memorandum of understanding) between LTC organizations on resource sharing Create formal/informal networks between LTC organizations	Organizations

lable 3. Continued.				
Factor	Stra (cha	Strategy (change method)	Practical applications (suggested)	Target (directed at)
Access to knowledge & information	17.	Facilitation	 Use change agents/facilitators to build a relationship with healthcare professionals Provide practical support, relevant materials and means, and information to enhance the use of TCIs (e.g., helpdesk, outreach support, information zone) 	
	18.	Develop and distribute educational materials	 Develop supporting educational materials (e.g., manual, toolkit, brochures, guidebook, reminders, videos) to inform about the TCI and how to implement it, including technology-delivered content (e.g., online, smartphone applications) 	Lookhooro oroforocionalo
	19.	Conduct educational meetings and ongoing training	 Conduct training, workshops, conferences, lectures, and meetings to inform about the TCI and how to implement it (use both didactic and interactive formats) 	הפמנוזכמיב אומיביא מומוא
	20.	Create a learning collaborative	 Foster the formation of a long-term network group of healthcare professionals for learning and exchanging ideas on how to provide optimal transitional care. Organize educational outreach visits by a trained person to meet with healthcare professionals in their work settings to educate them about the TCI and provide support in using it. Establish academic partnerships with a university or academic unit to share research sitis on indhemoting the TCI. Use video stories from other organizations that implemented a TCI 	
Sense of urgency	21.	Consciousness raising & Scenario-based risk information	 Use media to spread testimonials of what went wrong during care transitions Conduct informational meetings to raise awareness and highlight the need for the TCI (use a real case to discuss risks and adverse outcomes of poor care transitions) Provide stories or videos depicting alternative courses of events as told by care staff healthcare workers can identify with 	Leaders of organizations (including management staff), Healthcare professionals
	23.	Organizational diagnosis and feedback	 Assess the various organizational aspects that should be in place and the organization's degree of readiness to implement the TCI (e.g., use surveys, and focus groups) Establish a list of barriers and facilitators in the organization that can influence the implementation of the TCI and how to overcome/leverage them Enter into discussions with organizational leaders on the above and explore solutions 	Organizations

Factor	Strategy (change	method)	Practical applications (suggested)	Target (directed at)
Relative priority	24. 25.	Belief selection & Persuasive communication	Provide information and testimonials about the positive/negative consequences Leaders of organizations (including of implementing/not implementing the TCI (e.g., cases, stories) management staff), Healthcare Share findings of needs assessment (e.g., case report) and describe the priority professionals to implement the TCI within the organization.	Leaders of organizations (including management staff), Healthcare professionals
	26.	Organizational diagnosis and feedback	Conduct a survey to collect perspectives on implementing the TCI Discuss the results with organizational and operational leaders and explore solutions with them	Organizations
Reflecting & evaluating	27.	Audit and feedback	Prepare and perform audits for assessing the quality of transitional care Develop a summary of performance regarding the TCI implementation (key measures on process /outcomes) over a specific period Organize regular implementation team meetings to relay data on performance (provide feedback), reflect on implementation effort (get feedback), and share lessons learned.	Organizations, Healthcare professionals
	28.	Monitoring the performance of the delivery of healthcare	Develop tools/systems for quality monitoring with measures specific to the TCl implemented (e.g., processes, patient outcomes, implementation outcomes) Assess continuously the progress of the TCl implementation and adjust clinical practices when needed	
Targeted groups	29.	Tailoring	Asses the local care & social needs of the target population of older persons Develop an adaptable TCI (clarify the core and modular components) to meet the local needs	Older persons
Transition roles	30. 32.	Structural redesign, Role expansion or task shifting, & Revise professional roles	Map current inter-organizational interactions, work processes, and staff roles and identify changes necessary to accommodate the TCI implementation Expand, shift, or revise job descriptions, roles, and tasks of staff to include implementing the TCI	Organizations, Healthcare professionals

	Stra (cha	Strategy (change method)	Practical applications (suggested)	Target (directed at)
Knowledge, beliefs, and personal attributes	с; с	Active learning	 Conduct interactive lectures, group discussions, and teach-back sessions (e.g., exercises with a workbook, case study solving) about the TCI components and features 	
or nearmcare professionals	34. 35.	Chunking & Advance organizers	 Use labels, color-coding, index in the educational materials about the TCI Structure and write texts in the most accessible way. 	Healthcare professionals
	36. 38. 38.	Shifting perspective, Arguments, & Belief selection	 Challenge organizations and healthcare professionals to take the perspective of an older person facing a care transition and ask them to consider what they would find important Organize meetings with healthcare professionals to introduce the new TCl and deliver convincing arguments and important messages (e.g., evidence on the effects of TCs, need for person-centeredness, etc.) Perform pen discussions with healthcare provide guidance towards adopting the TCl (e.g., integrate new beliefs about the TCl). 	
	39.	Direct experience	 Assign to a care setting where a TCI is implemented to obtain direct observation 	Healthcare professionals
	40.	Guided practice	 Bring experts on innovations in transitional care to model tasks and skills required and to provide ongoing implementation support on-site Bring TCI coaches to support on innovations, transitional care, and implementation 	

they can be adapted and further expanded as needed; Target: individual/entity that will decide to uptake a TCI and put it in practice; Organizations: all long-term care organizations involved in improving care transitions between settings and implementing a TCI; Stakeholders: important individuals who are key and have a great influence on the implementation success, includes healthcare providers, direct implementation staff, leaders, patients/older persons, informal caregivers, family; Healthcare providers, frontline staff who are une strategies, and פוולפ to operation nplementation specialisty; Practical applications: are suggestions on how directly involved in implementing the TCI. TCI=transitional care innovation, LTC=long-term care. histrators, payer, provider, leader, manager, resear care (e.g., aamır

DISCUSSION

Despite the rapidly increasing development and implementation of various innovations in transitional care and healthcare in general, literature highlighted common problems related to selecting and using implementation strategies (16). Limited assessment of implementation factors, insufficient use of a systematic method to develop implementation strategies, and little consideration of relevant theories and evidence in the selection of strategies are key issues that impede the success of implementing innovations, hence TCIs (16, 31, 63). The current project is a novel work that applied Implementation Mapping (31) and developed a set of implementation strategies carefully selected for TCIs. Initially, our findings identified 20 TCIs whereby the majority aimed to improve care transitions and had care continuity including the presence of staff with transition roles as a fundamental component. Consequently, we determined 12 priority factors, mainly linked to the organizational setting, which influence the implementation of TCIs and hence require specific strategies to address them. This culminated in the formulation of a set of various implementation strategies at the organizational, individual, policy, and innovation levels. We systematically selected strategies supported by theory and evidence on their effectiveness in implementing change in healthcare settings. The larger part of the selected implementation strategies aimed at targeting the organizational commitment for change, leadership behaviors and skills, and structural features. In addition, key strategies were selected to enhance the individuals' attitudes, awareness, beliefs, knowledge, and skills to implement TCIs. Fewer strategies were selected at the policy and innovation levels, which could be explained by less implementation factors reported at these levels.

The final selection of implementation strategies is rather exhaustive than earlier projects that developed strategies to implement TCIs and which focused only on one TCI and did not follow a thorough procedure such as Implementation Mapping (10, 26). However, similar to our selection, the majority of the strategies used in these projects were also at the organizational level (i.e. audit and feedback, revise professional roles) (10); whereas strategies used in projects to implement for example health promotion interventions tend to be more at the individual level (64, 65). Furthermore, our selection of strategies corresponds to and expands further on the list of strategies proposed by McArthur et al., designed to improve the implementation of evidence-based guidelines in long-term care and which included education, training, environmental restructuring, persuasion, modeling, and enablement (66).

As for the large number of selected strategies, we opted to propose multiple potential strategies for each factor given the heterogeneity of organizations (e.g., hospital, homecare, nursing home, and transition unit) involved in transitional care. The variety of strategies to choose from, allows for further tailoring to the different settings where TCIs may be implemented.

Further, the selected implementation strategies align with the guidelines on describing and operationalizing strategies recommended by Proctor et al. (17). Though, while we suggested potential actors (i.e. individuals and entities to deliver the strategies), we did not specify the dose and temporality for each strategy. Hence, the dosage (e.g., number of training sessions, frequency of monitoring, or amount of time spent with an expert on implementing innovations) of each strategy should be considered by actors and possibly set beforehand to achieve a certain effect. Likewise, temporality or sequence of strategy use is critical and should be well thought of; e.g. assessing the organizational aspects needed for implementing a TCI should precede any structural redesign or changes in staffing models.

Literature on the empirical evidence on the effectiveness of the strategies was limited. Most studies in general healthcare concentrated on examining the effects of a few strategies such as audit and feedback, educational materials and activities, clinician reminders, opinion leaders, and revising professional roles (27, 49, 67). Moreover, evidence for strategies used to implement innovations specifically in the context of transitional care or long-term care was scarce. It is also important to note that many studies evaluated the effects of multiple implementation strategies used together simultaneously rather than individually, which thus makes it harder to disentangle the relative effect per strategies used in various empirical studies. This resonates with known constraints to interpret implementation efforts, which hinders the ability to comprehend the success and effectiveness of implementation strategies (68).

IMPLICATIONS FOR PRACTICE AND RESEARCH

We emphasize again that the use of implementation strategies is a focal task in the prospective implementation of any TCI in practice. Hereby we suggest to future implementers to first perform an analysis of the setting at hand, and to consider the priority factors we identified as most likely present in their respective context. Second, to utilize the relevant implementation strategies resulting from this project as both a backbone and starting point for implementing a TCI. Further deliberations among implementers of a specific TCI are also recommended and may be necessary to choose from the strategies which are feasible, applicable, and perceived as important in relevance to their particular context.

This work merits further research and notably empirical studies to assess the effectiveness of the implementation strategies in the context of transitional care. Other suggested investigations may include assessing the use of different combinations of strategies as well as if a potential hierarchy of importance or optimal sequencing can maximize effects. This will help add to the currently limited evidence on implementation strategies.

STRENGTHS AND LIMITATIONS

To our knowledge, this project is the first to apply Implementation Mapping as recommended by Fernandez et al. (31) to improve specifically the implementation of TCIs in long-term care settings. Therefore, our approach was valuable to advance efforts to better implement innovations in transitional care. This method allowed us to identify and understand factors thoroughly and then to use theory and evidence to deliberately select strategies with specific mechanisms adept to bring about the desired change in the behaviors of individuals or organizations (69). Furthermore, we performed deliberate preparatory work for this project beforehand by conducting a series of research studies on the implementation of TCIs. Lastly, we acknowledge that the selected strategies can be similar to strategies used for implementing other types of innovations in long-term care. Yet, our strategies thoroughly selected to address factors that influence specifically the implementation of TCIs can be more effective than strategies not specially selected for the implementation of TCIs (70).

On the other hand, we note some limitations of this project. First, the views and interpretations of the research team might have affected certain steps of the Implementation Mapping process. This includes the final consideration and judgment of whether to select a strategy or not. However, we tried to limit subjectivity by obtaining different perspectives from experts in the fields of implementation science and transitional care. Second, we performed a rapid scan of empirical evidence for the strategies rather than a systematic literature search. Therefore we might have missed some studies on evidence for certain strategies, yet we used a flexible terminology/description for strategies in our search and explored multiple databases. However, it is important to highlight that we experienced a literature gap in empirical evidence for implementation strategies, which might have influenced our final selection of strategies.

CONCLUSIONS

The implementation of TCIs in practice is complex and challenged by multiple factors, particularly at the intra and inter-organizational levels. The use of theory-driven and effective implementation strategies carefully selected to address the relevant factors is highly needed to better implement TCIs. The current project provides an extensive set of implementation strategies for this purpose. We strive for this work to be utilized by future implementers of TCIs in long-term care. As well, that it will inspire other researchers to use a similar approach for prospective efforts aiming to improve the implementation of diverse innovations in healthcare.

REFERENCES

- Naylor MD, Aiken LH, Kurtzman ET, Olds DM, Hirschman KB. The care span: The importance of transitional care in achieving health reform. Health Aff (Millwood). 2011;30(4):746-54.
- Coleman EA. Falling through the cracks: challenges and opportunities for improving transitional care for persons with continuous complex care needs. J Am Geriatr Soc. 2003;51(4):549-55.
- 3. Fakha A, Groenvynck L, de Boer B, van Achterberg T, Hamers J, Verbeek H. A myriad of factors influencing the implementation of transitional care innovations: a scoping review. Implement Sci. 2021;16(1):21.
- Fønss Rasmussen L, Grode LB, Lange J, Barat I, Gregersen M. Impact of transitional care interventions on hospital readmissions in older medical patients: a systematic review. BMJ Open. 2021;11(1):e040057.
- 5. Enderlin CA, McLeskey N, Rooker JL, Steinhauser C, D'Avolio D, Gusewelle R, et al. Review of current conceptual models and frameworks to guide transitions of care in older adults. Geriatr Nurs. 2013;34(1):47-52.
- 6. Groenvynck L, Fakha A, de Boer B, Hamers JPH, van Achterberg T, van Rossum E, et al. Interventions to improve the transition from home to a nursing home: a scoping review. Gerontologist. 2021.
- Lutz BJ, Reimold AE, Coleman SW, Guzik AK, Russell LP, Radman MD, et al. Implementation of a transitional care model for stroke: perspectives from frontline clinicians, administrators, and COMPASS-TC implementation staff. Gerontologist. 2020;60(6):1071-84.
- Lee JY, Yang YS, Cho E. Transitional care from hospital to home for frail older adults: a systematic review and meta-analysis. Geriatr Nurs. 2022;43:64-76.
- Hirschman KB, Hodgson NA. Evidence-based interventions for transitions in care for individuals living with dementia. Gerontologist. 2018;58(suppl_1):S129-S40.
- Yip O, Huber E, Stenz S, Zullig LL, Zeller A, De Geest SM, et al. A contextual analysis and logic model for integrated care for frail older adults living at home: the INSPIRE project. Int J Integr Care. 2021;21(2):9.
- Dresden SM, Hwang U, Garrido MM, Sze J, Kang R, Vargas-Torres C, et al. Geriatric emergency department innovations: the impact of transitional care nurses on 30-day readmissions for older adults. Acad Emerg Med. 2020;27(1):43-53.
- Bryant-Lukosius D, Carter N, Reid K, Donald F, Martin-Misener R, Kilpatrick K, et al. The clinical effectiveness and cost-effectiveness of clinical nurse specialist-led hospital to home transitional care: a systematic review. J Eval Clin Pract. 2015;21(5):763-81.
- 13. Hwang U, Dresden SM, Rosenberg MS, Garrido MM, Loo G, Sze J, et al. Geriatric emergency department innovations: transitional care nurses and hospital use. J Am Geriatr Soc. 2018;66(3):459-66.
- Eldh AC, Almost J, DeCorby-Watson K, Gifford W, Harvey G, Hasson H, et al. Clinical interventions, implementation interventions, and the potential greyness in between -a discussion paper. BMC Health Serv Res. 2017;17(1):16.
- Fakha A, de Boer B, van Achterberg T, Hamers J, Verbeek H. Fostering the implementation of transitional care innovations for older persons: prioritizing the influencing key factors using a modified Delphi technique. BMC Geriatr. 2022;22(1):131.
- Powell BJ, Fernandez ME, Williams NJ, Aarons GA, Beidas RS, Lewis CC, et al. Enhancing the impact of implementation strategies in healthcare: a research agenda. Front Public Health. 2019;7:3.
- Proctor EK, Powell BJ, McMillen JC. Implementation strategies: recommendations for specifying and reporting. Implement Sci. 2013;8(1):139.
- Kirchner JE, Smith JL, Powell BJ, Waltz TJ, Proctor EK. Getting a clinical innovation into practice: An introduction to implementation strategies. Psychiatry Res. 2020;283:112467.
- 19. Cochrane. Effective Practice and Organisation of Care (EPOC). EPOC Taxonomy; 2015 [Available from: epoc. cochrane.org/epoc-taxonomy]
- Powell BJ, Waltz TJ, Chinman MJ, Damschroder LJ, Smith JL, Matthieu MM, et al. A refined compilation of implementation strategies: results from the Expert Recommendations for Implementing Change (ERIC) project. Implement Sci. 2015;10:21.
- 21. Mazza D, Bairstow P, Buchan H, Chakraborty SP, Van Hecke O, Grech C, et al. Refining a taxonomy for guideline implementation: results of an exercise in abstract classification. Implement Sci. 2013;8:32.
- 22. Michie S, Richardson M, Johnston M, Abraham C, Francis J, Hardeman W, et al. The behavior change technique taxonomy (v1) of 93 hierarchically clustered techniques: building an international consensus for the reporting of behavior change interventions. Ann Behav Med. 2013;46(1):81-95.
- Kok G, Gottlieb NH, Peters G-JY, Mullen PD, Parcel GS, Ruiter RAC, et al. A taxonomy of behaviour change methods: an Intervention Mapping approach. Health Psychol Rev. 2016;10(3):297-312.
- 24. Brody AA, Arbaje AI, DeCherrie LV, Federman AD, Leff B, Siu AL. Starting up a hospital at home program: facilitators and barriers to implementation. J Am Geriatr Soc. 2019;67(3):588-95.
- Everink IH, van Haastregt JC, Maessen JM, Schols JM, Kempen GI. Process evaluation of an integrated care pathway in geriatric rehabilitation for people with complex health problems. BMC Health Serv Res. 2017;17(1):34.
- Toles M, Frerichs A, Leeman J. Implementing transitional care in skilled nursing facilities: evaluation of a learning collaborative. Geriatr Nurs. 2021;42(4):863-8.

- Grimshaw JM, Eccles MP, Lavis JN, Hill SJ, Squires JE. Knowledge translation of research findings. Implement Sci. 2012;7:50.
- Bartholomew LK, Parcel GS, Kok G, Gottlieb NH, Fernández ME. Planning health promotion programs: an Intervention Mapping approach: Jossey-Bass; 2011.
- 29. Waltz TJ, Powell BJ, Matthieu MM, Damschroder LJ, Chinman MJ, Smith JL, et al. Use of concept mapping to characterize relationships among implementation strategies and assess their feasibility and importance: results from the Expert Recommendations for Implementing Change (ERIC) study. Implement Sci. 2015;10:109.
- 30. Cochrane. Effective Practice and Organisation of Care (EPOC). EPOC Taxonomy. 20152015 [Available from: epoc.cochrane.org/epoc-taxonomy.
- 31. Fernandez ME, ten Hoor GA, van Lieshout S, Rodriguez SA, Beidas RS, Parcel G, et al. Implementation mapping: using intervention mapping to develop implementation strategies. Front Public Health. 2019;7.
- Ogrinc G, Davies L, Goodman D, Batalden P, Davidoff F, Stevens D. SQUIRE 2.0 (Standards for QUality Improvement Reporting Excellence): revised publication guidelines from a detailed consensus process. BMJ Qual Saf. 2016;25(12):986-92.
- Naylor MD, Shaid EC, Carpenter D, Gass B, Levine C, Li J, et al. Components of comprehensive and effective transitional care. J Am Geriatr Soc. 2017;65(6):1119-25.
- 34. Fakha A, Leithaus M, de Boer B, van Achterberg T, Hamers JP, Verbeek H. Implementing four transitional care interventions for older adults: a retrospective collective case study. Gerontologist. 2022.
- Damschroder LJ, Aron DC, Keith RE, Kirsh SR, Alexander JA, Lowery JC. Fostering implementation of health services research findings into practice: a consolidated framework for advancing implementation science. Implement Sci. 2009;4(1):50.
- Michie S, Johnston M, Abraham C, Lawton R, Parker D, Walker A. Making psychological theory useful for implementing evidence based practice: a consensus approach. Qual Saf Health Care. 2005;14(1):26-33.
- Powell BJ, Waltz TJ, Chinman MJ, Damschroder LJ, Smith JL, Matthieu MM, et al. A refined compilation of implementation strategies: results from the Expert Recommendations for Implementing Change (ERIC) project. Implement Sci. 2015;10(1):21.
- CFIR Research Team Center for Clinical Management Research . CFIR-ERIC matching tool [Internet]. 2022. Available from: https://cfirguide.org/choosing-strategies/.
- Curran GM, Bauer M, Mittman B, Pyne JM, Stetler C. Effectiveness-implementation hybrid designs: combining elements of clinical effectiveness and implementation research to enhance public health impact. Med Care. 2012;50(3):217-26.
- 40. Weiner BJ. A theory of organizational readiness for change. Implement Sci. 2009;4(1):67.
- Cheng H, McGovern MP, Garneau HC, Hurley B, Fisher T, Copeland M, et al. Expanding access to medications for opioid use disorder in primary care clinics: an evaluation of common implementation strategies and outcomes. Implement Sci Commun. 2022;3(1):72.
- 42. Walunas TL, Ye J, Bannon J, Wang A, Kho AN, Smith JD, et al. Does coaching matter? Examining the impact of specific practice facilitation strategies on implementation of quality improvement interventions in the Healthy Hearts in the Heartland study. Implement Sci. 2021;16(1):33.
- Rogal SS, Yakovchenko V, Waltz TJ, Powell BJ, Kirchner JE, Proctor EK, et al. The association between implementation strategy use and the uptake of hepatitis C treatment in a national sample. Implement Sci. 2017;12(1):60.
- Rogal SS, Yakovchenko V, Waltz TJ, Powell BJ, Gonzalez R, Park A, et al. Longitudinal assessment of the association between implementation strategy use and the uptake of hepatitis C treatment: year 2. Implement Sci. 2019;14(1):36.
- 45. Köpke S, Mühlhauser I, Gerlach A, Haut A, Haastert B, Möhler R, et al. Effect of a guideline-based multicomponent intervention on use of physical restraints in nursing homes: a randomized controlled trial. Jama. 2012;307(20):2177-84.
- 46. Beeckman D, Clays E, Van Hecke A, Vanderwee K, Schoonhoven L, Verhaeghe S. A multi-faceted tailored strategy to implement an electronic clinical decision support system for pressure ulcer prevention in nursing homes: a two-armed randomized controlled trial. Int J Nurs Stud. 2013;50(4):475-86.
- Basinska K, Zúñiga F, Simon M, De Geest S, Guerbaai RA, Wellens NIH, et al. Implementation of a complex intervention to reduce hospitalizations from nursing homes: a mixed-method evaluation of implementation processes and outcomes. BMC Geriatr. 2022;22(1):196.
- Birken SA, Lee SY, Weiner BJ, Chin MH, Chiu M, Schaefer CT. From strategy to action: how top managers' support increases middle managers' commitment to innovation implementation in health care organizations. Health Care Manage Rev. 2015;40(2):159-68.
- Lau R, Stevenson F, Ong BN, Dziedzic K, Treweek S, Eldridge S, et al. Achieving change in primary care effectiveness of strategies for improving implementation of complex interventions: systematic review of reviews. BMJ Open. 2015;5(12):e009993.
- Ivers N, Jamtvedt G, Flottorp S, Young JM, Odgaard-Jensen J, French SD, et al. Audit and feedback: effects on professional practice and healthcare outcomes. Cochrane Database Syst Rev. 2012(6).
- 51. Giguère A, Zomahoun HT, Carmichael PH, Uwizeye CB, Légaré F, Grimshaw JM, et al. Printed educational

materials: effects on professional practice and healthcare outcomes. Cochrane Database Syst Rev. 2020(8).

- Forsetlund L, O'Brien MA, Forsén L, Mwai L, Reinar LM, Okwen MP, et al. Continuing education meetings and workshops: effects on professional practice and healthcare outcomes. Cochrane Database Syst Rev. 2021(9).
- Kirchner JE, Ritchie MJ, Pitcock JA, Parker LE, Curran GM, Fortney JC. Outcomes of a partnered facilitation strategy to implement primary care–mental health. J Gen Intern Med. 2014;29(4):904-12.
- 54. Greene J, Hibbard JH. Why does patient activation matter? An examination of the relationships between patient activation and health-related outcomes. J Gen Intern Med. 2012;27(5):520-6.
- 55. Santos WJ, Graham ID, Lalonde M, Demery Varin M, Squires JE. The effectiveness of champions in implementing innovations in health care: a systematic review. Implement Sci Commun. 2022;3(1):80.
- Flodgren G, O'Brien MA, Parmelli E, Grimshaw JM. Local opinion leaders: effects on professional practice and healthcare outcomes. Cochrane Database Syst Rev. 2019(6).
- 57. Lyon AR, Koerner K. User-Centered Design for Psychosocial Intervention Development and Implementation. Clin Psychol (New York). 2016;23(2):180-200.
- Dugstad J, Eide T, Nilsen ER, Eide H. Towards successful digital transformation through co-creation: a longitudinal study of a four-year implementation of digital monitoring technology in residential care for persons with dementia. BMC Health Serv Res. 2019;19(1):366.
- Aarons GA, Ehrhart MG, Farahnak LR, Hurlburt MS. Leadership and organizational change for implementation (LOCI): a randomized mixed method pilot study of a leadership and organization development intervention for evidence-based practice implementation. Implement Sci. 2015;10(1):11.
- 60. Bandura A. Social foundations of thought and action: a social cognitive theory. Englewood Cliffs, N.J: Prentice-Hall; 1986.
- 61. Grilli R, Ramsay C, Minozzi S. Mass media interventions: effects on health services utilisation. Cochrane Database Syst Rev. 2002(1):Cd000389.
- 62. Birken SA, Bunger AC, Powell BJ, Turner K, Clary AS, Klaman SL, et al. Organizational theory for dissemination and implementation research. Implement Sci. 2017;12(1):62.
- Davies P, Walker AE, Grimshaw JM. A systematic review of the use of theory in the design of guideline dissemination and implementation strategies and interpretation of the results of rigorous evaluations. Implement Sci. 2010;5(1):14.
- Highfield L, Valerio MA, Fernandez ME, Eldridge-Bartholomew LK. Development of an implementation intervention using intervention mapping to increase mammography among low income women. Front Public Health. 2018;6:300.
- 65. Ibekwe LN, Walker TJ, Ebunlomo E, Ricks KB, Prasad S, Savas LS, et al. Using implementation mapping to develop implementation strategies for the delivery of a cancer prevention and control phone navigation program: a collaboration with 2-1-1. Health Promot Pract. 2022;23(1):86-97.
- 66. McArthur C, Bai Y, Hewston P, Giangregorio L, Straus S, Papaioannou A. Barriers and facilitators to implementing evidence-based guidelines in long-term care: a qualitative evidence synthesis. Implement Sci. 2021;16(1):70.
- 67. Spoon D, Rietbergen T, Huis A, Heinen M, van Dijk M, van Bodegom-Vos L, et al. Implementation strategies used to implement nursing guidelines in daily practice: a systematic review. Int J Nurs Stud. 2020;111:103748.
- Bunger AC, Powell BJ, Robertson HA, MacDowell H, Birken SA, Shea C. Tracking implementation strategies: a description of a practical approach and early findings. Health Res Policy Syst. 2017;15(1):15.
- 69. Lewis CC, Boyd MR, Walsh-Bailey C, Lyon AR, Beidas R, Mittman B, et al. A systematic review of empirical studies examining mechanisms of implementation in health. Implement Sci. 2020;15(1):21.
- Baker R, Camosso-Stefinovic J, Gillies C, Shaw EJ, Cheater F, Flottorp S, et al. Tailored interventions to address determinants of practice. Cochrane Database Syst Rev. 2015;2015(4):Cd005470.

APPENDICES

Appendix 6A. Matrices of change

Appendix 6A. Matrices of change

Matrix 1					
		Determinants			
TCIs' use Outcomes (Fakha et al. 2021*)	Performance objectives	Commitment	Organizational commitment	Leadership	Attitudes, beliefs, & motivation
 Ensure safe & timely care transitions Ensure provision of continuous transitional care services Reduce avoidable care transitions (e.g. hospital readmissions, institutional care) Improve communication and information transfer among care providers Reduce medical errors and mortality Ensure patient safety Improve patient satisfaction & quality of life 	Actors I: Leaders and organizations. PO1. Ensure complexity management is performed as a TCI core component such as: • Decide to develop individualized care plans • Decide to perform advanced care planning & comprehensive patient assessment • Support a goal- orientated and individualized delivery of transitional care • Set up a patient- centered holistic approach PO2. Deliver care continuity as a TCI core component such as: • Allocate for and establish staff with a transition role (e.g. advanced practice nurse, transitional care manager, health coach, care coordinator, transition care nurse) • Ensure care continuity tasks are performed by HCP (e.g. home visits, telephone follow-up, discharge planning, coordination among multidisciplinary teams, exchange of patient information, case management, finalization of long-term care arrangements)	C.1: Recognize that a TCI is important, effective, and needed C.2: Demonstrate the ability to ensure the start of and an ongoing TCI implementation	OC.1: Illustrate that TCI fits with organizational structure and goals OC.2: Express that a supportive organizational environment is important for the implementation of a TCI OC.3: Appraise that implementing a TCI is a necessary and crucial change within an organization	L1: Demonstrate an open- minded, innovative leadership style receptive to new ideas	ABM.1: Believe that engaging the key individuals is necessary for implementing a TCI ABM.2: Recognize the high benefit c implementing a TCI
Appendix 6A. Continued.

Determinants							
Environmental conditions (structural, organizational) & professional role	Social networks	Policy	Risk perception	Feedback, reinforcement& monitoring	Innovation's compatibility		
EC.1: Set up health IT systems and platforms that will allow information exchange among care settings EC.2: Arrange for a certain level of resources required to implement the TCI EC.3: Create the knowledge & information platforms and channels about the TCI to facilitate the implementation EC.4: Create and organize transition roles to facilitate the TCI implementation	SN.1: Formulate work relationships and networks between the healthcare providers in different settings in order to facilitate the TCI implementation	P1: Propose policies supportive to implement the TCI	RP1: Predict that implementing a TCI eliminates adverse risks associated with poor transitional care delivery	FRM.1: Estimate that providing feedback on a TCI implementation will enable making revisions and refinements FRM.2: Estimate that providing factual data on the TCI's benefit will support the implementation	IC.1: Recognize that designing the TCI components to match the profile of the older persons will ensure its successful implementation IC.2: Appraise the advantage of implementir a tailored TCI will improve th match betwee older persons' values and car		

Appendix 6A. Continued.

Matrix 2							
Determinants							
TCls' use Outcomes (ref, Fakha et al. 2021)	Performance objectives	Knowledge	Attitudes & beliefs	Motivation	Skills		
 Ensure safe & timely care transitions Ensure provision of continuous transitional care services Reduce avoidable care transitions (e.g. hospital readmissions, institutional care) Improve communication and information transfer among care providers Reduce medical errors and mortality Ensure patient safety Improve patient satisfaction & quality of life 	Actors II: Healthcare professionals PO1. Engage the patient & caregiver as TCI core components such as: • Establish trusting relationships with patient and caregiver • Understand patient's goals & preferences • Enable collaboration between patient, family caregivers, and primary care providers PO2. Educate the patient & caregiver as TCI core components such as: • Conduct discharge planning using "teach-back" methods with patient • Coach the patient in: • medication self- management • using a patient- centered health record • knowledge of "red flags" • making primary care provider/ specialist appointments • Provide patient/ caregiver education tools • Coordinate education & community services to develop self- management skills of the patient	K.1: Describe the components of the TCI K.2: Recall the tasks, steps, & procedures required to implement the TCI	AB1: Express positive attitude towards the TCI as an innovation AB2: Believe that the TCI is beneficial to enhance transitional care for older persons	M1: Express interest and intention to implement the TCI	S1: Demonstrate the ability to deliver the components of the TCI S2: Demonstrate the ability to navigate and adhere to the TCI's implementation tasks		

Matrix 2									
Determinants									
TCIs' use Outcomes (ref, Fakha et al. 2021)	Performance objectives	Knowledge	Attitudes & beliefs	Motivation	Skills				
	 PO3. Provide complexity management as a TCI core component such as: Develop individualized care plans Carry out advanced care planning & comprehensive patient assessment Apply a patient- centered holistic approach PO4. Provide care continuity as a TCI core component such as: Execute the duties of a transition role professional Conduct care continuity tasks (e.g. home visits, telephone follow-up, discharge planning, coordination among multidisciplinary teams, exchange of patient information, case management, finalization of long-term care 								

Appendix 6A. Continued.

*Reference: Fakha A, Groenvynck L, de Boer B, van Achterberg T, Hamers J, Verbeek H. A myriad of factors influencing the implementation of transitional care innovations: a scoping review. Implement Sci. 2021;16(1):21. TCIs = transitional care innovations.

arrangements)



CHAPTER

General Discussion

Transitional care innovations (TCIs) are emerging innovative solutions designed to enhance care continuity and coordination for older persons when transferring between multiple healthcare settings. However, the implementation of TCIs into practice settings is challenging. The aim of this dissertation was to explore what influences the implementation of TCIs and how to improve it. In this chapter, the main findings of this dissertation are discussed, and a reflection on methodological and theoretical considerations is presented. Furthermore, recommendations for practice and future research are provided.

MAIN FINDINGS

In performing a scoping review of the literature, we identified numerous types of TCIs that were developed and implemented to enhance care transitions for older persons, the majority of which focused on improving transitions from hospital to home settings. In addition, the review determined several factors (barriers, facilitators) that influenced the implementation of TCIs at multiple domains. Notable barriers were linked to the overall organizational readiness to implement TCIs, and key facilitators were related to the innovation's characteristics and the implementation process. Furthermore, in our Delphi study, experts conceded that organizational leadership, engaged key stakeholders, continuous information exchange across care settings, and financing of TCIs' implementation are top priority factors and have the most important influence on the implementation of TCIs. Moreover, there was a consensus on the presence of interrelationships among these factors.

A real-life case, studying the implementation of four TCIs in Belgium showed the intuitive manner commonly present in implementing transitional care innovations in practice. Here, findings similar to those of the previous studies were found on the main hindering influence of organizational factors. On the other hand, the presence of highly committed individuals played a key facilitating role in the implementation of the four TCIs. The engagement of key persons was seen as a significant strategy in the implementation of the four TCIs. Moreover, positive experiences of stakeholders involved in implementing the four TCIs were captured. Autonomous work motivation and the ability to develop multidisciplinary care partnerships helped stakeholders to communicate better and facilitated the implementation of the TCIs. Thus, the studies in this dissertation showed that the implementation of multiple TCIs is influenced by many factors. Nevertheless, these factors are not always considered. Therefore, forty implementation strategies were selected to improve the prospective implementation of TCIs. These strategies aim to address key influencing factors at the organizational, individual, policy, and innovation levels.

METHODOLOGICAL CONSIDERATIONS

The methodological considerations regarding individual studies in this dissertation have been discussed in the previous chapters. This chapter presents general considerations regarding the use of qualitative methods, the implementation framework - Consolidated Framework for Implementation Research (CFIR) - throughout the studies, and the involvement of healthcare professionals and older persons in implementation research.

QUALITATIVE METHODS

The studies in this dissertation comprised a combination of diverse qualitative research methods and followed specific tools and frameworks selected from implementation science, defined as the scientific study of methods to promote the systematic uptake of research findings and other evidence-based practices into routine practice to improve the quality and effectiveness of health services and care (1). Our choice in using qualitative methods coincides with the high relevance and importance of such methods in conducting implementation research specifically (2). Qualitative methods are integral in describing what happens in implementation, uncovering contextual elements, and capturing the individuals' perceptions that can affect implementation (3). Hence, it helped to explain the complexity of implementing TCIs as well as to explore how and why implementation succeeds or fails (4). Furthermore, while each of the various studies conducted, (chapters 2 to 6) served a certain function in meeting the overall objectives of this dissertation; a key strength was the sequential use of findings from each study into the next one. The buildup of findings led to a solid and well-rounded understanding of the implementation of TCIs.

IMPLEMENTATION FRAMEWORK

A common methodological element across all the studies was the use of the CFIR framework. This implementation science framework provides a comprehensive listing of constructs (i.e. factors) thought to influence implementation and assists to streamline the research process (5). Hence, its use constituted a strength and provided an efficient way to clarify the larger context surrounding the TCIs, helped identify key factors that may hinder or facilitate the implementation, and indicated potential relationships among the factors (6). Moreover, CFIR tools were instrumental in the data collection, analysis, and reporting of results for all studies performed (7).

However, a few limitations of the CFIR pertaining to some constructs and domains must be acknowledged. Overall, the CFIR constructs touch minimally on the patients/older persons and their informal/family caregivers (i.e. who are the key target group of an innovation) and are less centered around reflecting their perspectives and experiences (8). Moreover, it combines all constructs related to characteristics of individuals in one domain and without a clear distinction of roles of individuals as being for example either a patient, informal

CHAPTER 7

caregiver, healthcare professional, implementation leader, or implementation team member. Hence, the construct 'knowledge and beliefs about the intervention' can for instance be about the knowledge and beliefs of the healthcare professional or that of the patient. This might have impacted our results by capturing less of the perspectives of the patients/older persons and their informal/family caregivers, which could have informed us more about their priorities, preferences, and needs to guide a better implementation of TCIs. However, this limitation was minimized by dividing the two constructs 'knowledge and beliefs about the intervention', and 'other personal attributes' into that of a) healthcare professionals or b) patient/older persons, and then mapping the collected data accordingly (chapters 3, 4). Alternatively, tools such as the Tailored Implementation for Chronic Diseases (TICD) checklist of determinants of practice could have identified more specifically these factors with its separate domains on individual health professional factors and patient factors (9).

Furthermore, the CFIR presents the two domains outer and inner settings more as silos featuring little constructs to highlight the interactions between one organization and its external environment (e.g., another care organization, health authority, policymakers, funding agencies, patient advocacy groups, community and social services). Particularly, these interconnections such as partnerships and collaborations among organizations are common and crucial in delivering transitional care and specifically in implementing TCIs across multiple settings (10). Accordingly, this dissertation selected and combined constructs from the Care Transitions Framework (CTF) to the CFIR such as 'information continuity' across different care settings, and 'transition roles' of frontline staff (11). This provided further understanding of barriers and facilitators to the implementation of TCIs through a robust consideration of transitional care and inter-organizational interactions. Nevertheless, other frameworks exist that might have better captured these interactions such as the Exploration, Preparatory, Implementation, and Sustainment (EPIS) framework. This focuses more on the nature of interconnections between inner-outer contexts with its domain on interconnections (10). However, similar interconnections were captured under the CFIR/CTF constructs of cosmopolitanism, and engaging organizations and external context.

INVOLVEMENT OF HEALTHCARE PROFESSIONALS AND OLDER PERSONS IN IMPLEMENTATION RESEARCH

In this dissertation, the involvement of healthcare professionals as to obtain their perspectives on implementing TCIs had its limitations. The retrospective collective case study (chapter 4) was the only study conducted with healthcare professionals exclusively as participants. Given the complex and dominant theoretical aspects of conducting implementation science research, there was an inclination throughout the studies to include more scientific experts and researchers from this area to explore the implementation of TCIs. Specifically, in the Delphi study (chapter 3), the panel of experts' task was to assimilate the meanings, judge,

GENERAL DISCUSSION

and prioritize the factors that influence the implementation of TCIs, and this required strong scientific knowledge about implementation. This presented a difficulty regarding the inclusion of more healthcare professionals as participants (12). Hence, while this added to the study's strength of obtaining high-quality consensus, we potentially gave more voice to scientific experts than to the professionals. Therefore, capturing further the point of view of healthcare professionals, who are often at the frontline of delivering transitional care, might have added further insights on implementing TCIs.

Another challenge regarding the involvement of healthcare professionals relates to the development of a set of implementation strategies selected for implementing TCIs (chapter 6). Although our application of Implementation Mapping and including a group of experts (some of which had direct experience in patient care) was a form of participatory design approach (13), it would have been valuable to learn more from the perspectives of healthcare professionals who will ultimately use the TCIs (14). Conducting interviews or focus group discussions with healthcare professionals to obtain reflective feedback on the set of selected implementation strategies might have added to our participatory approach (15). Furthermore, it could have helped to shift our dependency on the theoretical effects of the implementation strategies and could have added more insights on their compatibility, feasibility, and/or affordability in practice settings (16). However, while the importance of practice involvement is acknowledged, the implementation strategies were intended to be used for different types of TCIs and care settings. Hence, we did not focus on one specific setting or TCI, nor had a defined group of healthcare professionals to co-develop the strategies with them. Instead, we opted for experts knowledgeable about the Implementation Mapping technique and with experience in implementing healthcare innovations.

The involvement of older persons and their informal/family caregivers to obtain their viewpoints on the implementation of TCIs was limited in this dissertation. Though patient involvement is important and recommended in health research (17), it is challenging and unclear how to do so specifically in implementation research, which aims to comprehend and modify healthcare professionals' and organizations' behaviors rather than the patients' behaviors (18). Hence, there is little guidance on acquiring the reflections of patients on implementation efforts. In this regard, our studies focused on investigating the overall implementation scope of TCIs mainly from the perspectives of scientific experts, organizations, and healthcare professionals as they could provide input on various aspects of TCIs' implementation. Nonetheless, in the scoping review (chapter 2) we tried to retrieve the perception of older persons and/or their families and informal caregivers regarding the implementation of TCIs. However, we saw that capturing the older persons' perspectives on the implementation of TCIs (i.e. challenges, factors, and process) is limited in literature (only 6 out of 21 studies included the patient perspective), and if present, it focused on their overall satisfaction and experiences with the innovation. Moreover, in the retrospective collective case study (chapter 4) we considered

225

the importance of involving the older persons and hearing their perspectives on the implementation of the four TCIs in Belgium. However, this was not feasible due to challenges in data collection during Covid-19 pandemic restrictions as well as constraints in locating and sampling older persons with a retrospective case study design. This corresponds to similar difficulties indicated in the literature on involving older persons (19).

THEORETICAL CONSIDERATIONS

In this dissertation, organizational factors belonging to the inner context were found to influence significantly the implementation of TCIs. Moreover, due to the nature of transitional care, our findings highlighted key influencing factors that described the interconnections between the organizational (inner) and external (outer) contexts. Accounting for these linking factors helped explain the complexity of implementing TCIs involving various entities (i.e. organizations, service systems, and regulatory agencies) as well as offered avenues for improving the implementation process. This section provides theoretical considerations regarding first, the interconnections between inner-outer contexts in implementing TCIs, and second, alternative strategies to improve the implementation of TCIs.

INTERCONNECTIONS BETWEEN INNER-OUTER CONTEXTS IN IMPLEMENTING TCIS

Our findings depicted the inevitable dynamics that occur between the inner-outer contexts in implementing TCIs. Hence, some of the identified factors gave the opportunity to understand the implementation of TCIs from an open systems perspective by looking beyond the borders of one organization (20). The open systems concept denotes that organizations are permeable, and have bi-directional interactions with their external context. Along this line of thought, participants in our studies considered the existing financing structures, reimbursement systems of healthcare services, and governmental healthcare policies in the external context of an organization as very important factors and powerful enough to hinder or enable the organization's ability to implement a TCI. Similarly, the presence of staff with a designated transition role in an organization was determined as a key facilitator to implementing a TCI. Individuals with transition roles behave as intermediaries and can implement the core components of the innovation by linking the organization with the external context (21).

Moreover, our results feed into the concept of bridging factors as critical to the implementation of innovations (22). Bridging factors are defined as "factors that cross or link the outer system and inner organizational context" (21). Traditionally, there has been more focus on examining the inner context and outer contexts as separate entities, and less on investigating bridging factors in implementation. Therefore, our work coincides with the evolving need in implementation science to identify bridging factors as they reveal how organizations interact with the external context and its impact on implementing innovations (22, 23).

GENERAL DISCUSSION

Specifically, our results showed the influence of interactions between two or more organizations on the implementation of TCIs. These captured interactions, which match with the concept of bridging factors, included inter-organizational collaborations, preexisting partnerships (e.g. between hospital and homecare organizations), sharing of practices among various healthcare organizations, promoting external collaborations with resourceful organizations (e.g., community services agencies), and information continuity (e.g., exchanging patient information between organizations). Similar types of interactions were also indicated in the literature on delineating the functions and forms that bridging factors could take such as relational ties, formal arrangements, or data-sharing processes between organizations (23).

Finally, inter-organizational interactions can also be reflected on from an organizational theory perspective. In general, organizations develop external relationships to achieve goals, improve performance, create powerful allies and diverse networks, and gain access to resources (24). Across this dissertation, the lack of organizational resources was frequently reported as a main barrier to implement TCIs, and simultaneously, engagement between organizations was described as a key factor. However, with our results, we can only assume a potential association between the lack of organizational resources and an organization's tendency to engage with other external organizations to implement a TCI. This could be explained also by the relevance of resource dependency theory to transitional care, which denotes that organizations establish relationships with other organizations to secure resources needed to implement innovations, such as TCIs (25).

ALTERNATIVE STRATEGIES - SMART CONNECTIONS OR CHOICES TO IMPLEMENT TCIs?

The use of implementation strategies selected to address the specific factors that influence the implementation of TCIs is critical. Particularly, in transitional care, strategies to address factors related to inter-organizational interactions are needed. The set of implementation strategies developed in chapter 6 included various strategies for spanning the boundaries of one organizational context and collaborating with multiple organizations in the implementation of TCIs (e.g., building a coalition, enhancing network linkages, developing resource-sharing agreements, and creating learning collaboratives). These strategies suggest the creation of inter-organizational alliances, which is supported by literature on the advantages of creating strategic alliances and cooperative partnerships between organizations to improve the organization's operations and innovativeness (26, 27). Thus, there is a large potential for strategic alliances, but also there is less insight into the key elements of such alliances and their value in the context of transitional care.

Therefore, the proposed implementation strategies can be reflected on from the relational view theory perspective to illustrate how alliances and value-creating relationships between

CHAPTER 7

organizations are formed. According to this theory, superior advantages are created when organizations in an alliance combine and exchange their peculiar assets, knowledge, and complementary resources/capabilities, and employ effective governance (28). This implies that if a hospital is implementing a TCI to improve care transitions from hospital to home settings within a certain region, forming an alliance with physically proximate homecare organizations is an asset. Over time, the two organizations can develop experience in working together and accumulate a specific 'know-how' on coordinating care transitions for older persons in that region. Hence, this permits a more effective and efficient interorganizational communication that further promotes the TCI's implementation. Likewise, creating inter-organizational knowledge-sharing routines to transfer or combine specialized knowledge is a valuable element of alliances. For example, an alliance between multiple nursing home facilities can promote information sharing and collaborative learning on transitional care and help implement a TCI to avoid unnecessary transitions of older persons to hospital. Moreover, organizations in an alliance can boost their ability to implement a TCI by combining their distinctive resources in a complementary way. Distinctive resources relevant to TCIs can include a care transition nurse or community resources.

Another important aspect to consider is that alliances are dynamic and affected by factors, such as a decrease in resource complementarity among organizations, which can lead to a decline in their value over time (29). Therefore, organizations involved in implementing TCIs need to be diligent in their choice of partners. They should form an alliance with a strategic partner that has the specific capabilities needed to enable the TCIs' implementation and assess the value of the alliance continuously (30). In conclusion, relational exchanges between organizations are instrumental in implementing TCIs. Thus crossing the boundaries of one organizational context is at the core of these innovations, making them unique and challenging.

On the other hand, in this dissertation, additional implementation strategies were selected to address factors at the individual level. Yet, most of the strategies we proposed to change the individuals' behavior focused on the conscious cognitive processing. Indeed, our results indicate that implementing TCIs is determined more by the individuals' underlying attitudes, beliefs, motivation, skills, and knowledge. Likewise, studies on developing implementation strategies have mostly presumed that healthcare professionals behave in a rational way (31, 32). However, we could have missed the individuals' automatically enacted habits, as important factors, and this could be a limitation. Recent literature indicates that implementation strategies should consider whether the behavior that needs to be changed to implement a new intervention is perhaps more automatic than deliberate (33). Yet, this is still less explored in implementation science (32). In this dissertation, not capturing the influence of the individuals' habitual behavior on implementing TCIs can be explained in several ways. First, the scoping review did not reveal implementation factors pointing

GENERAL DISCUSSION

at habits or automated behaviors, which could be due to the data collection methods of the included studies. Second, the views of the Delphi experts, self-reports of healthcare professionals, and the use of a rhetoric focused on innovation and implementation concepts across our studies might have exposed more cognitive-related influences. Third, throughout this research, there was no active exploration for such 'non-rational' or habitual determinants. Nevertheless, it cannot be completely ruled out that the individuals' habitual behaviors were not addressed in this dissertation. Few of the selected implementation strategies were based on theories of behavioral change including theories of automatic, impulsive, and habitual behavior (34).

Furthermore, given the issue of overlooking automatic behaviors and habits in implementation, current research suggests nudge strategies to modify the context in which individuals behave and make their decisions (31). Nudge strategies alter the choice architecture in a context, and reshape the different ways in which choices are presented to individuals (35). Moreover, these strategies were shown to be effective in changing behavior across a wide range of contexts (36). Hence, this may be relevant to improve the implementation of TCIs in care settings if habits and automated behavior hinder or enable implementation. Possible applications can include automated computer reminders used for a care transition nurse to perform a telephone call to follow-up with an older person 24 hours post-hospital discharge. Alternatively, a brightly colored notice can be placed in the hospital medical record of an older person indicating the need for a care transition nurse to arrange and send medication information to the community pharmacist. Another suggestion is to prohibit proceeding with documenting patient care procedures in a medical record if the older person's priorities for care transitions are not noted. Thus, accounting for the habit concept implies a potential positive role in improving the implementation of TCIs (37).

FUTURE DIRECTIONS

The findings of this dissertation highlighted the incessant gap existing between the TCIs' ability to improve transitional care for older persons and their implementation in real-world care settings. This dissertation provides several implications for practice and research, discussed in this section.

PRACTICE

First, this dissertation showed that organizations involved in implementing TCIs tend to have little awareness of their context. Moreover, organizations tend to be enthusiastic and fast-forward the implementation of TCIs without a prior thorough assessment and understanding of the contextual factors. To ensure the successful implementation of TCIs, it is crucial to study the context carefully and identify the barriers and facilitators in it before any

implementation effort. Hence, the list of priority factors provided in this dissertation could be used as a starting point. By exploring these factors locally, organizations can obtain an early and essential knowledge of what will most likely hinder or enable the implementation of TCIs, as well as explore new or unique factors in their context. Second, we recommend that organizations use implementation strategies for TCIs to address the relevant influencing factors identified within their context. Therefore, our selection of implementation strategies can be used by organizations as a guide to implement TCIs in a better way.

Third, this dissertation highlighted the critical role of inter-organizational interactions in implementing TCIs. Thus, we advise organizations involved in implementing TCIs to foster their inter-organizational links to promote their readiness to innovate (38), as well as form alliances to share resources and knowledge.

Fourth, this dissertation pointed out that the field of transitional care requires more implementation science expertise. Thus, practice settings and organizations can align with the emerging concept of implementation support practitioners. Instituting such practitioners in daily practice can help build the frontline implementation capability of healthcare providers and improve their ability to use implementation frameworks, strategies, or other tools to implement TCIs more successfully (39).

RESEARCH

In general, researchers in transitional care and long-term care should incorporate and use more of the wide range of available implementation research theories, frameworks, and models. This will increase the rigor of trials or pilots on implementing TCIs. In addition, it will enhance clarity and consistency across various studies as well as allow for comparisons or compiling common elements/lessons learned on the implementation of TCIs (40).

Furthermore, it is a necessary and foundational step to conduct a contextual analysis (i.e. exploring implementation factors) prior to any research study (e.g., trial, pilot, feasibility, effectiveness) aiming to implement a TCI (41, 42). Results in this dissertation highlighted that there is often a lack of considering the context before implementing TCIs and instead reporting on factors (barriers, facilitators) after the implementation had occurred. Hence, researchers should shift their approach and consider the overall context and TCI early on (43). Examining prospectively the context (e.g., factors, wider healthcare system, long-term care policies, care delivery patterns) in which a TCI will be implemented can inform researchers beforehand on the various implementation challenges, guide their choices of implementation strategies, and help interpret implementation and effectiveness outcomes (41). In addition, we stress the importance of considering the bridging factors in the context of transitional care to highlight the interconnections between two or more care settings and their influence on implementing TCIs (21). Similarly, more research is needed to examine the

role of inter-organizational networks and links across multiple settings in particular while investigating the context of transitional care as well as guidance on how to develop them.

Furthermore, evidence supporting the effectiveness of implementation strategies in implementing innovations specifically in the context of transitional care and long-term care is needed. Therefore, our selection of implementation strategies should be used as a first step toward assessing the effectiveness of these strategies to implement TCIs in different care settings. This will help identify the prevalent components of effective implementation strategies for TCIs as well as build a repository of evidence for these strategies (16). Besides, future studies on the implementation of TCIs should more often involve older persons and their families/informal caregivers as participants. However, there is also a need to explore how to involve them and more guidance on how to do so specifically in implementation research studies.

CONCLUSIONS

The studies in this dissertation explored the implementation of TCIs, which are promising solutions to enhance the care continuity for older persons during transitions between healthcare settings. TCIs operate at the juncture of various care organizations and services, which makes their implementation complex and ambiguous. An interplay of multiple factors challenges the implementation of TCIs, and these at the organizational level were found to be highly important as well as factors depicting interactions with other organizations and the external context. Careful consideration of these factors and usage of specific implementation strategies to address them provide a huge opportunity to implement TCIs successfully and improve transitional care for older persons.

REFERENCES

- 1. Eccles MP, Mittman BS. Welcome to implementation science. Implement Sci. 2006;1(1):1.
- Peters DH, Adam T, Alonge O, Agyepong IA, Tran N. Republished research: Implementation research: what it is and how to do it. Br J Sports Med. 2014;48(8):731-6.
- Aarons GA, Green AE, Trott E, Willging CE, Torres EM, Ehrhart MG, et al. The roles of system and organizational leadership in system-wide evidence-based intervention sustainment: a mixed-method study. Adm Policy Ment Health. 2016;43(6):991-1008.
- Hamilton AB, Finley EP. Qualitative methods in implementation research: An introduction. Psychiatry Res 2019;280:112516.
- Damschroder LJ, Aron DC, Keith RE, Kirsh SR, Alexander JA, Lowery JC. Fostering implementation of health services research findings into practice: a consolidated framework for advancing implementation science. Implement Sci. 2009;4(1):50.
- 6. Kirk MA, Kelley C, Yankey N, Birken SA, Abadie B, Damschroder L. A systematic review of the use of the Consolidated Framework for Implementation Research. Implement Sci. 2016;11:72.
- 7. Birken SA, Powell BJ, Shea CM, Haines ER, Alexis Kirk M, Leeman J, et al. Criteria for selecting implementation science theories and frameworks: results from an international survey. Implement Sci. 2017;12(1):124.
- Safaeinili N, Brown-Johnson C, Shaw JG, Mahoney M, Winget M. CFIR simplified: Pragmatic application of and adaptations to the Consolidated Framework for Implementation Research (CFIR) for evaluation of a patientcentered care transformation within a learning health system. Learn Health Syst. 2019;4(1):e10201.
- Flottorp SA, Oxman AD, Krause J, Musila NR, Wensing M, Godycki-Cwirko M, et al. A checklist for identifying determinants of practice: a systematic review and synthesis of frameworks and taxonomies of factors that prevent or enable improvements in healthcare professional practice. Implement Sci. 2013;8(1):35.
- 10. Aarons GA, Hurlburt M, Horwitz SM. Advancing a conceptual model of evidence-based practice implementation in public service sectors. Adm Policy Ment Health. 2011;38(1):4-23.
- 11. Dy SM, Ashok M, Wines RC, Rojas Smith L. A framework to guide implementation research for care transitions interventions. J Healthc Qual. 2015;37(1):41-54.
- 12. Nasa P, Jain R, Juneja D. Delphi methodology in healthcare research: How to decide its appropriateness. World J Methodol. 2021;11(4):116-29.
- Becker-Haimes EM, Ramesh B, Buck JE, Nuske HJ, Zentgraf KA, Stewart RE, et al. Comparing output from two methods of participatory design for developing implementation strategies: traditional contextual inquiry vs. rapid crowd sourcing. Implement Sci. 2022;17(1):46.
- Rapport F, Smith J, Hutchinson K, Clay-Williams R, Churruca K, Bierbaum M, et al. Too much theory and not enough practice? The challenge of implementation science application in healthcare practice. J Eval Clin Pract 2021;28(6):991-1002.
- Robertson T, Simonsen J. Challenges and opportunities in contemporary participatory design. Des Issues. 2012;28:3-9.
- 16. Beidas RS, Dorsey S, Lewis CC, Lyon AR, Powell BJ, Purtle J, et al. Promises and pitfalls in implementation science from the perspective of US-based researchers: learning from a pre-mortem. Implement Sci. 2022;17(1):55.
- Domecq JP, Prutsky G, Elraiyah T, Wang Z, Nabhan M, Shippee N, et al. Patient engagement in research: a systematic review. BMC Health Serv Res. 2014;14(1):89.
- Gray-Burrows KA, Willis TA, Foy R, Rathfelder M, Bland P, Chin A, et al. Role of patient and public involvement in implementation research: a consensus study. BMJ Qual Saf. 2018;27(10):858-64.
- Schilling I, Gerhardus A. Methods for involving older people in health research a review of the literature. Int J Environ Res Public Health. 2017;14(12).
- Scott WR, Davis GF. Organizations and organizing: Rational, natural and open systems perspectives: Routledge; 2015.
- 21. Moullin JC, Dickson KS, Stadnick NA, Rabin B, Aarons GA. Systematic review of the Exploration, Preparation, Implementation, Sustainment (EPIS) framework. Implement Sci. 2019;14(1):1.
- Lengnick-Hall R, Willging C, Hurlburt M, Fenwick K, Aarons GA. Contracting as a bridging factor linking outer and inner contexts during EBP implementation and sustainment: a prospective study across multiple U.S. public sector service systems. Implement Sci. 2020;15(1):43.
- Lengnick-Hall R, Stadnick NA, Dickson KS, Moullin JC, Aarons GA. Forms and functions of bridging factors: specifying the dynamic links between outer and inner contexts during implementation and sustainment. Implement Sci. 2021;16(1):34.
- 24. Parmigiani A, Rivera-Santos M. Clearing a path through the forest: a meta-review of interorganizational relationships. J Manage. 2011;37(4):1108-36.
- Birken SA, Bunger AC, Powell BJ, Turner K, Clary AS, Klaman SL, et al. Organizational theory for dissemination and implementation research. Implement Sci. 2017;12(1):62.
- 26. Tomlinson PR. Co-operative ties and innovation: some new evidence for UK manufacturing. Res Policy.

2010;39(6):762-75.

- 27. Elmuti D, Kathawala Y. An overview of strategic alliances. Manag Decis. 2001;39(3):205-18.
- Dyer JH, Singh H. The Relational View: cooperative strategy and sources of interorganizational competitive advantage. Acad Manage Rev. 1998;23(4):660-79.
- 29. Dyer JH, Singh H, Hesterly WS. The relational view revisited: a dynamic perspective on value creation and value capture. Strateg Manag J. 2018;39(12):3140-62.
- Gulati R, Lavie D, Singh H. The nature of partnering experience and the gains from alliances. Strateg Manag J. 2009;30(11):1213-33.
- Yoong SL, Hall A, Stacey F, Grady A, Sutherland R, Wyse R, et al. Nudge strategies to improve healthcare providers' implementation of evidence-based guidelines, policies and practices: a systematic review of trials included within Cochrane systematic reviews. Implement Sci. 2020;15(1):50.
- 32. Nilsen P, Roback K, Broström A, Ellström P-E. Creatures of habit: accounting for the role of habit in implementation research on clinical behaviour change. Implement Sci. 2012;7(1):53.
- 33. Nilsen P, Potthoff S, Birken SA. Conceptualising four categories of behaviours: implications for implementation strategies to achieve behaviour change. FrontHealth Serv. 2022;1.
- Kok G, Gottlieb NH, Peters G-JY, Mullen PD, Parcel GS, Ruiter RAC, et al. A taxonomy of behaviour change methods: an Intervention Mapping approach. Health Psychol Rev. 2016;10(3):297-312.
- Last BS, Buttenheim AM, Timon CE, Mitra N, Beidas RS. Systematic review of clinician-directed nudges in healthcare contexts. BMJ Open. 2021;11(7):e048801.
- 36. Mertens S, Herberz M, Hahnel UJJ, Brosch T. The effectiveness of nudging: a meta-analysis of choice architecture interventions across behavioral domains. Proc Natl Acad Sci USA. 2022;119(1).
- Potthoff S, Rasul O, Sniehotta FF, Marques M, Beyer F, Thomson R, et al. The relationship between habit and healthcare professional behaviour in clinical practice: a systematic review and meta-analysis. Health Psychol Rev. 2019;13(1):73-90.
- 38. van den Hoed MW, Backhaus R, de Vries E, Hamers JPH, Daniëls R. Factors contributing to innovation readiness in health care organizations: a scoping review. BMC Health Serv Res. 2022;22(1):997.
- Albers B, Metz A, Burke K. Implementation support practitioners a proposal for consolidating a diverse evidence base. BMC Health Serv Res. 2020;20(1):368.
- 40. Moullin JC, Dickson KS, Stadnick NA, Albers B, Nilsen P, Broder-Fingert S, et al. Ten recommendations for using implementation frameworks in research and practice. Implement Sci Commun. 2020;1(1):42.
- Mielke J, Leppla L, Valenta S, Zullig LL, Zúñiga F, Staudacher S, et al. Unraveling implementation context: the Basel Approach for coNtextual ANAlysis (BANANA) in implementation science and its application in the SMILe project. Implement Sci Commun. 2022;3(1):102.
- Rogers L, De Brún A, McAuliffe E. Defining and assessing context in healthcare implementation studies: a systematic review. BMC Health Serv Res. 2020;20(1):591.
- 43. Pfadenhauer LM. Conceptualizing context and intervention as a system in implementation science: learning from complexity theory; comment on "stakeholder perspectives of attributes and features of context relevant to knowledge translation in health settings: a multi-country Analysis". Int J Health Policy Manag. 2022;11(8):1570-3.



CHAPTER

Summary Samenvatting

SUMMARY

Care transitions between multiple care settings and providers are common among older persons with chronic diseases and multimorbidity. Yet, these transitions are frequently hampered by various issues such as fragmented care, medication errors, or poor communication among healthcare providers leading to adverse events for older persons. Therefore, Transitional care innovations (TCIs) are novel solutions designed to enhance care continuity and coordination for older persons when transferring between multiple care settings.

While the development of TCIs has flourished in the recent past, their implementation into "real-world practice settings" is difficult and complex. The aim of this dissertation was to explore the factors that influence the implementation of TCIs, and to develop a set of implementation strategies to address them and enhance the process. The current chapter summarizes all the studies performed in this dissertation.

Starting with a scoping literature review (chapter 2), four overarching different types of TCIs were reported to have been implemented. The majority of these TCIs aimed to improve care transitions and mostly focused on transitions from hospital to home settings, other pathways were from hospital to intermediary care places then to a final destination, and hospital or home to a nursing/residential care facility. Few TCIs were aimed at preventing care transitions, specifically from a nursing facility or home to a hospital. Twenty-five common and prominent factors were found to influence the implementation of these diverse TCIs. Notable hindering factors included the low organization's readiness for implementation, the innovation's complexity and mismatch between its components and the intended profile of the recipients (i.e. older persons), lack of clear implementation plans, and misconceptions or insufficient knowledge about the innovation by healthcare professionals or older persons. While key enabling factors comprised a high perceived advantage of the innovation by healthcare professionals, presence of frontline healthcare professionals with designated transition roles, and continuous monitoring and evaluation of the innovation's implementation process. Moreover, other factors such as leadership engagement, engaging of key stakeholders, external policy and incentives, and skills, competencies, and other personal attributes of healthcare professionals were identified, yet had an almost equivalent influence as both impeding and facilitating the implementation of TCIs.

Furthermore, in a Delphi study (chapter 3) conducted with a panel of international experts in the fields of implementation of innovations, transitional care, and long-term care, 11 factors were conceded upon and prioritized as the most important (consensus level \geq 85%) in the implementation of TCIs. The majority of these factors were at the organizational level and

included leadership engagement, availability of resources, information continuity, sense of urgency, and relative priority. In addition, the engagement of stakeholders and reflecting / evaluating of the TCIs' implementation were also priority factors. The experts also concluded certain interrelationships among the priority factors, whereby some factors are catalysts (e.g., sense of urgency, relative priority) to induce the implementation of TCIs. Otherwise, there was a prevalent agreement among experts on the difficulty to address these priority factors with implementation strategies, indicating the perceived struggle to change factors at the organizational level.

In chapter 4, a retrospective collective case study on the implementation of four TCIs in Belgium aiming to improve care transitions between hospital and home/community settings for older persons highlighted the intuitive manner commonly present in implementation. A lack of understanding about the influencing factors prior to implementation as well as the absence of a deliberate selection of implementation strategies to be used were found across four different cases of transitional care innovations. Similar to the findings from the previous studies, organizational factors constituted once again the key barriers to implementing the four innovations. While, the presence of highly committed, motivated, and enthusiastic individuals played a key role to facilitate the implementation. Engagement of key persons, actors, and partners was the most significant implementation strategy used to implement the four innovations. Concerning the implementation outcomes of the four innovations, high adoption was a key outcome as well as acceptability and benefit to the older persons. While the appropriateness of the interventions' components to match the needs of the older persons was reported as not always achieved. In another qualitative study (chapter 5), the stakeholders' experiences with the same four TCIs implemented in Belgium were further explored. Stakeholders indicated that their active involvement and taking their suggestions to develop the TCIs was empowering. Moreover, they valued their work autonomy, which reinforced their decision-making process and increased their motivation to implement the TCIs. In addition, they could build multidisciplinary care partnerships and collaborations, which supported them to improve transitional care delivery for older persons.

The compilation of the cumulative findings of all the studies led to meeting the final objective of this dissertation. By following the Implementation Mapping methodology, a set of forty implementation strategies for TCIs was systematically developed **(chapter 6)**. These strategies addressed the priority factors at the organizational, individual, policy, and innovation levels that influence the implementation of TCIs. Moreover, the selected strategies were supported by theories on either behavioral change or organizational change, and empirical evidence on their effectiveness in implementing change in healthcare settings. The larger number of strategies were at the organizational level (e.g., structural redesign, changes in staffing models, organizational diagnosis and feedback) and followed

237

by strategies at the individual level (e.g., active learning, belief selection, guided practice). Fewer strategies were at the policy (e.g., advocacy and lobbying) and innovation levels (e.g., tailoring). Suggestions for practical applications of the strategies (e.g., bring experts on innovations in transitional care to model tasks and skills required and to provide ongoing implementation support on-site) as well as who would be the target person/entity (e.g., care transition nurse) were provided to facilitate their use by future implementers of TCIs.

Chapter 7 summarizes the main findings of all studies included in this dissertation, followed by methodological and theoretical considerations. Furthermore, it presents several implications for the practice setting and future research. First, it provides knowledge on priority factors that influence the implementation of TCIs and a selection of theory and evidence-based strategies to address these factors and improve the implementation. Second, it provides directions for future research in the field of transitional care and implementation of TCIs by indicating the need to utilize more implementation science concepts, conduct contextual analysis prior to implementation, and build further evidence on the effectiveness of implementation strategies for TCIs.

SAMENVATTING

Bij ouderen met chronische ziekten en multimorbiditeit zijn zorgtransities tussen verschillende zorginstellingen en zorgverleners een veel voorkomend verschijnsel. Toch worden deze transities vaak belemmerd door verschillende oorzaken, zoals gefragmenteerde zorg, medicatiefouten of slechte communicatie tussen zorgverleners, wat leidt tot ongewenste uitkomsten voor ouderen. Transitional care innovations (TCI's) zijn daarom nieuwe oplossingen om de zorgcontinuïteit en -coördinatie voor ouderen bij de transitie tussen verschillende zorg instellingen te verbeteren.

Hoewel de ontwikkeling van TCI's in het recente verleden snel is gegaan, is de implementatie ervan in "echte praktijksituaties" moeilijk en complex. Het doel van dit proefschrift is de factoren te onderzoeken die van invloed zijn op de implementatie van TCI's, en implementatiestrategieën te ontwikkelen om deze factoren aan te pakken en het proces te verbeteren. Het huidige hoofdstuk geeft een samenvatting van alle studies die in dit proefschrift zijn uitgevoerd.

Uit een verkennend literatuuronderzoek (hoofdstuk 2) bleek dat er vier verschillende soorten TCI's waren geïmplementeerd. De meeste van deze TCI's waren gericht op het verbeteren van zorgtransities en richtten zich meestal op transities van het ziekenhuis naar de thuissituatie, andere trajecten waren van het ziekenhuis naar intermediaire zorgplaatsen en vervolgens naar een eindbestemming en van ziekenhuis of thuis naar een verpleeg-/verzorgingsinstelling. Weinig TCI's waren gericht op het voorkomen van transities, specifiek van een verpleeghuis of thuis naar het ziekenhuis. Er werden vijfentwintig gemeenschappelijke en prominente factoren gevonden die de implementatie van deze diverse TCI's beïnvloeden. Opvallende belemmerende factoren waren de geringe bereidheid van de organisatie tot implementatie, de complexiteit van de innovatie en de mismatch tussen de onderdelen ervan en de beoogde doelgroep (d.w.z. ouderen), het ontbreken van duidelijke implementatieplannen, en misvattingen of onvoldoende kennis over de innovatie bij zorgverleners of ouderen. Belangrijke faciliterende factoren waren daarentegen een verwachting van een groot voordeel van de innovatie door medewerkers, de aanwezigheid van 'frontline' medewerkers met een aangewezen transitierol en voortdurende monitoring en evaluatie van het implementatieproces van de innovatie. Bovendien werden andere factoren, zoals betrokkenheid van het leiderschap, betrokkenheid van de voornaamste belanghebbenden, extern beleid en beloningen, en vaardigheden, competenties en andere persoonlijke eigenschappen van medewerkers, geïdentificeerd, maar deze werden zowel als belemmerend en bevorderend bevonden voor de implementatie van TCI's.

Vervolgens werden in een Delphi-studie **(hoofdstuk 3)** met een panel van internationale deskundigen op het gebied van implementatie van innovaties, transitie-zorg en langdurige

CHAPTER 8

zorg, 11 factoren gevonden en geprioriteerd als de belangrijkste (consensusniveau ≥ 85%) bij de implementatie van TCI's. De meeste van deze factoren lagen op het niveau van de organisatie, waaronder leiderschapsbetrokkenheid, beschikbaarheid van middelen, informatiecontinuïteit, gevoel van urgentie en relatieve prioriteit. Daarnaast waren ook de betrokkenheid van de belanghebbenden en de reflectie/evaluatie van de uitvoering van de TCI's belangrijke factoren. De deskundigen concludeerden ook bepaalde onderlinge verbanden tussen de factoren, waarbij sommige factoren als katalysator fungeren (bijv. gevoel van urgentie, relatieve prioriteit) om de uitvoering van de TCI's te stimuleren. Tot slot waren de deskundigen het overwegend eens over de moeilijkheid om deze factoren met implementatiestrategieën aan te pakken, hetgeen wijst op de complexiteit om factoren op organisatieniveau te veranderen.

In hoofdstuk 4 werden retrospectieve collectieve casestudie uitgevoerd over de implementatie van vier TCI's in België ter verbetering van zorgtransities tussen ziekenhuis en de thuissetting. Hier werd de intuïtieve manier die vaak aanwezig is bij de implementatie benadrukt. Een gebrek aan inzicht in de beïnvloedende factoren voorafgaand aan de implementatie en het ontbreken van een weloverwogen selectie van te gebruiken implementatiestrategieën werden aangetroffen in vier verschillende gevallen van TCI's. Vergelijkbaar met de bevindingen van de vorige studies, vormden organisatorische factoren opnieuw de belangrijkste belemmeringen voor de implementatie van de vier innovaties. De aanwezigheid van zeer betrokken, gemotiveerde en enthousiaste personen speelde een sleutelrol bij het vergemakkelijken van de implementatie. De betrokkenheid van sleutelpersonen, actoren en partners was de belangrijkste implementatiestrategie die gebruikt werd om de vier innovaties te implementeren. Met betrekking tot de resultaten van de implementatie van de vier innovaties was een hoge adoptie een belangrijk resultaat, evenals de aanvaardbaarheid en de voordelen voor de ouderen. De geschiktheid van de onderdelen van de interventies voor de behoeften van de ouderen werd echter niet altijd bereikt. In een andere kwalitatieve studie (hoofdstuk 5) werden de ervaringen van de belanghebbenden met dezelfde vier in België uitgevoerde TCI's verder onderzocht. De stakeholders gaven aan dat hun actieve betrokkenheid en het meenemen van hun suggesties bij de ontwikkeling van de TCI's een stimulans was. Bovendien waardeerden zij hun werkautonomie, wat hun besluitvormingsproces versterkte en hun motivatie om de TCI's te implementeren verhoogde. Bovendien konden zij multidisciplinaire zorgpartnerschappen en samenwerkingsverbanden opzetten, wat hen ondersteunde bij het verbeteren van de transities voor ouderen.

De compilatie van de cumulatieve bevindingen van alle studies leidde tot de uiteindelijke doelstelling van dit proefschrift. Door de methode van Implementation Mapping te volgen, werd systematisch een reeks van veertig implementatiestrategieën voor TCI's ontwikkeld **(hoofdstuk 6)**. Deze strategieën hadden betrekking op de belangrijkste factoren op

SUMMARY | SAMENVATTING

organisatorisch, individueel, beleids- en innovatieniveau die de implementatie van TCl's beïnvloeden. De geselecteerde strategieën werden ondersteund door theorieën over gedrags- of organisatieverandering, en door empirisch bewijs van hun effectiviteit bij het doorvoeren van veranderingen in zorgomgevingen. Het grootste aantal strategieën lag op organisatorisch niveau (bijv. structureel herontwerp, veranderingen in personeelsmodellen, organisatorische diagnose en feedback), gevolgd door strategieën op individueel niveau (bijv. actief leren, selectie van overtuigingen). Er waren minder strategieën op beleidsniveau (bv. belangenbehartiging en lobbyen) en op innovatieniveau (bv. maatwerk). Er werden suggesties gedaan voor praktische toepassingen van de strategieën (bv. deskundigen op het gebied van innovaties bij transities inschakelen om als model te dienen en de vereiste taken en vaardigheden te illustreren en ter plaatse voortdurende ondersteuning bij de implementatie te bieden). Tevens werden suggesties gedaan over wie de doelgroep zou zijn van de implementatie strategieën (bv. de verpleegkundige in de transitie-zorg) om het gebruik ervan door toekomstige uitvoerders van TCl's te vergemakkelijken.

Hoofdstuk 7 vat de belangrijkste bevindingen van alle in dit proefschrift opgenomen studies samen, gevolgd door methodologische en theoretische overwegingen. Verder worden verschillende implicaties voor de praktijk en toekomstig onderzoek gepresenteerd. Ten eerste biedt het kennis over de belangrijkste factoren die de implementatie van TCI's beïnvloeden en geeft het een selectie van theorie en evidence-based strategieën om deze factoren aan te pakken en de implementatie te verbeteren. Ten tweede geeft het suggesties voor toekomstig onderzoek op het gebied van transitie-zorg en implementatie van TCI's door aan te geven dat het nodig is meer concepten van implementatiewetenschap te gebruiken, contextuele analyse uit te voeren voorafgaand aan implementatie, en meer bewijs op te bouwen over de effectiviteit van implementatiestrategieën voor TCI's.



CHAPTER

Impact

Successful implementation of Transitional Care Innovations (TCIs) should not be taken for granted. Consideration of contextual factors and usage of implementation strategies are fundamental for any prospective initiative to implement TCIs in long-term care (LTC).

The outputs of this dissertation contribute to the key objectives of the European TRANS-SENIOR research consortium (1) to improve the implementation of TCIs and hence enhance the delivery of transitional care for older persons.

PROMOTING SOLUTIONS FOR AN AGEING SOCIETY

As global societies age, older persons demand more LTC services and face frequent care transitions between multiple care settings (2). Care transitions come with a high risk of negative health consequences and poor quality of care (3, 4). Therefore, innovations to deliver better transitional care for older persons are needed. TCIs are desirable solutions to enhance transitional care for older persons (at the individual level) and to relieve the increased pressure on LTC service demands and social care systems (at the societal level) (5). Worldwide, the development and implementation of TCIs have gained momentum and been applied more recently. Thus, the findings of this dissertation can have a large societal impact. Primarily, it will inform governments, policymaking entities, organizations, and innovators in LTC on the existing challenges in implementing TCIs and provide evidence-based practical methods to improve the process. Furthermore, it will offer guidance on the proper implementation of TCIs to help translate these innovations into the real world, increase their reach/use, and improve their sustainability. This will contribute to strengthening LTC delivery systems, reversing the negative effect of care transitions in societies, and thus achieving better population health.

GUIDANCE FOR LTC ORGANIZATIONS (PRACTICE)

Guidance on how to implement successfully available TCIs is needed in practice. LTC organizations can benefit greatly from first, knowing the existing types of TCIs and their key components; second, knowing what factors predominantly influence the implementation of TCIs in different care settings; and third practical methods to use in implementing TCIs. Key guiding tools developed in this dissertation can be valuable for practical use. The established compilation of 20 different TCIs (chapter 2) can increase the knowledge and awareness of healthcare professionals on a wide range of available innovations implemented previously in multiple care settings. Moreover, the thorough description of these TCIs (i.e. care transition pathways, target population, aims, and key components) is informative and helps limit recreating similar innovations in practice. Instead, practice can select an existing TCI and focus

more on its implementation rather than development. This can help save time and effort, often scarce in LTC organizations, to develop TCIs from scratch.

Furthermore, the list of priority factors that influence the implementation of TCIs is another guiding tool for practice (chapter 3). Insight into the critical barriers and facilitators is instrumental in helping LTC organizations assess their readiness and capabilities to implement a TCI. This can serve as a checklist to support innovators and implementers in practice to understand early on their specific implementation context and judge whether a new TCI has a chance to be implemented.

Additionally, the set of selected implementation strategies for TCIs (chapter 6) provides an extensive list of strategies to guide implementers in practice. This set provides multiple suggestions on how to use each strategy in a practical way to implement TCIs. Therefore, future projects on implementing a TCI in practice can start with a clear guidance on implementation by utilizing the three complimentary outputs of this research – the list of available TCIs, important factors to consider, and the relevant implementation strategies and their practical applications. Moreover, this work will be shared with various LTC partner organizations within the TRANS-SENIOR consortium. Specifically, the Living Lab in Ageing and Long-term care in the Netherlands will use the set of implementation strategies (chapter 6) in future studies on implementing innovations for care transitions from home to nursing home settings.

INFORMING FUTURE TRANSITIONAL CARE POLICY

This dissertation can help inform future LTC policy at a European level. Findings on priority factors that influence the implementation of TCIs can provide input for the development of future policies to support the implementation of TCIs (6). For example, financing of TCIs' implementation is an important barrier and thus can be used as a key issue to address in a policy brief presented to policymakers at national/European levels. In addition, having frontline staff with a transition role is an important facilitator and can be used as a key recommendation in a policy brief created for LTC organizations. Similarly, the selection of implementation strategies for TCIs can guide organizational policies on implementing innovations. For instance, the strategy 'perform organizational diagnosis and feedback' can be developed into a policy that demands LTC organizations to assess their context and prepare before any prospective project to implement a TCI.

Furthermore, partner organizations within the TRANS-SENIOR consortium - AGE Platform Europe and the World Health Organization - are prominent parties in shaping future LTC policies. Hence, they can use the findings of this dissertation as insights for their advocacy and policy development work at the European level. Moreover, based on the findings of this dissertation, a set of policy briefs will be created and dialogues with national stakeholder groups on LTC in Europe will be held. These dialogues will highlight the existing issues in implementing TCIs, the research findings, options for consideration, and recommendations to promote successful implementation.

DISSEMINATION OF FINDINGS

The findings of this dissertation were disseminated through various channels. Studies conducted were published in peer-reviewed, international, and high-impact open-access scientific journals such as Implementation Science and The Gerontologist. Moreover, findings were presented at multiple international conferences such as the European Implementation Event, and the Gerontological Society of America Annual Scientific Meeting. Furthermore, findings were shared among the TRANS-SENIOR consortium network, other researchers, and partners through regular webinars and training events.

The TRANS-SENIOR consortium website hosts all the published articles of this dissertation and a YouTube video summarizing this research (1). An online toolbox of implementation strategies will be developed based on the outputs of this dissertation and will be made available in English on the websites of the Living Lab in Ageing and Long-term care (Academische Werkplaats Ouderenzorg Limburg) (7) and TRANS-SENIOR (1). This toolbox will be freely accessible for use by healthcare professionals planning to implement TCIs and will help to widespread innovation in LTC practices.

BECOMING A HEALTHCARE INNOVATOR

On a professional level, performing this research within TRANS-SENIOR has equipped me with the necessary knowledge, research expertise, and transferable skills needed to become both a healthcare innovator and an independent researcher. Being part of a multidisciplinary and multi-sectoral consortium as well as working with proficient research groups on LTC and innovation has built my capabilities to create solutions to improve transitional care. Moreover, I gained a large network of fellow researchers, innovators, implementers, and experts across Europe.

REFERENCES

- 1. TRANS-SENIOR EU Project. [Available from: https://www.trans-senior.eu/].
- European Commission, Directorate-General for Employment, Social Affairs and Inclusion. Long-term care report: trends, challenges and opportunities in an ageing society. Volume I. Luxembourg: Publications Office of the European Union; 2021.
- 3. Coleman EA, Boult C, American Geriatrics Society Health Care Systems C. Improving the quality of transitional care for persons with complex care needs. J Am Geriatr Soc. 2003;51(4):556-7.
- 4. Coleman EA. Falling through the cracks: challenges and opportunities for improving transitional care for persons with continuous complex care needs. J Am Geriatr Soc. 2003;51(4):549-55.
- Birtwell K, Planner C, Hodkinson A, Hall A, Giles S, Campbell S, et al. Transitional care interventions for older residents of long-term care facilities: a systematic review and meta-analysis. JAMA Netw Open. 2022;5(5):e2210192.
- Oh A, Abazeed A, Chambers DA. Policy implementation science to advance population health: the potential for learning health policy systems. Front Public Health. 2021;9.
- 7. Academische Werkplaats Ouderenzorg Limburg. [Available from: https://www.awolimburg.nl/].



ADDENDA

Publications Acknowledgments About the Author Living Lab in Ageing and Long-Term Care

PUBLICATIONS

INTERNATIONAL SCIENTIFIC JOURNALS:

- Fakha A, Groenvynck L, de Boer B, van Achterberg T, Hamers J, Verbeek H. A myriad of factors influencing the implementation of transitional care innovations: a scoping review. Implement Sci. 2021; 16(1):21. <u>https://doi.org/10.1186/s13012-021-01087-2</u>
- Fakha A, de Boer B, van Achterberg T, Hamers J, Verbeek H. Fostering the implementation of transitional care innovations for older persons: prioritizing the influencing key factors using a modified Delphi technique. BMC Geriatr. 2022; 22(1):131. <u>https://doi.org/10.1186/ s12877-021-02672-2</u>
- Fakha A, Leithaus M, de Boer B, van Achterberg T, Hamers JP, Verbeek H. Implementing four transitional care interventions for older adults: a retrospective collective case study. Gerontologist. 2022. https://doi.org/10.1093/geront/gnac128
- **Fakha A**, de Boer B, Hamers J, Verbeek H, van Achterberg T. Systematic development of a set of implementation strategies for transitional care innovations in long-term care. (submitted, under review).
- Groenvynck L, Fakha A, de Boer B, Hamers JPH, van Achterberg T, van Rossum E, Verbeek H. Interventions to improve the transition from home to a nursing home: a scoping review. Gerontologist. 2021. <u>https://doi.org/10.1093/geront/gnab036</u>
- Leithaus M, Fakha A, Flamaing J, Verbeek H, Deschodt M, van Pottelbergh G, Goderis G. Stakeholders' experiences and perception on transitional care initiatives within an integrated care project in Belgium: a qualitative interview study. BMC Geriatr. 2023; 23(1):41. https://doi.org/10.1186/s12877-023-03746-z

CONFERENCE ABSTRACTS:

- Fakha A, de Boer B, van Achterberg T, Hamers J.P., Verbeek H. Fostering the implementation of innovations to improve care transitions for older persons: an expert consensus. Evidence and Implementation Summit (EIS), March 30-31, 2021; Australia.
- Fakha A, de Boer B, van Achterberg T, Hamers J.P., Verbeek H. Implementation of innovations in transitional care: exploring and consolidating the influencing factors. European Implementation Event (EIE), May 27-28, 2021; Rotterdam, the Netherlands.
- Fakha A, de Boer B, van Achterberg T, Hamers J.P., Verbeek H. Implementation of transitional care innovations: considering the organizational context and process is key. (Symposium - transitions to long-term residential care settings). The Gerontological Society of America (GSA) Annual Scientific Meeting, November 10-14, 2021; Phoenix, AZ, USA.
- Fakha A, de Boer B, van Achterberg T, Hamers J.P., Verbeek H. Innovations in transitional care. (Symposium - transitional care in senior citizens (TRANS-SENIOR): lessons from the literature). International Scientific Nursing and Midwifery Conference, 4th edition (CARE4 2022), February 8-10, 2022; Ghent, Belgium.

- Fakha A, de Boer B, Hamers J.P., Verbeek H, van Achterberg T. Systematic development of implementation strategies tailored for Transitional Care Innovations (TCIs): a practical guide. European Geriatric Medicine Society (EuGMS), 18th International Congress, September 28-30, 2022; London, UK.
- Fakha A, Leithaus M, de Boer B, van Achterberg T, Hamers J.P., Verbeek H. Exploring the implementation of four transitional care interventions for older adults: a collective case study. The Gerontological Society of America (GSA) Annual Scientific Meeting, November 2-6, 2022; Indianapolis, IN, USA.

ACKNOWLEDGMENTS

My PhD journey has been an enriching, exciting, and challenging experience. This transformative journey has taught me to be resilient; stay focused on the future, and be persistent in overcoming obstacles that may have appeared insurmountable. It made me believe that there is always a solution.

I thank every person who provided me with unwavering support throughout my PhD, making it a remarkable, memorable, and successful accomplishment. Foremost, I express my sincere gratitude to *Maastricht University & KU Leuven and my supervisory team Prof. Dr. Hilde Verbeek, Prof. Dr. Jan P.H. Hamers, Prof. Dr. Theo van Achterberg, and Dr. Bram de Boer.* I was fortunate to have such a team that always provided me with continuous guidance, knowledge, and encouragement. I cannot express enough appreciation for having you as advisors by my side, surpassing all expectations.

Prof. Dr. Hilde Verbeek – Thank you for directing me at every step, moment, and decision along this journey and for challenging me to explore new knowledge and research methods. Importantly, you constantly encouraged me to look at things from a different angle and challenged me to embrace novel ways of thinking. Your support has profoundly shaped my academic journey.

Prof. Dr. Jan P.H. Hamers – Thank you for providing me with the opportunity to perform my PhD within an international research consortium and for believing in my capabilities. Your vision has been instrumental in refining and improving my research to its fullest potential.

Prof. Dr. Theo van Achterberg – Thank you for introducing me to the world of implementation science research and for challenging me at every step to build my expertise in this field. You were not only an advisor but also a mentor. Your invaluable guidance has played an integral role in elevating the quality of my research. Thanks for always pushing me to think critically and to go beyond my limits in the pursuit of excellence.

Dr. Bram de Boer – Thank you for the daily supervision, prompt advice, and sharp direction in navigating the ins and outs of performing a PhD. I learned a lot from you, mostly on how to be pragmatic, think big, and act in a smart way to resolve problems.

Special thanks to my fellow researchers at the **Department of Health Services Research at Maastricht University** and the **Living Lab in Ageing and Long-Term Care** for exchanging knowledge, collaborating, innovating together, and helping each other in our academic journeys. Moreover, thanks to the researchers at the **Academic Centre for Nursing and Midwifery at KU Leuven** who have welcomed me during my joint-degree research work. I am grateful for their open-mindedness and insightful discussions, and stimulating academic exchanges that we had.

I thank greatly the **TRANS-SENIOR Research Consortium** and the **EU** for this lifetime opportunity to be part of an international team and conduct world-class research and develop into a true healthcare innovator. It is a badge of honor to have been a researcher, within this elite Marie Skłodowska-Curie innovation program, an exceptional chance that has expanded my horizons significantly.

Moreover, I extend my appreciation to all the partner organizations of the **TRANS-SENIOR Consortium** for their amazing efforts, diverse expertise, and incessant collaboration, which enriched my skills and knowledge and allowed me to grow professionally. In addition, a special thanks to **MeanderGroep Zuid-Limburg** and **Wit-Gele Kruis** for the unique onsite experiences I had and the valuable knowledge sharing moments I enjoyed with their dedicated teams.

To my fellow **TRANS-SENIOR** researchers and now lifetime friends and networks – thanks for the amazing memories we built together during our PhD journeys, from trainings, to study trips, conferences, webinars, as well as Xmas markets, dinners, and exploring new cities. It was a great addition to my journey.

Dr. Ramon Daniels – Thank you for being my mentor and for your guidance and wisdom, as well as your energy, enthusiasm, and confidence in me to attain new career heights.

I thank my new colleagues at the **Department of Innovation Management & Strategy at the University of Groningen** for their warm welcome and support.

Lindsay – If I were to repeat my PhD years, I would definitely choose you again as my PhD comrade. It was a great pleasure and honor to have gone through this journey together. Thank you for your impeccable advice on both work and life.

Monique – It was a great chance that our research journeys have crossed paths. Thanks for the meaningful informal chats we always enjoyed on "innovations in healthcare" and for the exchange of tips & tricks for doing a PhD, as well as the good laughs.

Cosette – The definition of a lifetime friend, advisor, and mentor. We did not meet by mistake, and our life journeys always intertwined. Thank you for being a role model in many ways. You are always a phone call away. Thank you for giving me the "best PhD advice" and

for laughing hard when things were difficult. I cannot thank you enough.

Rayan – My amazing friend. Thanking you is an underrated thing; you are my rock. Thank you for being my greatest supporter and for teaching me how to live practically and to look at things objectively.

Tatjana – Thanks for being a valuable friend and supporter throughout my PhD. Your advice was always right. Thanks for the many funny moments and memories that we had through our similar academic journeys.

Thanks to all my friends across the world (Lebanon, France, UK, Qatar, Dubai, Switzerland). Thanks for having faith in me and for always checking on me during the PhD.

To my beautiful sisters, thank you for always being there for me and for being my pillars of strength. I love you all immensely.

Hiam – Thank you for being my safety net, and for seeing far ahead and guiding me as such. Thanks for your wise counsel and continuous encouragement.

Fatina – Thank you for being my family in Europe, for your constant support, guidance, and care. You were always there for me throughout this journey and helped me in many ways, mostly to stay determined and focused on the end goal.

Nouhad – Thank you for being my biggest fan, for always being ready to lend an ear, and for encouraging me to stay positive and keep going no matter what. Thank you for your endless support and for always being by my side.

To my special nieces and nephews (Kenda, Farah, Alizee, Annabel, Karim, Jude, Zein) – thank you for all your love, care, and support.

Dear Father (Adnan) – Thank you for teaching me to never settle for less, to stay strong, and to always aim high.

Dear Mother (Samira) – I can only imagine how much you would have been happy and proud of me becoming a Dr., it was one of your dreams, I made it true and I made you proud. Thank you for your love and for staying with me throughout my PhD journey - with your presence, smile, touch, smell, heart, and soul. You gave me the inspiration, strength, and determination to always seek and conquer new achievements. Thank you, Mom!

ABOUT THE AUTHOR

Amal Fakha was born on March 4, 1984, in Beirut, Lebanon. She obtained her Bachelor of Sciences in Nutrition and Dietetics in 2005 from the American University of Beirut, Lebanon. Amal is a licensed dietitian by the Ministry of Public Health in Lebanon and had worked previously as a Clinical Nutrition Therapist specializing in pediatrics and surgery care units at a governmental university hospital in Lebanon.

In 2009, she moved to Paris, France, and completed her Master's degree in International Public Health (concentration in Management & Health Policy) at the École des Hautes Etudes en Santé Publique (EHESP). Following her master's studies, Amal returned to the Middle East (Qatar, United Arab Emirates) where she worked as a Healthcare Consultant with national (Qatar Foundation) and international companies (Munich Re). Notably, she was a key planning consultant for a green-field project to develop a health informatics-driven hospital facility. Moreover, she worked on the business development of multiple strategic health insurance companies in the Middle East and North Africa region. Before starting her doctoral studies, Amal led a national health promotion project funded by the European Union in Lebanon and lectured in the global public health program at EHESP as an alumni contributor.

In 2019, Amal was awarded to pursue her joint-doctoral degree as a Marie Skłodowska-Curie early-stage researcher within the TRANS-SENIOR research consortium, a European Union's Horizon 2020 research and innovation program. Since then, Amal has been living in the Netherlands and have performed jointly her PhD research project at Maastricht University (Care and Public Health Research Institute), and Katholieke Universiteit Leuven (KU Leuven). Amal's PhD research focused on improving the implementation of innovations in long-term care.

In between her major career steps, Amal took a sabbatical and traveled solo to Vietnam, Thailand, and Sri Lanka, where she enjoyed beautiful yoga retreats, nature, and cultural adventures.

Currently, Amal is an Assistant Professor of Innovation Management and Strategy at the Faculty of Economics and Business, University of Groningen.

LIVING LAB IN AGEING AND LONG-TERM CARE

This dissertation is part of the Living Lab in Ageing and Long-Term Care, a formal and structural multidisciplinary network consisting of Maastricht University, nine long-term care organizations (MeanderGroep Zuid-Limburg, Sevagram, Envida, Cicero Zorggroep, Zuyderland, Vivantes, De Zorggroep, Land van Horne & Proteion), Intermediate Vocational Training Institutes Gilde and VISTA college and Zuyd University of Applied Sciences, all located in the southern part of the Netherlands. In the Living lab, we aim to improve quality of care and life for older people and quality of work for staff employed in long-term care via a structural multidisciplinary collaboration between research, policy, education and practice. Practitioners (such as nurses, physicians, psychologists, physio- and occupational therapists), work together with managers, researchers, students, teachers and older people themselves to develop and test innovations in long-term care.

ACADEMISCHE WERKPLAATS OUDERENZORG LIMBURG

Dit proefschrift is onderdeel van de Academische Werkplaats Ouderenzorg Limburg, een structureel, multidisciplinair samenwerkingsverband tussen de Universiteit Maastricht, negen zorgorganisaties (MeanderGroep Zuid-Limburg, Sevagram, Envida, Cicero Zorggroep, Zuyderland, Vivantes, De Zorggroep, Land van Horne & Proteion), Gilde Zorgcollege, VISTA college en Zuyd Hogeschool. In de werkplaats, draait het om het verbeteren van de kwaliteit van leven en zorg voor ouderen en de kwaliteit van werk voor iedereen die in de ouderenzorg werkt. Zorgverleners (zoals verpleegkundigen, verzorgenden, artsen, psychologen, fysio- en ergotherapeuten), beleidsmakers, onderzoekers, studenten en ouderen zelf wisselen kennis en ervaring uit. Daarnaast evalueren we vernieuwingen in de dagelijkse zorg. Praktijk, beleid, onderzoek en onderwijs gaan hierbij hand in hand.

PHD-THESES LIVING LAB IN AGEING AND LONG-TERM CARE

PROEFSCHRIFTEN ACADEMISCHE WERKPLAATS OUDERENZORG LIMBURG

- **Christoph Golz.** Technostress among health professionals: The blame game between health professionals and technology. 2023
- **Teuni Rooijackers**. Supporting older adults to STAY ACTIVE AT HOME. Process, effect and economic evaluation of a reablement training program for homecare staff. 2022
- **Anne van den Bulck.** Differences that matter: Understanding case-mix and quality for prospective payment of home care. 2022
- Marlot Kruisbrink. Towards enhanced management of fear of falling in older people. Unravelling interventions and measuring related avoidance of activity. 2022
- **Ruth Vogel.** Nurses in the Lead: empowering community nurse leaders to implement evidence into practice. 2022
- Fabian Groven. The bed bath with or without water? It's a wash! Experiences with the washing without water intervention used for the bed bath. 2021

- **Roy Haex.** Take a look through my eyes: The development of an experienced quality measure with clients, informal, and formal caregivers in Dutch home care. 2021
- Sascha Bolt. The fundamentals of a DEDICATED palliative approach to care for people with dementia. 2021
- **Angela Mengelers.** To risk or to restrain? Involuntary treatment use in people with dementia living at home. 2021
- **Katya Sion.** Connecting Conversations. Experienced quality of care from the resident's perspective: a narrative method for nursing homes. 2021
- **Linda Hoek.** Change begins with choice. Supporting the autonomy of nursing home residents with dementia through partnership. 2020
- Mirre den Ouden. Every step counts. Daily activities of nursing home residents and the role of nursing staff. 2018
- **Theresa Thoma-Lürken.** Innovating long-term care for older people. Development and evaluation of a decision support app for formal caregivers in community-based dementia care. 2018
- **Eveline van Velthuijsen.** Delirium in older hospitalised patients: diagnosis and management in daily practice. 2018
- **Bram de Boer.** Living at a green care farm. An innovative alternative for regular care in nursing homes for people with dementia. 2017
- **Nienke Kuk.** Moving forward in nursing home practice. Supporting nursing staff in implementing innovations. 2017
- **Irma Everink.** Geriatric rehabilitation. Development, implementation and evaluation of an integrated care pathway for older patients with complex health problems. 2017
- Ramona Backhaus. Thinking beyond numbers. Nursing staff and quality of care in nursing homes. 2017
- Martin Van Leen. Prevention of pressure ulcers in nursing homes, a big challenge. 2017
- Mariëlle Daamen-Van der Velden. Heart failure in nursing home residents. Prevalence, diagnosis and treatment. 2016
- Armand Rondas. Prevalence and assessment of (infected) chronic wounds. 2016
- Hanneke Beerens. Adding life to years. Quality of life of people with dementia receiving long-term care. 2016 (Cum Laude)
- **Donja Mijnarends.** Sarcopenia: a rising geriatric giant. Health and economic outcomes of community-dwelling older adults with sarcopenia. 2016
- **Tanja Dorresteijn.** A home-based program to manage concerns about falls. Feasibility, effects and costs of a cognitive behavioral approach in community-dwelling, frail older people. 2016
- **Basema Afram.** From home towards the nursing home in dementia. Informal caregivers' perspectives on why admission happens and what they need. 2015
- **Noemi Van Nie-Visser.** Malnutrition in nursing home residents in the Netherlands, Germany and Austria. Exploring and comparing influencing factors. 2014

- **Silke Metzelthin.** An interdisciplinary primary care approach for frail older people. Feasibility, effects and costs.
- **Esther Meesterberends.** Pressure ulcer care in the Netherlands versus Germany 0-1. What makes the difference? 2013

Math Gulpers. EXBELT: expelling belt restraints from psychogeriatric nursing homes. 2013

- **Hilde Verbeek.** Redesigning dementia care. An evaluation of small-scale homelike care environments. 2011
- Judith Meijers. Awareness of malnutrition in health care, the Dutch perspective. 2009
- Ans Bouman. A home visiting program for older people with poor health. 2009
- **Monique Du Moulin.** Urinary incontinence in primary care, diagnosis and interventions. 2008
- **Anna Huizing.** Towards restraint free care for psychogeriatric nursing home residents. 2008
- **Pascalle Van Bilsen.** Care for the elderly, an exploration of perceived needs, demands, and service use. 2008
- **Rixt Zijlstra.** Managing concerns about falls. Fear of falling and avoidance of activity in older people. 2007

Sandra Zwakhalen. Pain assessment in nursing home residents with dementia. 2007

ADDENDA

